

The Journal of Indian Management & Strategy

"J-Gate, EBSCO Discovery, Summon (Proquest), Google Scholar, Indian Science Abstracts, Indian Citation Index (RII - 0.016), InfoBase Index (IB Factor 2016 - 2.6), SJIF Impact Factor 2016 - 4.756, Cosmos Impact Factor, Emerging Sources Citation Index (Thomson Reuters), Web of Science."

Approved by UGC-Care (Group-II)

Changing Financial Landscape Through Mudra Yojana – A Journey Towards Entrepreneurial Ecosystem Vikram K. Joshi	4	
Green Insolvency: An Intertwined Connection Between Climate Change and Credit Risk Shefali Sethi, Anand Sharma	16	
A Study of Workplace Spirituality and Job Commitment with the Mediation of Occupational Stress Archana Mittal, Dr. Geeta Mishra, Dr. Sunita Bishnoi	23	Research
Fintech and Industry 4.0: A Study on Drivers with Reference to Digital Banking Sector Deepa Joshi, Sanjay Sharma	32	
A Study on Antecedents of Founder Endorsement and Their Impact on Brand Equity and Consumer Buying Intentions in Jewellery Industry Devi Prasad Ungarala, SudhaVemaraju	40	
What Makes online Gaming Platforms Sticky: A Study of Five Indian Metros Soumitra Bose, Kiran Bala	49	
IKEA India: Building Diversity And Inclusion Arulsenthilkumar S, Punitha N	58	Case Study



A TRUE VISIONARY

"You see things and you say **Why**? But I dream of things that never were and say **Why** not?"

- George Bernard Shaw



Shri Jagannath Gupta (1950 - 1980)

Also a true visionary...who dared to dream! He lives no more but his dreams live on....and on!

JIMS (Rohini)	-	1993
JIMS (Kalkaji)	-	1997
JIMS (Vasant Kunj)	-	2003
JIMS (Jaipur)	-	2003
JNIT (Jaipur)	-	2004
JIMS (Greater Noida)	-	2008
Jagannath University (Jaipur)	-	2008
Jagannath University (Bahadurgarh)	-	2013

And more dreams to come!



(ADMINISTRATION) SATISH KUMAR DOGRA PRODUCTION ASSISTANTS KRISHNA ARYA NEELAM VISHWAKARMA *****

Editorial Offices & Subscriber Service Strategic Consulting Group OCF, Pocket-9, Sector-B, Vasant Kunj New Delhi-110070 Phone: 011-40619300, E-mail: jims.8m@jagannath.com, Website: www.jimsd.org Available Online at www.indianjournals.com RNI No. 64562/96 Online ISSN No: 0973-9343 Exclusively Marketed and Distributed by indianjournals.com

Editor's Desk India's Phenomenal Decade of Growth: A Journey of Progress and Potential

In the span of the last decade, India has emerged as a global economic powerhouse, orchestrating a remarkable transformation that has left the world in awe. The nation's growth trajectory over these years has been nothing short of phenomenal, propelling it to the forefront of the global stage. As we reflect upon India's journey of the past decade, it becomes evident that the nation's growth has been multifaceted, encompassing economic, technological, and societal dimensions. Economic Metamorphosis: Over the past ten years, India's economy has undergone a significant transformation. The adoption of pro-business policies, economic reforms, and emphasis on entrepreneurship has led to a substantial increase in foreign direct investment. The 'Make in India' initiative spurred manufacturing and contributed to the nation's position as a manufacturing hub. Sectors like information technology, pharmaceuticals, and renewable energy have flourished, creating new avenues for growth and employment. India's consistent GDP growth rate has bolstered its status as one of the world's fastest-growing major economies. Technological Advancements: In the technological sphere, India has rapidly advanced, leveraging its vast pool of skilled IT professionals. The 'Digital India' campaign has fostered digital literacy and access to technology across the country, narrowing the urban-rural divide. India's software services have been instrumental in driving global innovation, and the startup ecosystem has witnessed a surge, with a flurry of unicorns making their mark on the global stage. The 'Atmanirbhar Bharat' initiative has further underscored the country's commitment to technological self-reliance. Societal Progress: The last decade has seen significant societal progress in India. There have been substantial improvements in sectors such as healthcare, education, and women's empowerment. Initiatives like 'Swachh Bharat Abhiyan' have led to enhanced sanitation and cleanliness, impacting the quality of life for millions. While challenges remain, the nation has shown its resilience in addressing issues like poverty, malnutrition, and access to basic amenities. Geopolitical Significance: India's growth over the last decade has not only benefited its citizens but also positioned the nation as a crucial player on the global stage. Strengthening diplomatic ties, trade relationships, and international collaborations have enhanced India's geopolitical significance. The 'Act East' policy has deepened India's engagement with Southeast Asian nations, while the nation's role in forums like BRICS and G20 has grown in prominence. Challenges on the Horizon: As India charts its path forward, there are challenges that need attention. Income inequality, environmental sustainability, and infrastructure development are pressing concerns that require concerted efforts. The country must ensure that growth is inclusive and benefits every stratum of society. In conclusion, India's growth story of the last decade is a testament to the nation's resilience, adaptability, and potential. The strides made in the economic, technological, and societal realms have positioned India as a global leader, poised to shape the course of the 21st century. However, with great progress comes great responsibility. As India continues its journey, it must remain committed to sustainable growth, social equity, and international collaboration, thus ensuring that the next decade is even more transformative and generative.

About the Journal

JIMS 8M: The Journal of Indian Management and Strategy is committed to publishing scholarly, empirical and theoretical research articles that have a high impact in the field of Management. The Journal is peer-reviewed and is published quarterly. It covers domains such as business strategy and policy, human resource management, organizational behavior, operations, finance, entrepreneurs ip, organizational theory and research methodology. The journal provides an intellectual platform for advancement and dissemination of management knowledge and also fosters collaborative research. It has an inclusive ethos and is open to a wide range of methodological approaches and philosophical underpinnings.

Views and factual claims expressed in individual contributions are personal to the respective contributors and are not necessarily endorsed by the editors, their advisors, or the publishers of the journal.

Guidelines for Authors

Authors are strongly encouraged to submit their manuscripts electronically via email to jims.8m@jagannath.org and on receipt of the manuscript, an acknowledgement is emailed to the author. The manuscript after passing through the Plagiarism check through software is evaluated for the original content. If the original content is less than 85% excluding the references, then the author is advised to revise and rewrite the original content. Once this is achieved by the author, the manuscript is sent to the reviewer for review. Based on the final report of the reviewer the final decision for publishing the article is taken by the Managing editor and same is conveyed to the author.

Guidelines for Research Papers/Case Studies/ Perspectives/Book Reviews

- Articles must be sent by e-mail to jims.8m@jagannath.org.
- Each manuscript must be accompanied with an abstract of 150-200 words.
- Manuscript should not exceed 5000 words.
- The font should be 12 points and Times New Roman with 1.5-line spacing.
- The author's name, designation, affiliation, complete address with mobile number must be provided on a separate sheet.
- All drawings, graphs, and tables should be provided on separate pages.
- Case Studies should be original and not published anywhere.
- Book reviews should include name of the author, publisher, price and year of publication and ISBN number. If any references are used, details of the same have to be provided.
- Perspectives should depict emerging issues and ideas that contribute to the think-tank of managers, administrators and policy makers.
- Authors must acknowledge the sources consisting of other author's concepts or data or findings and exhibits. References cited in the main text should also be listed in the reference list. Citation should also be included as a reference in the end of the paper which should be in APA style in alphabetical and chronological order.
- Articles submitted for consideration in JIMS 8M have to be accompanied with a declaration by the author/authors that they have not published or submitted the manuscript for publication elsewhere.
- Editorial decisions are communicated within a period of 8 weeks of the receipt of manuscript.
- In case the reviewer suggests revision of the manuscript, the author has to modify the manuscript and submit the revised manuscript within 7-10 days.
- The first author will receive one hard copy of the journal.
- The format of the article should start with the Title, Authors, Abstract, Keywords, Introduction, I. Review of Literature, II. Research Design and Methods, III. Results and Discussion, IV. Conclusion, References (APA Style) and the figures and tables will be inserted in the text as when illustrated and explained by the author.
- Plagiarism: Authors should contribute their original work. To ensure originality and essence of the research contribution of authors we use plagiarism software and the authors are bound to adhere to it.

*For details of the guidelines, please refer to the link -

https://www.jimsd.org/resources/journals/GuidelinesAuthors8M.pdf

SUBSCRIPTION FORM

	Subscription R	ates (4 Issues)	
CATEGORY	Period	Print*	Online
	1 Year	1,600.00	863.00
Institution/Individuals	2 Year	3,000.00	
	3 Year	4,500.00	
Students**	1 Year	750.00	500.00
	Foreign	(USD)	
CATEGORY	Period	Print*	Online
Institution/Individuals	1 Year	135.00	118.00
Students**	1 Year	100.00	80.00

Subscription Rates-2023

Terms & Conditions:

- 1. *Print** Subscription includes online access.
- 2. *Students*** should send a photocopy of their identity cards.
- 3. Print Subscription is Volume Based, whereas Online Subscription is Calendar Year Based and is subject to renewal.
- 4. Online Subscription includes current subscriptions + back files from 2005, if available.

Ordering Information

Subscriptions: Payment has to be made in favor of "indianjournals.com" payable at New Delhi, India.

The Manager				
Sales and Marketingindianjou				
B-9, Basement, Local Shopp	ing Complex, "A" Block,Nar	aina Vihar, Ring H	Road, New De	elhi - 110028
Dear Sir,				
I/We would be interested	in subscribing to JIMS 8M	for	year(s).	I/We would be interested in
availingmyself/ourselves of	the subscription offer in	the	_category.	I/We am/are enclosing a
cheque/DD				
No	dated	drawn on		(specify Bank),
favouring				
indianjournals.com for Rs.				
	My/Our pai	rticulars are as	under:	
Name: Mr./Ms./M/s:				
Profession:				
Tel No:	Fax:			
	For subscri	iption please co	ntact:	
		aniournals com		

B-9, Basement, Local Shopping Complex, "A" Block Naraina Vihar, Ring Road, New Delhi-110 028Ph: 91 11-45055535

CHANGING FINANCIAL LANDSCAPE THROUGH MUDRA YOJANA – A JOURNEY TOWARDS ENTREPRENEURIAL ECOSYSTEM

Vikram K. Joshi*

Purpose: The government of India launched Pradhan Mantri Mudra Yojana (PMMY) on 8 April, 2015 to fund the unfunded. This initiative has high potential to change the entrepreneurial ecosystem in country and achieve economic inclusion of the informal sector members. Hence, the study is proposed to evaluate the said potential of the initiative.

Design/Methodology/Approach: The study is descriptive study based on primary data of 288 Mudra loan beneficiaries and 42 bank officials of Nagpur district of Maharashtra State, India. The data collected was quantitative based on various attributes using a structure questionnaire for beneficiaries pre and post-Mudra loan and bank officials to evaluate the impact of Mudra loans on Entrepreneurial activity of small and informal businesses.

Findings: The analysis shows that the beneficiaries in Nagpur district received loans as per their requirements. The employment, average salary paid, number of suppliers, monthly purchases from suppliers, investment in fixed capital and approximate profits all have increased significantly in all the categories of Mudra loan and in overall Nagpur district post-Mudra loans and created good expansionary effect for businesses.

Originality/value: A new process of Loan disbursal is recommended to avoid the leakage in the process and avoid possible defaults. The new process can help not only to achieve the financial inclusion, but also will create an entrepreneurial ecosystem that will foster equity for every participant member.

Key Words: Informal Businesses, Formal Banking System, Pradhan Mantri Mudra Yojana. **JEL Classification Code:** E26, E51, I38

The Economic Survey of 2018-19 released on July 4, 2019 mentioned that almost 93% of the total workforce is informal. According to Niti Aayog's Strategy for New India at 75, released in November 2018, 'India's informal sector employees approximately 85% of all workers'. ILO in its recent report says that close to 81% of all employed persons in India make a living by working in the informal sector. Unfortunately, the members of this sector have suffered through economic exclusion even after 30 years of globalization era which is evident from the fact that Crisil Inclusix Index registered the score of 50.1 at the end of fiscal 2013 and increased marginally to 58.0 at the end of FY16 (Crisil Inclusix Report 2018). To ensure the effective participation of these economic agents and to give boost to the economic activity, it is necessary to change the economic architecture of the country. This demands for evolving a mechanism where money-deficient informal sector gets an access to money through formal banking system. Identifying the gap rightly, the Government of India launched Pradhan Mantri Mudra Yojana (PMMY) on 8 April 2015. Under this initiative a new institution called Micro Units Development and Refinance Agency Bank (MUDRA Bank) has been set up for the development of micro units and refinance of Micro Finance Institutions (MFIs) to encourage entrepreneurship in India & provide the funding to the non-corporate small business sector. The initiative has completed more than 5 years of operations and hence it is necessary to evaluate the performance of PMMY initiative from the perspectives of beneficiaries who are at the receiving end. The present study tries to address the following research questions: RO1: Whether the initiative has contributed significantly in employment generation, business and profit expansion? RQ2: How much is the contribution of the initiative in creating startup and entrepreneurial ecosystem through expansionary effect it has created in the market in terms of employment, income and profit growth? RQ3: Are there any leakages in the loan disbursal process? RQ4: What should be the model for better deliverables? The study is presented in six sections: Section I reviews the performance of initiative based on Mudra data, Section II presents the summary of literature review, Section III discusses the data and methodology, Section IV discusses the performance of the initiative from beneficiaries' perspective, Section V presents the new process evolved from the study and Section VI concludes.

Review of Performance of PMMY during 2015-20

Under the Pradhan Mantri MUDRA Scheme, MUDRA Bank has launched its three-initiative products: MUDRA Bank Shishu Loans up to Rs 50,000; MUDRA Bank Kishor Loans between Rs 50,001 and Rs 5 lakh & MUDRA Bank Tarun Loans of Rs 5-10 lakh. They signify the stage of growth and funding needs of the micro units or entrepreneur.

* Assistant Professor – DMT, Shri Ramdeobaba College of Engineering and Management, Nagpur The scheme has a total corpus of Rs 20,000 crore which can lend between Rs 50,000 and Rs 10 lakh to small entrepreneurs and banks and Micro-finance institutions can draw refinance under the MUDRA Scheme after becoming member-lending institutions of MUDRA. A brief review of performance of PMMY initiative based on Mudra data is presented below:

Credit Accounts Opened & Amount Disbursed under PMMY

The credit accounts opened and amount disbursed under PMMY during the year 2015-16 to 2019-20 in 36 States /Union territories of India is presented in table 1:

		f Credit Accou		edit A/Cs & Al India)	Women	New	SC / ST / OBC
		Kishore	Tarun	Total		Entrepreneur	
						A/Cs	
2015-16	3,24,01,046	20,69,461	4,10,417	3,48,80,924	2,76,28,265	1,24,74,668	1,84,01,499
	(92.89)	(5.93)	(1.18)	(100)	(79.21)	(35.76)	(52.75)
2016-17	3,64,97,813	26,63,502	5,39,732	3,97,01,047	2,91,46,894	99,89,470	2,25,00,194
	(91.93)	(6.71)	(1.36)	(100)	(73.42)	(25.16)	(56.67)
2017-18	4,26,69,795	46,53,874	8,06,924	4,81,30,593	3,35,58,238	1,25,59,327	2,62,24,114
	(88.65)	(9.67)	(1.68)	(100)	(69.72)	(26.09)	(54.49)
2018-19	5,15,07,438	66,06,009	12,56,871	5,92,70,318	3,70,62,562	1,33,93,802	2,81,35,095
	(86.90)	(11.15)	(2.96)	(100)	(62.53)	(22.60)	(47.47)
2019-20	5,44,90,617	64,71,873	12,85,116	6,22,47,606	3,91,03,349	1,19,13,903	2,97,50,100
	(87.54)	(10.40)	(2.06)	(100)	(62.82)	(19.14)	(47.79)
Growth	13.63 %	42.54 %	42.62 %	15.69 %	8.30 %	-0.89 %	12.33 %
	Amo	ount Disbursen	nent (In Rs Cr	ores)			
2015-16	62,027.69	41,073.28	29,853.76	1,32,954.73	63,190.43	58,908.08	49,196.33
	(46.65)	(30.89)	(22.45)	(100)	(47.53)	(44.31)	(37.00)
2016-17	83,891	51,063.12	40,357.13	1,75,312.13	78,249.77	69,973.96	66,279.82
	(47.85)	(29.13)	(23.02)	(100)	(44.63)	(39.91)	(37.81)
2017-18	1.04,228.05	83,197.09	59,012.25	2,46,437.41	1,00,170.55	93,656.61	83,686.97
	(42.29)	(33.76)	(23.95)	(100)	(40.65)	(38.00)	(33.96)
2018-19	1,39,651.55	99,868	72,291.84	3,11,811.38	1,29,153.23	1,00,925.58	98,023.68
	(44.79)	(32.03)	(23.18)	(100)	(41.42)	(32.37)	(31.44)
2019-20	1,62,813	91,427	75,475	3,29,715	1,42,846.22	94,896.09	1,12,029.49
	(49.38)	(27.73)	(22.89)	(100)	(43.32)	(28.78)	(33.98)
Growth	32.49 %	24.52 %	30.56 %	29.59 %	25.21 %	12.21 %	25.54 %
	Avera	age Amount D					
2015-16	19,143.73	1,98,473.32	7,27,400.66	38,116.75	22,871.66	47,222.16	26,734.95
2016-17	22,985.45	1,91,714.21	7,47,725.35	44,158.06	26,846.69	70,047.72	29,457.44
2017-18	24,426.66	1,78,769.53	7,31,323.52	51,201.82	29,849.76	74,571.36	31,912.22
2018-19	27,112.89	1,51,177.51	4,11,480.63	52,608.35	34,847.35	75,352.45	34,840.35
2019-20	29,879.09	1,41,268.22	5,87,301.10	52,968.30	36,530.43	79,651.55	37,656.84
2019-20. M	thor's Computati udra Bank (www nts percentage of	v.mudra.org.in)		erall Performan	ce Report 2015	-16, 2016-17, 201	7-18, 2018-19 &

Table 1: MUDRA Loan – Credit A/Cs & Amount Disbursement

As seen in table 1, the total credit accounts opened has almost doubled and the total amount disbursed has tripled during 2015-16 to 2019-20, i.e., in just 5 years' time. The simple growth rate is calculated for the same and it is observed that the total number of credit accounts opened increased at 15.69% and the amount disbursed increased at 29.59% during

the same period. The credit accounts opened increased at higher rate for Kishore (42.54%) and Tarun (42.62%) compared to Shishu (13.63%), but the amount disbursed increased at higher rate for Shishu (32.49%) compared with Kishore (24.52%) and Tarun (30.56%).

In case of Women and SC/ST/OBC it increased with good numbers but increase is relatively low for New Entrepreneurs. Secondly, during 2018-19 to 2019-20, the number of accounts opened in Kishore, Tarun and New Entrepreneurs category has declined, but increased in case of Shishu and for Women & SC/ST/OBC category. It is noteworthy that pandemic situation erupted in India during February-March 2020. However, the rise in credit a/cs opened in Shishu, Women and SC/ST/OBC is indicative that the decline in Kishore and Tarun category loans cannot be attributed to the impact of pandemic.

Thirdly, the percentage of credit accounts opened and the loan amount disbursed is highest in Shishu followed by Kishore and Tarun category during 2015-16 to 2019-20. But, the average loan amount disbursed increased in Shishu, but declined in Kishore and Tarun loans during the same period. Also, the percentage of credit a/cs opened by Women and SC/ST/OBC seems to high but, the percentage of amount disbursed seems to be less, whereas the percentage of credit accounts opened is relatively high in comparison with the loan amount disbursed. Thus, it can be inferred that most of the loans given to Women and SC/ST/OBC are Shishu loans, as the average amount disbursed to Women (Rs 22,871.66 – 36,530) and SC/ST/OBC (Rs 26,734.95 – 37,656.84). But the average loans given to New Entrepreneurs (Rs 47,222.16 – 79,651.55) include mostly the beneficiaries of Kishore and Tarun loans since the maximum limit of Shishu loan is Rs 50,000.

Thus, looking at the higher percentage of credit accounts opened and loan amount disbursed in Shishu category, it can be drawn that the micro and small businesses which were denied credit by formal banking system are benefited most by this initiative. The women entrepreneurs are also adequately served by the initiative which is evident from the rise in average amount disbursed to them. Also, the financial needs of the budding entrepreneurs belonging to SC/ST/OBC category is also fairly addressed.

I. Review of Literature

The summary of various similar initiatives is presented below that are taken by the other countries to uplift the members of informal sector and micro and small enterprises and its comparison is done with MUDRA Loan initiative:

	Name of Fund	Referred Studies	Criteria	Activities / Remarks	MUDRA Loan
SN	/Country/Year				
1	PCF – People's Credit Fund Vietnam (July 1993)	 Seibel and Thac, 2012 SBV, Annual Report, 2018 	 To fulfil the needs of the poor and marginalize classes. Offers tailor-made affordable financial products. Collaterals required. 	• Training provided to the members.	 Fulfils the needs of poor and marginalize classes with affordable financial products. Collateral free. No training component.
2	Uwezo Fund – Kenya (Published on 21 February, 2014)	 Uwezo Fund Uwezo Fund Guide, 2013 Otieno, 2014 Olima, 2016 	 Disbursed to group or Institutions with members aged between 18-35. Collaterals required. Eligible qualifying amounts for a group shall be minimum of Kshs 50,000 and a maximum of Kshs 500,000 at any one time. 	 Training provided to groups. Training certificates are issued. Offers non-financial services like – information, skills and mentorship. 	 Disbursed to individuals. No collaterals. Three loan
3	KUR – Kredit	 International 	• Designed for MSMEs	• For KUR Mikro,	• In Mudra: Shishu

 Table 2: Funds in Other Countries catering to SMEs – Criteria & Activities

	Usaha Rakyat – Indonesia (2007)	Finance Corporation Report, 2016; Bank Mandiri • The Jakarta Post, 2019	 who have feasible business but are not bankable. Collateral Free Loan for KUR mikro and KUR migrant workers. Security required for KUR Retail. 	Indonesian rupiah 25 million, max tenure 2 years, for KUR Migrant, Indonesian rupiah 25 million, max tenure 12 months, for KUR Retail, between Indonesian rupiah 25-500 million, max tenure 3-5 years.	upto Rs 50,000, Kishore upto Rs 5 lakhs and Tarun upto Rs 10 lakhs. • Max teneure 5 years. • After one loan repayment, another Mudra loan can be taken. • Collateral free loans. • For micro, small and informal businesses.
4	SEFA – Small Enterprise Finance Agency – South Africa (1 April, 2012)	 Zeleke Worku, 2016 SEFA Fund Leadership Magazine, April 2017 SEFA Services 	 Caters to the financial needs of informal and micro entrepreneurs Collaterals required Survivalists and microenterprises – Loans between R500 and R5000. Small Enterprises – Loans between R50000 and R1million. Medium Enterprises – Loans between R1 million and R5 million. 	 Offers skill-based training on entrepreneurial activities. Offers post loan business support and institutional strengthening support during the life cycle stages of the business. 	 Caters to the financial needs of micro, small and informal businesses. No skill-based training. No non-financial business support.
5	SMEDCO – Small and Medium Enterprise Development Corporation – Zimbabwe (Formed in 1983, amended in Feb. 2014)	 Reserve Bank of Zimbabwe Zeparu & Baz, 2014 SMEDCO 	 Caters to members of informal sector. Provides Microloan - \$500 and less, tenure 4 months, moveable property is collateral. For loans within the threshold of \$501-\$5000, collateral security & business plan or proposal. 	• Vocational and technical training is provided to rural enterprises.	 Caters to the financial needs of micro-small and informal businesses. Collateral free with max tenure of 5 years. No training.
6	FAMPE – Fundo de Aval para as micro e-pequenas Empresas – Brazil (1996)	 Roberto Zavatta, 2008 ILO, 2014 SABREA, Small Business Experts 	 Caters to the needs of informal business activities. Collaterals required. Provides guarantees typically up to US\$40,000, the guarantee can cover up to 50% of the value of the loan, but in some special cases the limit can be raised to 80%. 	Offers financial and non-financial support in the form of entrepreneurial education, promoting, production and competitive environment for small business linkages between small and large	 Caters to the financial needs of micro, small and informal businesses. Collateral free. No non-financial support is provided.

(2009) Manoj Kumar, collaterals free loan 2018 • The loan an	roup earsto apply for loan.irrespective education level.• Insurance provided
---	---

Source: *Compiled from the Literature Review of various funds.*

From the table 2, it is observed that the other countries are providing non-financial services which are beyond funding the businesses while tackling the challenge of funding informal sector and small businesses. Training seems to be the important component amongst the other non-financial services.

Review of Studies on MUDRA Loan:

SN	Studies	Remarks / Conclusions
1	Venkatesh & Lavanya Kumari (2015)	Capacity to hugely contribute through financing small business entrepreneurs and can create an ecosystem for
2	Shahid & Irshad (2016); Haris & Subash (2018); Vijai (2018)	 entrepreneurship. Small manufacturing units and self-employed individual in rural and urban areas will be benefited, formalize the informal sector and fill the finance requirement gap by funding the unfunded. Boost morale of aspiring
		young educated and skilled workers to kick-start entrepreneurial activity and reduce the indebtedness through access to formal banking.
3	Rudrawar & Uttarwar (2016)	• Will act as a catalyst for growth of GDP, employment generation and entrepreneurship at large in the country.
4	Kumbhare & Ujjwal Kumar (2016); Aravindaraj & Bala (2018)	• Less awareness amongst the prospective beneficiaries especially those belonging to SCs and OBCs due to less exposure to banking services, so possibility that many prospective beneficiaries may be deprived of the initiative and is a

		matter of approximate addressed
		matter of concern to addressed immediately.
5	Joshi V. K. (2017)	 The initiative appears to be encouraging for small businesses and self-employed as evident from the number of accounts opened and amount disbursed. It may enhance the socio-economic status of small business owners and has capacity to create multiplier effect, change the financial inclusion landscape and create entrepreneurial ecosystem in the economy. The possibility of default must also be appropriately addressed
6	Agarwal & Dwivedi (2017); John et. al. (2018)	 as the loans are collateral free. Will be a game changer in financial inclusion by funding the unfunded population, and will create fair competition.
7	Biswas (2019)	• Not only encouraged educated youth but also underprivileged for self- employment. Contributed significantly in women empowerment and making them self-reliant.
8	Mahajan (2019)	• Served fairly well to small businesses as most of the loans disbursed are Shishu category. But the actual impact on income generation, employment generation and business expansion needs to be evaluated.

ce: Compiled from the Literature Review of MUDRA Loans.

Thus, keeping the above aspects in mind, the present study tries to analyze the Mudra loan initiative in terms of its prospects and leakages by surveying the beneficiaries of Mudra loan in selected Nagpur district.

II. Research Design and Methods

The descriptive study is based on field survey of actual beneficiaries of MUDRA Loan in selected talukas of Nagpur

district. Nagpur district comprises of 14 talukas namely; Katol, Kalmeshwar, Kuhi, Kamptee, Narkhed, Umred, Ramtek, Nagpur rural, Nagpur city, Saoner, Parsheoni, Hingna, Bhiwapur and Mouda. Out of these, total eight talukas were selected using convenience sampling. Two talukas Parsheoni and Hingna are combined due to nonThe sample composition area/taluka wise is presented below in table 3:

SN	Area/Taluka	MUDRA Loan Beneficiaries			Bank officials (All Nationalized Banks)			
		Shishu	Kishore	Tarun	Total	Target	Actual	Target
1	Kalmeshwar	14	12	5	31	30	02	02
2	Kamptee	11	13	8	32	30	02	02
3	Katol	9	11	11	31	30	02	02
4	Nagpur	33	40	23	96	90	30	30
5	Nagpur Rural	14	13	07	34	30	02	02
6	Narkhed	14	09	08	31	30	02	02
7	Parsheoni+Hingna	14	15	04	33	30	02	02
8	Total	109	113	66	288	270	42	42
1	(Nagpur District)							

Table 3: Sample Composition and Sample Size

Source: Author's Computation based on Primary data.

Nagpur city being the highest populated taluka 90 sample size (30 each category) and from other talukas 30 sample size (10 each category) of Mudra beneficiaries is considered as representative sample. Similarly, for bank officials, 30 sample from Nagpur city and 02 each from other talukas is considered as representative sample based on availability of banks in the region. Total targeted sample of beneficiaries was 270 and actual beneficiaries surveyed were 288, and total targeted sample of Bank officials were 42 and 42 actual officials were surveyed. The sample of beneficiaries and officials is taken based on convenience sampling method. A structured questionnaire was designed to take the responses of bank officials and beneficiaries of Mudra loan in selected talukas of Nagpur district.

III. Results and Discussions

Mudra Loan Disbursements and Impact Analysis

The impact analysis of Mudra loans is done firstly, on the basis of average amount disbursed to beneficiaries in Nagpur district in comparison with National average, and secondly on the basis of various business indicators viz., employment generation, average salary paid, number of suppliers, purchases from suppliers, investment in fixed capital and profit pre and post Mudra loan. This impact analysis ignores the businesses not opting for any loans or opting for other types of loans or finances other than Mudra loans. Hence, the Mudra loan beneficiaries are considered as the niche group to analyse the impact pre and post Mudra loans.

Average Amount Received by Beneficiaries of Mudra Loan

The average amount received by beneficiaries under each Mudra loan category and area/taluka wise and overall is presented in below in table 4:

Table 4: Area/Taluka wise Average Amount Received by
Beneficiaries

		Den	enciaries		
Area/	n	Shishu	Kishore	Tarun	Average
Taluka					
Kalmesh	31	48,214	2,85,833	8,30,000	2,66,290
war		(n = 14)	(n = 12)	(n = 5)	
Kamptee	32	46,818	3,40,000	7,25,000	3,35,469
		(n = 11)	(n = 13)	(n = 8)	
Katol	31	48,889	3,23,636	7,77,273	4,04,839
		(n = 9)	(n = 11)	(n = 11)	
Nagpur	96	47,682	3,12,300	8,73,913	3,55,891
		(n = 33)	(n = 40)	(n = 23)	
Nagpur	34	43,143	2,23,462	8,32,143	2,74,529
Rural		(n = 14)	(n = 13)	(n = 7)	
Narkhed	31	46,071	2,74,444	9,12,500	3,35,968
		(n = 14)	(n = 9)	(n = 8)	
Parsheoni	33	45,000	2,80,333	7,67,500	2,39,545
+Hingna		(n = 14)	(n = 15)	(n = 4)	
Overall	288	46,628	2,96,300	8,30,227	3,24,165
(Nagpur		(n =	(n =	(n = 66)	
District)		109)	113)		

National	 29,879	1,41,268	5,87,301	52,968
Average				
(2019-20)				
[Taken				
from				
Table 1]				

Source: Authors' Computation based on Primary Data.

The disbursement of loans in Nagpur district seems to be highly commendable and much higher than the national average of 2019-20. Thus, it is pertinent to note that the majority of beneficiaries received the loans in respective categories as per their requirements due to fulfilment of all the necessary documentation the important amongst which is the registration of business.

Impact of Mudra Loan on Various Business Indicators

The impact of Mudra loan on various business indicators viz., Employment, Average salary paid, Number of suppliers, Purchases from suppliers, Investment in Fixed Capital and Profit is shown below in table 5:

Mudra Loan Category	Mean	5. Dusiness mu	Mean Diff.	Paired t-	P value	Remarks
	Pre-Mudra	Post-Mudra		test value		
			nt Generatio	n	I	
Overall (n=288)	1.552	2.333	0.781	10.504	2.2e-16	Significantly increased
Shishu (n=109)	0.477	0.761	0.284	5.439	0.0000	Significantly increased
Kishore (n=113)	1.566	2.353	0.787	5.708	0.0000	Significantly increased
Tarun (n=66)	3.303	4.893	1.590	9.761	1.15e-14	Significantly increased
	Ave	rage Salary Paio	l to Workers	Employed		
Overall (n=288)	5053.819	6986.458	1932.639	15.255	2.2e-16	Significantly increased
Shishu (n=109)	2146.789	3041.284	894.4954	6.580	8.69e-10	Significantly increased
Kishore (n=113)	5517.699	7677.876	2160.177	10.114	2.2e-16	Significantly increased
Tarun (n=66)	9060.606	12318.182	3257.576	12.473	2.2e-16	Significantly increased
		No. of	Suppliers			
Overall (n=288)	6.701	9.256	2.555	22.619	2.2e-16	Significantly increased
Shishu (n=109)	5.100	7.229	2.128	15.423	2.2e-16	Significantly increased
Kishore (n=113)	6.955	9.451	2.495	15.356	2.2e-16	Significantly increased
Tarun (n=66)	8.909	12.272	3.363	10.584	4.43e-16	Significantly increased
		Ionthly Purchas	es from the S	uppliers		
Overall (n=288)	24880.21	43072.57	18192.36	17.259	2.2e-16	Significantly increased
Shishu (n=109)	9866.972	18962.39	9095.413	16.476	2.2e-16	Significantly increased
Kishore (n=113)	32247.788	54168.14	21920.35	10.863	2.2e-16	Significantly increased
Tarun (n=66)	37060.606	63893.94	26833.33	12.07	2.2e-16	Significantly increased
		Investment	n Fixed Capi			
Overall (n=288)	25,760.42	93,907.99	68,147.57	7.485	4.39e-13	Significantly increased
Shishu (n=109)	6,302.752	20,766.06	14,463.3	11.871	2.2e-16	Significantly increased
Kishore (n=113)	17,274.336	65,371.68	48,097.35	6.375	0.0000	Significantly increased
Tarun (n=66)	72,424.242	2,63,560.61	1,91,136.4	5.741	0.0000	Significantly increased
			mate Profit			
Overall (n=288)	32,282.92	59,355.87	27,072.95	21.877	2.2e-16	Significantly increased
Shishu (n=109)	15,726.85	32,138.89	16,412.04	18.28	2.2e-16	Significantly increased
Kishore (n=113)	36,181.82	65,109.09	28,927.27	12.395	2.2e-16	Significantly increased
Tarun (n=66)	53,857.14	95,968.25	42,111.11	19.104	2.2e-16	Significantly increased

Table 5: Business Indicators – Impact Analysis

Source: Author's Computation based on Primary data.

From the table 5, it can be seen that the employment, average salary paid, number of suppliers, monthly purchases from suppliers, investment in fixed capital and approximate profits all have increased significantly in all the categories of Mudra loan and in overall Nagpur district at 1% level of significance. Thus, the Mudra loan initiative not only expanded the businesses of the beneficiaries, but also helped in creating the jobs in the market, increased salary per worker, generated the indirect employment in terms of increase in average number of suppliers, increased purchases from the suppliers, increased fixed capital of the businesses required to run the businesses in the long run and also increased significantly the average profit earned by the beneficiaries across all the categories of Mudra loan. Thus, the initiative is an achievement for the government for creating an ecosystem for entrepreneurship as evident from the business flourishment post Mudra loans.

Expansionary Effect of Mudra Loan

The expansionary effect is evaluated in terms of employment generation potential, income generation potential and profit generation potential of the initiative. The proxy used for expansionary effect of the initiative is the multiplier effect. It shows the snowball effect of economic change (capital infusion) of any kind and its effect on various aspects of economic activity. The study attempted to find this snowball effect on the economy in terms of employment generation (proxy used employment multiplier given by R.F. Kahn in 1931), income generation (proxy used is income multiplier given by J.M. Keynes in 1936) and profit generation (derived on the same lines as that of income multiplier) to understand the infusion of investment through Mudra loans based on the following formulas from the date of receipt of loan till October 2020. These values are calculated based on the data available with respect to direct and indirect employment generated, income generated by the known participants and profit earned by the businesses till mentioned dates. The possible unknown participants who are directly or indirectly benefited are not included in the measurement hence, the numbers are just an indicative measure.

1	
Employment	Income Multiplier, M
Multiplier, K =	$= \frac{Y}{T}$,
ΔΝ	$-\frac{1}{I}$,
$\overline{\Delta N1}$,	where Y stands for the
where K stands for	total income generated,
the employment	I for the change in
multiplier, ΔN for	investment
the increase in total	
employment, and	
ΔN_1 for the increase	
in primary	
employment.	
· · ·	Total Profit Multiplier,
	n -
Multiplier,	$P = \frac{P}{L}$,
$P = \frac{\Delta P}{L}$,	I
where ΔP star	nds for total net profit pre
	ra, P for total profit
-	for total investment. In
,	ne profit multiplier, the
•	
-	by the businesses are only
taken into cor	sideration.

The expansionary effect for employment, income and profits are measured for overall Nagpur district, Mudra loan category wise and are/taluka wise from the date of receipt of Mudra loan till October 2020 and the results are presented below in table 6:

 Table 6: Expansionary Effect: Post Mudra – Area/Taluka and Mudra Loan Category Wise –

 Till October 2020 & Year wise

Mudra Loan	No of	Employment	Income Expansion	Net Profit	Total Profit
Category/Area/	beneficiaries	expansion =	$= (\mathbf{Y}/\mathbf{I})$	Expansion =	Expansion = P /I
Taluka		$\overline{\Delta}N/\Delta N1$		ΔΡ /Ι	_
Overall	288	1.404291	2.122704	2.1395	4.5232
		Year wise Expar	nsionary Effect		
2016	15	1.61971	5.91392	7.58416	12.8853
2017	42	1.43131	3.22306	3.16333	6.8763
2018	110	1.43510	2.21735	2.40939	5.0257
2019	98	1.37417	1.53717	1.24474	2.9003
2020	23	1.30396	0.57949	0.70471	1.8041
	Mud	ra Loan Category w	ise Expansionary Effe	ct	
Shishu	109	1.432566	6.17635	9.64663	18.2122
Kishore	113	1.385254	2.464169	2.35021	5.1264
Tarun	66	1.405707	1.53806	1.31446	2.8849
		Area/Taluka-Wise E	xpansionary Effect		
Nagpur	96	1.42449	2.10008	1.95305	4.1111
Nagpur Rural	34	1.779817	3.157414	3.96035	7.1731
Kalmeshwar	31	1.429658	1.765354	2.37807	4.5439
Kamptee	32	1.358255	2.130694	1.78658	4.0012

Katol		31	1.325641	1.484502	1.62852	3.7575
Narkhed		31	1.414634	1.851224	1.60297	4.1151
Parsheoni Hingna	+	33	1.316017	2.731942	2.54383	5.6161
Source: Author's Computation based on Primary data.						

As can be seen in table 6, in overall Nagpur district, the employment expansion coefficient is 1.40; implying that 1 primary employment created 1.40 total employment (1 primary & 0.40 secondary (indirect) employment). The indirect employment is in the form of number of suppliers increased for the businesses. The year-wise employment generation coefficient during 2016-20 lies in the range 1.30 - 1.61 and area/taluka wise lies in the range 1.31 - 1.77. Thus, it is evident that the Mudra loan initiative has significantly contributed in employment generation in the economy and have high potential to contribute more in days to come.

In overall Nagpur district, the income expansion coefficient is 2.12. This implies that for one unit rise in investment, the total income in the economy increased by 2.12 units indicating high amount of income generation. The income expansion coefficient for 2016-2020 is in the range of 0.57 - 5.91. The area/taluka wise income expansion is in the range 1.48 - 3.15. It can be inferred that the income expansion in the economy created by Mudra loan initiative is highly encouraging as it has contributed significantly in generation of income in a short time span of just 4 - 5 years.

In overall Nagpur district, the net profit expansion coefficient is 2.1395 and total profit expansion coefficient is 4.5232. This implies that for one unit rise in investment, the net profit increased by 2.13 units and total profit increased by 4.52 units indicating high amount of profit expansion effect. The yearwise and area/taluka wise, the profit expansion is highly commendable. It is evident from the profit expansion coefficient that the Mudra loan initiative has contributed significantly in generation of profit in a time span of just 4-5years and is likely to contribute further in years to come. This shows that the PMMY initiative is significantly changing the landscape of non-corporate businesses in terms of growth and employment generation.

Mudra Initiative – Issues of Concerns

Irrespective of success of the initiative, there are certain concerns related to Mudra scheme expressed by various scholars like non-repayment of loans, problems faced by the borrowers etc. These issues of concerns are evaluated by interviewing 42 bank officials in various talukas/areas and obtained the data. The results are presented below in table 7:

SN	Mudra Loan Category	Average No. of Loans	Average Percentage of Borrowers Making Prompt Repayment
1	Shishu	43	74.28 %
2	Kishore	29	88.19 %
3	Tarun	15	94 %
4	Total (Nagpur District)	29	85.48 %

 Table 7: Avg No of Loans Distributed by Banks and Prompt Repayment (%)

The percentage of defaulters is less in all the categories of Mudra loans as evident from the figures in table 7. The borrowers not making the prompt repayment is highest in Shishu category. Also, there are few borrowers in all category of loans who are making delayed repayments. The reasons for non-repayment as identified by bank officials are: ill intentions; the belief of the borrowers that 'since it is the government money, so they need not have to pay back'; misuse of funds by beneficiaries; business failure and business losses; poor economic and financial condition; inability to handle the business or manage the business activities properly; and unethical behaviour on the part of some beneficiaries. The possible solution suggested by the bank officials to tackle these issues are: mandatory training to prospective beneficiaries, awareness about business

registration formalities, business guidance, etc., which will make them ethical in behaviour.

The beneficiaries were also asked about the difficulties they faced during the process of Mudra loan availability based on certain parameters viz., access (which consists of existing rules and regulation, formalities and procedures, repayment rules and conditions, repayment duration, existing rate of interest), level of funding (which includes receipt as per request, fulfilment of business needs, receipt of loan in stipulated duration, sufficiency of loan) and support services from bank (which includes encouraging role of bank, availability of adequate information from bank, gestation period after loan sanction) on five point Likert scale (1 - Strongly disagree and 5 - Strongly agree). The beneficiaries expressed their satisfaction about the process of obtaining the loan under this scheme at 1% level of significance.

beneficiaries were also asked whether they make regular repayment. There were some beneficiaries who did accept that they could not repay the loans on time due to reasons which include; unfavourable business conditions, sudden crisis at home, unforeseen challenges at family level. But they affirmed that they were not the defaulters and would pay lump-sum.

As far as the issue of obtaining the loan from the bank is concerned, the possibility of applicant resorting to corrupt mean like bribery or using the influence may not be denied. But it is worth mentioning that the beneficiaries strongly contended that they need not have to resort to such tactics for obtaining the loan.

New Process Recommendation for Mudra Loan

Considering the overall performance of Mudra Scheme on the basis of responses of beneficiaries of Mudra loans, an urgent need to revamp the process is felt to ensure the better deliverables. Also, on the basis of literature review of similar funds in other countries, it is found that the training is an essential component before sanctioning of the loan which is missing in Mudra Scheme. Further, the components like mentoring, business plan preparation, marketing and promotion and networking are also needed to be incorporated. The existing process have only two Stages; Stage 1: Apply for Stage 2: Documents verification and loan loan. sanction/rejection. Hence, the new process is recommended which comprise of the four stages which are given as below in figure 1.





Source: *Author's Recommendation*

In **Stage-1**, the prospective beneficiary will apply for loan in bank. In Stage-2, the applicant will be provided the list of training centres to pursue training. The institutes may be empanelled by the banks by signing the Memorandum of Understanding (MoU) with the educational institutes or established industries after verification of infrastructure facilities. The faculties or industry expert from these institutes can act as a trainer and/or mentor for small The institute will provide the certificate of businesses. completion to the prospective beneficiary, which will be submitted in the bank by the candidate. The bank will verify all the documents and complete the documentation part for further process. In Stage-3, the loan amount will be sanctioned and disbursed. If the beneficiary belongs to Shishu Loan category, then the mentor must be allotted. Shishu category loan beneficiaries are the candidates with relatively low profile in terms of educational qualifications

and economic backgrounds. So, they need mentor for guidance in businesses. Kishore category and Tarun category beneficiaries may be offered this facility if it is desired by them as their educational and economic background is better than Shishu category beneficiaries. In Stage-4, all the beneficiaries must be monitored by the bank officials on regular intervals may be in 3 to 4 months and must be asked for progress report. This will help the bank to monitor and understand the difficulties the borrowers are facing in businesses, and recommend for possible solutions. Also, the bank maintaining the information of all the borrowers who borrowed money for business purposes, such information in the form of list and contact details must be shared with the other beneficiaries so that they can develop the networking and create business linkages which will support their businesses to grow.

IV. Conclusion

The PMMY initiative has given the stimulus to start-up and entrepreneurship ecosystem as every member of the society irrespective his educational background can now think of opting for Mudra loan at the primitive stage of the business. One can dream of entrepreneurship by starting initially with meagre capital and then can opt for Mudra loan. It is a need of an hour to take up the start-up and entrepreneurship activities by the members, as the creation of employment opportunities for the youths is the major challenge the government is facing and is likely to aggravate further. Better penetration and more awareness are necessary, so that prospective beneficiaries will not be deprived of this facility. The initiative is contributing significantly in financial and economic inclusion as registration of the business is mandatory to get the loan. The initiative has proved to be a boon for the members of informal businesses as well as other micro and small businesses in real sense as it has significantly raised the standard of living and helped in business expansion. To sum up, the PMMY initiative with new process recommended if implemented, will create better lenderborrower relationship, ensure better monitoring and control, increase industry-institute interaction, scrutinization of genuine borrowers, and require less resources with high gains, will create innovative ideas arising out of student-facultyinteractions and problem-solving entrepreneur fast mechanism.

References

- Agarwal Manish & Dwivedi Ritesh (2017), Pradhan Mantri Mudra Yojana: A Critical Review, Parikalpana – KIIT Journal of Management, Odisha, 97-106. (https://doi.org/10.23862/kiitparikalpana/2017/v13/i2/164524).
- Annual Report 2018-19, Periodic Labour Force Survey (PLFS), Ministry of Statistics and Programme Implementation (MoSPI), Government of India, June, 2020.
- Aravindaraj K & Bala J G (2018), Performance Analysis of PMMY in South India, IOSR Journal of Business and Management (IOSR-JBM), 20(1), 24-28. (https://doi.org/10.9790/487X-2001062428).
- Bank Mandiri. (https://www.bankmandiri.co.id/en/kreditusaha-rakyat-kur-)
- Chaudhary, Manoj Kumar (2018), Economic Planning and Poverty Alleviation Strategies In Nepal, Research Nepal Journal of Development Studies, Year 1st, Issue 2nd, November, 47-61.
- Biswas Bhaskar (2019), Comparative Performance of Pradhan Mantri Mudra Yojana (PMMY) in West Bengal and other select States in India, Parikalpana – KIIT Journal of Management, Vol. 15, Dec., 153-165.

- Financing Technology Entrepreneurs & SMEs In Developing Countries: Challenges and Opportunities, an info Dev publication prepared by Roberto Zavatta Economisti Associati SRL in collaboration with Zernike Group BV Meta Group SRL (June 2008), pg. 41.
- Haris M & Subash T (2018), An Innovative Approach to Small Industries: Pradhan Mantri Mudra Yojana (PMMY), EPRA International Journal of Multidisciplinary Research (IJMR), 4(7), 189-194.
- Informal Economy in South Asia, ILO, 2020.
- International Finance Corporation March 2016 Report. Women-owned SMEs in Indonesia: A Golden Opportunity for Local Financial Institutions, Market Research Study.
- International Labour Organization (ILO) (2014), Notes on Policies for the Formalization of Micro and Small Enterprises in Brazil.
- (https://www.ilo.org/wcmsp5/groups/public/---americas/---rolima/documents/publication/wcms_318209.pdf)
- John J.S, Kabra N & Jose S M (2018), MUDRA Performance in Karnataka, International Journal of Research and Analytical Reviews, 5(4), 278-287.
- Joshi Vikram K. (2017), Appraisal of Pradhan Mantri Mudra Yojana (PMMY) – A Landmark Initiative for Financial Inclusion, International Journal of Applied Business and Economic Research, Volume 15, Number 16 (Part-II), 71-83.
- Kumbhare, Sagar & Kumar, Ujjwal (2016), Impact Assessment of Financial Inclusion schemes. 10.13140/RG.2.1.5154.7762.
- Mahajan, Yogesh (2019), A Study and Review of Pradhan Mantri Mudra Yojana (PMMY) in the State of Maharashtra, International Journal of Advance and Innovative Research, Volume 6, Issue 2(XVII): April-June, 1-7.
- National Commission for enterprises in the Unorganized Sector (NCEUS), 2007, Report on Conditions of Work and Promotion of Livelihoods in the Unorganised Sector, pn 3.
- News Desk, Government to reduce KUR annual interest rate to 6 percent in 2020, The Jakarta Post, Wed, November 13, 2019.
- Olima, G. A. (2016), Influence Of Uwezo Funds on Enhancing Livelihoods Among Youth and Women in Ainabkoi Sub County, UasinGishu County, Kenya.Moi University, Eldoret, Kenya.
- Otieno, A.G. (2014), Factors Influencing Sustainability of the Uwezo Fund Assisted Youth Projects in Migori County, Kenya.Journal of Finance, 7 (3), 183-190.
- PMMY-Annual Report 2019-20. Mudra Bank. (www.mudra.org.in)

- PMMY-Statewise Performance 2015-16, 2016-17, 2017-18, 2018-19 & 2019-20. Mudra Bank. (www.mudra.org.in)
- Report of the National Commission on Labour (1969), Ministry of Labour and Employment and Rehabilitation, Government of India, Chapter XXIX: Unorganised Labour, pn 417.
- Report on Harnessing Resources From The Informal Sector For Economic Development" by Zimbabwe Economic Policy Analysis and Research Unit (ZEPARU) & Bankers Association of Zimbabwe (BAZ), October 2014.

Reserve Bank of Zimbabwe.

(https://www.rbz.co.zw/index.php/regulation-

supervision/regulationsupervision/bankinginstitutions/142-the-small-andmedium-enterprises-development-corporationsmedco)

Rudrawar. A. A., & Uttarwar, V. R. (2016), An Evaluatory Study of MUDRA Scheme, International Journal of Multifaceted and Multilingual Studies, 3 (6).

SABRAE, Small Business Experts.

- (https://www.sebrae.com.br/sites/PortalSebrae/canais_adicion ais/sebrae_english)
- SEFA, Fund. (https://www.sefa.org.za/)
- SEFA Services. (https://www.sefa.org.za/services/product/11)
- Seibel Hans Dieter, Thac Nguyen Tam (2012), Growth and Resilience of Credit Cooperatives in Vietnam, In: Onafowokan O. Oluyombo (ed.), Cooperative Finance in Developing Economics, Lagos, Soma Prints Ltd., pp. 179-197.
- Shahid, M., & Irshad, M. (2016), A Descriptive Study on Pradhan Mantri Mudra Yojana (PMMY), International Journal of Latest Trends in Engineering and Technology, Special Issue, March, 121-125. (https://www.researchgate.net/publication/323829595).
- Shirin, Akter (2016), Overview of Financial Inclusion in Bangladesh, The Journal of the Study of Modern Society and Culture, pn 225-255.

- Shrestha Prithvi Man (2020), Over 70,000 youth entrepreneurs to benefit from reduced interest rate on loans, The Kathmandu Post, 2, August.
- (https://kathmandupost.com/money/2020/08/02/over-70-000youth-entrepreneurs-to-benefit-from-reducedinterest-rate-on-loans)
- Small Enterprise Finance Agency (SEFA)-Funding a future for SMMEs, Leadership Magazine Online, April 17, 2017.

(http://www.leadershiponline.co.za/articles/small-

- enterprisefinance-agency-sefa-21461.html)
- SMEDCO. (https://www.smedco.co.zw/service.html)
- State Bank of Vietnam, Annual Report, 2018.
- System of National Accounts (SNA), 1993. Prepared under the auspices of the Inter-Secretariat Working Group of National Accounts and jointly published by the five organisations. (Commission of the European Communities – Eurostat, IMF, OECD, UN & World Bank). pn 135.
- Uwezo Fund. (https://www.uwezo.go.ke/)
- Uwezo Fund Guide, 2013.
- Venkatesh, J., & Kumari, R. I. (2015), Performance of Mudra Bank: A Study on Financial Assistance to MSME Sector, International Journal of Research in Economics and Social Sciences, 5, 185-191.
- Vijai C (2018), A Study on the Performance of Mudra Yojana in Tamilnadu, Journal of Banking, Finance and Insurance Management, Volume 1 (2), 1-9.
- Youth and Small Entrepreneur Self-employment Fund (YSEF) Secretariat, Ministry of Youth and Sports, Government of Nepal. (http://www.ysef.gov.np/index.php)
- Zeleke Worku (2016), The Impact of Over-Regulation On Small Enterprises, Risk governance & control: financial markets & institutions, Volume 6, Issue 3, Summer 2016, Continued 2, Special issue "Risk Management and Corporate Governance in African Countries", p.n. 12-19.

Shefali Sethi*Anand Sharma**

Purpose: This study contributes to the literature of the novel concept of green insolvency, its growing importance in India and how it can be introduced in the current legislation.

Design/methodology/approach: A meta-analysis of literature has been performed using information which has been supported by publications from various agencies such as Task Force on Climate-Related Financial Disclosures and Oxford Law.

Findings: It suggests that in order to effectively manage the risk of insolvency and bankruptcy, environmental damages should be included in administrative charges, environmental claims should be given priority in liquidation proceedings, concept of insurance fund should be created and incentives in the form of grants in exchange for mandatory climate reporting should be introduced.

Originality/value: By synthesising and assessing the extensive literature on climate change and insolvency risk, the current study offers a fresh perspective which no study in India has done till date.

Keywords: *Insolvency, Bankruptcy, Green, Climate change, sustainability, credit risk* **JEL Classification Code**: *Q01, G33, G38*

The concept of sustainability development has expanded tremendously and been incorporated into many professions in the last two decades. A German forester named Hans Carl von Carlowitz coined the phrase for the first time several centuries ago to describe how forests should be managed over the long term. In the middle of the 1980s, the commission led by former Norwegian Prime Minister Gro Brundtland became the centre of this discussion, which resulted in the groundbreaking report Our Common Future (Development, 1987). This study provided the widely accepted contemporary definition of sustainable development:

"Sustainable development is development that meets the needs of the present without compromising the ability of future generations to meet their own needs"

Over the past few years, the discussion of climate change has moved into the mainstream as a result of increased public awareness, frequent extreme weather occurrences, and mounting scientific data. Climate change provides a serious threat to ecosystems and habitats, and as solvency risk increases, it also offers a challenge to the existence of enterprises. Businesses are particularly susceptible to the consequences of climate change if they work in the energy, transportation, agricultural, or forestry industries. Natural resources are important inputs for a variety of industries, and their depletion or scarcity can have a negative impact on the production processes of these industries, leading to higher costs and supply disruptions. At the same time, physical risks to the survival of businesses are posed by extreme weather conditions or natural disasters like floods, storms, wildfires, etc. Changes in climate patterns, such as persistent temperature increases, may eventually damage some enterprises' production activities. Certain companies, like oil conglomerates, are now facing massive litigation costs as more lawsuits are being filed against enterprises, holding them responsible for damages owing to environmental damage that may have been caused by their acts or omissions. Businesses that do not adapt to climate change, "including enterprises in the financial system, will go bankrupt without question, and yet there are tremendous riches to be earned," as former Governor of the Bank of England Mark Carney famously said, "will go bankrupt without question." This study, the first of its type, examines the green insolvency regime's underlying principles and how they interact with the Insolvency and Bankruptcy Code. The paper's main goal is to establish justifications for the nation's development of a climate-sensitive insolvency and bankruptcy law.

I. Review of Literature

Studies already conducted concentrate on the relationship between climate change and an economy's financial stability. According to (Gangi et al., 2020) environmental damage and climate change provide serious financial concerns and endanger economic viability. According to an American study (Saha and Viney, 2020) hurricanes and flooding might cause the US to lose more than 10% of its GDP. Due to the fact that businesses in a wide range of industries rely on the natural environment for vital resources, the state of the environment is deteriorating, which has led to a number of problems(Wagner, 2010). Economic experts have long warned of the negative impacts of climate change on the economy and the possibility of a financial collapse brought on by the explosion of the carbon bubble (Institute for Public Policy Research,2012).

* Research Scholar, Central University of Haryana

** Prof., Dean of Students Welfare, Central University of Haryana The flip side is the switch from an economy based on carbon to one that is predominantly fuelled by renewable resources results in a kind of danger known as transition risk which can have an impact on the global capital markets by slowly changing the price of commodities, corporate bonds, and stocks (Rudebusch, 2021). Another risk is the cost of complying to environmental regulations which can negatively impact its profitability by raising production costs and lowering productivity. For instance regulations requiring more stringent air controls may cause manufacturing plants to produce less than 5%, costing the industry as a whole USD 21 million annually (Greenstone et al., 2012).

Risk to businesses

Corporate community is becoming aware of the risk posed by climate change. According to the Australian Bureau of Agricultural and Resource Economics, farmers have lost more than AUD \$1 billion due to the changing climate in 20 years mostly due to drought (Gooday&Hughes, 2021). Similarly, almost all U.S. ski resorts could have a 50% shorter season by 2050 and up to an 80% shorter season by 2090 as a result of decreasing snowfall brought on by rising temperatures. As a result of wildfires making popular hiking destinations dangerous, the United States, Turkey, and Australia are losing money from tourism (Wobus et al., 2017). India is most vulnerable to climate change as it has one of the highest disaster risk levels in the world, ranking 38th out of 191 nations in the 2022 (Inform Risk Index, 2022). This is due to the country's significant susceptibility to disasters like floods, tropical storms, and drought. According to a study services, manufacturing, retail, tourism, construction, and transport would be the most affected industries, with these industries losing an average of more than USD 1.5 trillion in value added to the GDP year by 2070 (Deloitte Economics Institute,2021). Particularly susceptible to risk is the MSME sector which constitutes large portion of business in India and lacks resources to integrate climate risk (Gulati, 2001). On the other hand (Macwilliams et al., 2019) in their report looked at market variables to determine investor expectations for exposure to climate risk and expected cost distribution. They witnessed that investors think that the cost rises brought on by climate change won't have a significant influence on the present value of their investments until a very long time from now.

Hence, according to (Gonenc & Scholtens, 2017) businesses that actively adopt and incorporate pro-environmental strategies to reduce negative externalities would increase their market-competitive edge and assure their continued existence. A company's participation in corporate environmentalism may lower its risk of insolvency in a variety of ways because it influences decisions about the company's goods and services, consumer purchasing and brand engagement, as well as stakeholders' perceptions(Rahman et al., 2021). Additionally, corporate environmentalism lessens the danger to a company's reputation, which is the single biggest risk a company can face(Gasbarro et al., 2017). (Cai et al., 2016) using a sample of US businesses from 1990 to 2012, have examined the relationship between firms' Corporate Environmental Responsibility and risk and discovered convincing evidence that enhancing environmental performance lowers financial risk for businesses.(Aziz et al., 2021) revealed that businesses with strong green performance will see a decreased chance of bankruptcy as their z score rises.

Prior to recently, the intersection of bankruptcy law and environmental law did not receive enough attention. Business Enabling Environment (BEE), a new indicator developed by the World Bank to measure the business and investment climate of nations which examines good environmental regulatory practises during insolvency proceedings and addresses environmental obligations in bankruptcy is an effort in this direction. (Zou et. al, 2015) revealed that in order to comply with a 50% decrease in carbon footprint, the European Central Bank dropped a number of issues from its portfolio. Similarly, the central banks in France and Sweden have adjusted their investment strategies to take climate risk exposure into consideration after accepting the inescapable fact. Both domestically and abroad, the number of cases are rising claiming that corporations who have not appropriately disclose the dangers of climate change are guilty of securities fraud because these risks have an impact on stock prices. The plaintiffs in these lawsuits contend that climate change concerns are "material facts" that should have been disclosed because investors base their decisions on the financial value of the assets held by the corporation (Byrnes&Setzer,2019).

Objectives

- To examine growing importance of Green Insolvency in India followed by case study analysis.
- To analyse the current scenario of Green Insolvency in India.
- To recommend policy changes that could be made to the existing legislation.

II. Research Design and Methods

In order to respond to the research questions and present a case for an Indian green insolvency regime, numerous secondary sources have been used. To back up the claims and provide evidence, reports from eminent research and insights teams at multinational corporations (such KPMG and Deloitte) have been employed. A meta-analysis of the literature has been conducted to provide the backdrop for India's green insolvency policy using publications from the Task Force on Climate-Related Financial Disclosures (TCFD), OECD, Oxford Law, and British Columbia. Meta-analysis is the process of using statistical tools to compile the findings of various independent investigations. A meta-analysis does supplementary statistical analysis on the

findings of comparable research in addition to critique and integration. Empirical data on economic and financial losses attributable to climate change has been included which provides evidence to support the research statements such as the report of the Climate Disclosure Standard Board (CDSB) describes how climate change might result in financial hazards when combined with poor corporate governance. Similarly, according to a study by International Water Management Institute, 2007 by 2025, it is predicted that Indian businesses will more than triple their water consumption. The price and productivity of highly water-intensive businesses would be negatively impacted by the dwindling fresh water supply and rising demand. Also, according to a recent survey, Indian businesses calculated that the overall intrinsic financial impact of climate risks was \$7,138 billion, with an average risk of \$92 billion per business (CDP India Annual Report 2020).

III. Results and Discussion

Growing Importance of Green Insolvency in India

Over the past few years, the discussion of climate change has moved into the mainstream as a result of increased public awareness, frequent extreme weather occurrences, and mounting scientific data. According to the National Centers for Environmental Information, which monitors weather and climate events in the United States, 2017 was "a historic year of weather and climate disasters," with three tropical cyclones, eight severe storms, two inland floods, a crop freeze, drought, and wildfires contributing to a total cost of more than \$300 billion. (NOAA National Centers for Environmental Information ,2022). The Reserve Bank of India noted that "the immediate policy issue is to jump-start the global economy" in a paper titled Green Finance in India: Progress and Challenges published in January 2021(RBI, 2021). (World Economic Forum, 2019) categorised extreme weather, inadequate climate change adaptation, and natural catastrophes as the top three threats, in that order, according to the "probability" that they will occur. According to the United Nations Environment Programme, if things continue as they are, global temperatures could increase by more than 3°C this century. Such a spike in temperature has the potential to destroy economies, stymie trade, halt development, and drive more people into poverty (United Nations, 2022). By 2050, the global GDP is expected to have been reduced by up to 18% and will have lost an estimated 10% of its current worth (Swiss Re Institute, 2021). For a nation like India, where the government hopes to nearly quadruple the size of the economy to \$10 trillion by 2030 from roughly \$2.75 trillion at present, this magnitude of loss in economic value is not promising. The goal of generating net-zero emissions by 2070 was pledged by Indian Prime Minister Narendra Modi during the brainstorming session held at COP26 in Glasgow last year. He also announced an ambitious goal for India to develop 500 GW of energy from non-fossil sources and cut carbon intensity to 45% by 2030 as part of the five-pronged commitment "Panchamrit." In addition, the prime minister promised to reduce anticipated carbon emissions by 1 billion tonnes from 2021 to 2030 (www.pmindia.gov.in, 2021).

The Green Future Index 2022 is the second annual comparative evaluation of 76 countries and territories based on their capacity to create economies and society that are sustainable and low-carbon. India is ranked currently 42nd as compared to its previous ranking of 21 in the year 2021. It has been categorised as climate laggards which means it is making slow and uneven progress towards building a green future. Although, India continues to be one of the world's most active investors in renewable energy generation (solar, especially), and has started to make firmer policy commitments to decarbonization. Plans call for India's state coal companies to install 5.56 gigawatts of renewable power generation capacity by 2030(Singh, 2021) Below is the figure showing Green Future Index country ranking,2021-22:





Source: MIT Technology Review Insights, 2022

Similarly, global nations are evaluated according to the Environmental Performance Index based on indicators like waste management, air quality, biodiversity & habitat, fisheries, ecosystem services, and climate change. India is ranked 180 with EPI score of 18.9 which is lower than even Bangladesh, Myanmar, Pakistan, and Vietnam. The nation has also performed poorly on several metrics, including government performance, rule of law, and corruption control. The chief of the UN Development Program has warned that more than 50 of the world's poorest developing nations face the threat of going into default on their debt and effectively going bankrupt unless the developed world provides immediate assistance.

The impact of climate change is becoming a significant consideration for businesses. Climate change will now impact

a company's prospects for resolving its insolvency if it were to start the insolvency resolution procedure. For instance, a resolution applicant would consider a firm's present and potential carbon footprint for technological innovation and the implementation of energy-efficient operations to minimise it. The value of the company as a continuing concern could be negatively impacted by the reputational harm resulting from insolvency brought on by climate change.

Case studies

Pacific Gas &Electric Case study

The utility PG&E, which is based in California, submitted a second modified plan of reorganisation at the beginning of 2020. In response to more than \$30 billion in litigation claims made against it for its claimed participation in starting California wildfires, PG&E filed for Chapter 11 bankruptcy. It is termed as the first climate change bankruptcy in the wake of wildfires that engulfed its service region in 2017 and 2018. The severe weather, which was exceptionally hot and dry and led to more frequent and severe fires, was largely to blame for the massive damage. The conditions, according to the recently fired CEO of PG&E, were brought on by climate change and global warming and many experts have agreed (Macwilliams J. et. al., 2019). The spill over of such climate induced bankruptcy is:

1.PG&E's stock fell 85%, causing more than \$20 billion in losses for shareholders.

2. Huge expenses will be incurred by insurers as a result of losses in every type of insurance imaginable, including life, health, property, disability, and others.

3. During the next three years, customers, who now pay the second-highest rates in the nation, will see yearly increases of 12 to 24%. Rate rises are expected to continue for many years. The example of PG&E's bankruptcy shows how financial stress brought on by climate change can cause businesses to declare bankruptcy. The bankruptcy-related effects of climate change lead to higher debt costs and a decline in stock prices.

The Redwater Case

In the case of Orphan Well Association v. Grant Thorton Ltd. (also known as Redwater) an Albertan oil and gas firm that owned more than a hundred wells, pipelines, and facilities, filed for bankruptcy. The laws governing how oil and gas firms operate in Alberta also mandate that they "abandon" their pipelines and wells and "reclaim" or "restore" the land. The Supreme Court of Canada decided this matter and held that the trustee could not disclaim the site in favour of the Alberta Energy Regulator. Additionally, it decided that Redwater's environmental commitments could not be disregarded, regardless of whether they were funded by their own funds. Prior to settling any other claims, the environmental responsibilities resulting from reclamation and abandonment were to be met. This is a significant ruling since environmental commitments are not given top consideration in insolvency cases.

Current Scenario in India

In India, there have been numerous instances of commercial interruptions brought on by climate change. The following are some examples:

- Over the course of 2014–2015, India had dry spells that had a negative impact on enterprises across a variety of industries.
- Due to the drought and warmer temperatures in 2016, there was a 12% decrease in tea production in southern India.
- Due to a lack of available water for cooling, a coal power station in West Bengal had to shut down for ten days.
- The catastrophe in Uttarakhand lead to loss of innumerable assets and it will be difficult to reconstruct many of them.

The PHD Chamber of Commerce estimates that lost potential tourism costs have reduced the gross state product by 11%. The cost of the lost ecosystem services is incalculable, and the hundreds of billions of rupees now needed to rebuild something that was largely avoidable could have been used in other ways. Climate change risks are predicted to have a catastrophic financial impact of 7,138 billion (about \$100 billion) on Indian enterprises in the next five years (CDP India Annual Report, 2020). According to a study, the majority of coal plants in India only ran at 60% of their full potential and were at risk of going out of business soon. Only 36% of Indian businesses disclosed their greenhouse gas emissions in 2016, which is a very low number for a nation that is home to some of the world's most polluted cities (Down to earth, 2018). Large Indian firms are already being charged with intentionally concealing facts about their environmental impact by a significant number of investors from around the world (Sengupta.2020).

Both voluntary and compulsory disclosures are still in their infancy in India. Some SEBI disclosure requirements such as Regulation 24 and 70 of the SEBI (Issue of Capital and Disclosure Requirements) Regulations, 2018 mandated disclosure of material information and Schedule VI of the ICDR regulations mandates disclosure of material risk factors but it is not clear whether climate change disclosures come within its ambit. The "Voluntary Guidelines on Corporate Social Responsibility" were introduced by MCA, who urged businesses to publicly acknowledge their contributions to society. The National Voluntary Guidelines on Social, Environmental, and Economic Responsibilities of Business ,2011 came after this. Environmental, social, and governance (ESG) reporting has accelerated significantly in India and investor demand for these disclosures has increased, especially in light of the epidemic. In 2011the UN-backed Principles of Responsible Investment (PRI), "a global network of investors that tries to integrate ESG norms into investment processes," was signed by Kotak Mutual Funds, the first asset management firm (Souvik, 2020). This will motivate businesses to develop sustainable practices that will benefit them in long run and have a wider impact. While it

undoubtedly promotes openness for all parties involved, a firm's management of the climate risk is the only information provided, not the degree to which firm is exposed to the risk.

Policy Recommendations

Following policies must be adopted by the Government and industry:

- 1. The relationship between climate change and insolvency policy needs to be investigated in order to effectively manage the risk of insolvency and bankruptcy as well as to minimise and manage the loss of asset value. Environmental damages made by government agencies are not considered operational debt. Such claims should be included in the administrative charges or costs of the insolvency resolution procedure. A business should not be able to obtain a competitive edge over rivals and escape compliance with environmental rules by simply declaring bankruptcy.
- 2. Environmental authorities are currently not considered operational creditors under the Insolvency and Bankruptcy Code, 2016. Environmental claims are usually in the form of fines, penalties which are mostly unliquidated due to vast maiority of environmental laws and insolvency professionals are not quantifying correct value of these claims as seen in the case of Essar steel vs Satish kumar gupta & ors(Essar Steel India Limited v. Satish Kumar Gupta and Ors, 2020). In an effort to make a firm accountable for environmental harm, such procedures may be detrimental. Environmental claims should be given priority even before secured creditors. This will promote the larger objective of sustainable environmental goals. Moreover, taking this action could ultimately improve the credit market by allowing financial creditors to make environmental compliance a required prerequisite in the future, in line with The Equator Principles.
- 3. The government ought to offer incentives in the form of grants or subsidies that aid in corporate reorganisation. The policies should include mandatory climate reporting as a condition of receiving new investments or resurrection. Firms that undergo restructuring or revival based on climate-sensitive principles can also raise more money and maintain themselves as a "going concern."
- 4. When dealing with insolvencies brought on by climate change, insolvency experts will need to set up temporary financing, manage a variety of claims resulting from sudden incidents or long-term climate effects, and maintain effective corporate governance, including climate change disclosures.
- 5. Before approving the resolution or liquidation of a corporation, the courts that handle bankruptcy cases would also need to consider how public policies and regulatory frameworks are changing in response to climate change. This would enable an appraisal of the company using the ESG criteria and provide fresh investment prospects.

- 6. Government should create insurance legislation against catastrophes, much like California did in the PG&E case. A safety certificate for the utility should be issued, and an advisory board should be established to assess the utility's unique safety criteria. To access the insurance fund, the issuance of this certificate would be necessary. This encourages utilities to take appropriate safety precautions and switch to more environmentally friendly methods if they want to be saved in the event of a climate changerelated catastrophe.
- 7. To finance the switch to low-carbon manufacturing and to finance climate resilience, a wide range of financial instruments would need to be created. Companies should be required to make climate-related disclosures to reduce the information asymmetry in securities market. The task force on Climate change related financial disclosures(TFSD) has developed consistent climate-related financial risk disclosures for use by companies, banks, and investors in providing information to stakeholders. Similar to New Zealand and UK, India should mandate such disclosures.
- 8. For the benefit of financial institutions and investors, numerous ratings organisations, data companies, and actuaries would also need to create a new set of criteria to gauge the resilience of businesses to climate change threats.

Discussion

The relationship between insolvency laws and sustainability research has not been explored till now. By synthesising and assessing the extensive literature on climate change and insolvency risks, a case is made in favour of a climatesensitive insolvency regime. It draws attention to the need for a "green insolvency regime" in India by presenting strong justifications and supporting data from around the globe. particularly from India. The study suggests that going forward environmental damages should be included in the administrative charges, environmental claims should be given priority in liquidation proceedings, concept of insurance fund should be created and incentives in the form of grants in exchange for mandatory climate reporting should be introduced. Courts that handle bankruptcy should consider how public policies and regulatory frameworks are changing in response to climate change and make climate change related financial disclosures mandatory.

IV. Conclusion

There is a vast body of economic literature suggesting that climate change may trigger the next bankruptcy crisis and it is past time for regulators to intervene to prevent a financial catastrophe. Even while climate change will bring many difficulties for businesses in the future, there are also chances for them to become more competitive and discover new market opportunities. Businesses can gain an advantage in their industry by developing new, less carbon-intensive products and services as a result of climate change. The current regulatory framework in India gives SEBI plenty of room to take legal action against companies that fail to disclose climate change concerns. Amendments in the Insolvency law can go a long way in changing the way society perceives business failures and promote the objective of sustainable businesses. Although PG&E's bankruptcy may have been the first and only one related to climate change, it is undoubtedly not the last. Businesses may experience solvency issues if they don't act quickly to incorporate climate change disclosures and preparedness into their business strategies. Action on climate change is urgently needed to protect human lives and livelihoods, the environment, and the economy.

References

Aziz, S., Rahman, M., Hussain, D., & Nguyen, D. K. (2021). Does corporate environmentalism affect corporate insolvency risk? The role of market power and competitive intensity. *Ecological Economics*, 189, 107182.

https://doi.org/10.1016/j.ecolecon.2021.107182

- Byrnes.R&Setzer.J.(2019). Global Trends in Climate Change Litigation.
- "Building Back Greener India Inc. Demonstrates Climate Resilience", CDP India Annual Report 2020.
- Cai, L., Cui, J., & Jo, H. (2016). Corporate Environmental Responsibility and Firm Risk. *Journal of Business Ethics*, 139(3), 563–594. https://doi.org/10.1007/s10551-015-2630-
- Committee of Creditors of Essar Steel India Limited (through authorized signatory) v. Satish Kumar Gupta and Others, (2020) 8 SCC 531
- Discussion paper on "Facing the Impacts of Climate Change: Indian MSMEs and Adaptation" published by Deutsche Gesellschaft fur Internationale Zusammenarbeit GIZ) GmbH in collaboration with Small Industries Development Bank of India (SIDBI)
- Down to earth. (2018.5th June). India is not making businesses disclose climate risks. www.downtoearth.org.www.downtoearth.org.in/new s/climate-change/in-the-last-one-year-india-has-notmade-businesses-disclose-climate-risks-report-60765
- Gangi, F., Daniele, L. M., & Varrone, N. (2020). How do corporate environmental policy and corporate reputation affect risk-adjusted financial performance? *Business Strategy and the Environment*, 29(5), 1975– 1991. https://doi.org/10.1002/bse.2482
- Greenstone, M., List, J.A., & Syverson, C. (2012). The effects of environmental regulation on the competitiveness of U.S manufacturing. *National Bureau Of Economic Research Working Paper 18392*, 1–33.

- Gooday, P., & Hughes, N. (2021). Climate change impacts and adaptation on Australian farms. ABARES.
- Gasbarro, F., Iraldo, F., & Daddi, T. (2017). The drivers of multinational enterprises' climate change strategies: A quantitative study on climate-related risks and opportunities. *Journal of Cleaner Production*, 160, 8–26. https://doi.org/10.1016/j.jclepro.2017.03.018
- Gonenc, H., & Scholtens, B. (2017). Environmental and Financial Performance of Fossil Fuel Firms: A Closer Inspection of their Interaction. *Ecological Economics*, 132, 307–328. https://doi.org/10.1016/j.ecolecon.2016.10.004
- Gouzoules A. (2021), "Going Concerns and Environmental Concerns: Mitigating Climate Change Through Bankruptcy Reform", Boston College Law Review, Vol. 63
- Gulati, M. (2001), "Improving Environmental Compliance Through Mandatory Disclosure - A Home Grown Model For Indian SMEs", Indian Society for Ecological Economics.

https://www.cdp.net/en/info/about-us

- https://www.ippr.org/files/images/media/files/publication/201 2/09/complex-new- world_Aug2012_web_9499.pdf
- Institute for Public Policy Research.(2012). Complex New World: Translating new economic Thinking into public policy.
- International Water Management Institute (IWRM) (2007), "India's Water Future to 2025 – 2050: Business-as Usual Scenario and Deviations"
- Inform Risk Index 2022, Disaster Risk Management Knowledge Centre, European Commission.
- India's Turning Point: How Climate Action Can Drive Our Economic Future", Deloitte Economics Institute, August 2021.
- Macwilliams.J.J.,Lemonaca.S &Kobus.J.(2019). PG&E: Market And Policy Perspectives On The First Climate Change Bankruptcy.Centre On Global Energy Policy
- Michael Greenstone; John List and Chad Syverson, (2012), *The Effects of Environmental Regulation on the Competitiveness of U.S. Manufacturing*, No 18392, NBER Working Papers, National Bureau of Economic Research, Inc
- National Statement by PM at COP26 Summit in Glasgow(2021). www.pmindia.gov.in. Retrieved from

https://www.pmindia.gov.in/en/news_updates/nationa l-statement-by-pm-at-cop26-summit-in-glasgow/

NOAA National Centers for Environmental Information (NCEI) U.S. (2022). *Billion-Dollar Weather and Climate*

Disasters. https://www.ncei.noaa.gov/access/billions

Orphan Well Association v. Grant Thornton Ltd., 2019 SCC 5 (CanLII), [2019] 1 SCR 150, <https://canlii.ca/t/hx95f>, retrieved on 2023-02-22

- Rahman, M., Rodríguez-Serrano, M. Á., & Faroque, A. R. (2021). Corporate environmentalism and brand value: A natural resource-based perspective. *Journal* of Marketing Theory and Practice, 29(4), 463–479. https://doi.org/10.1080/10696679.2021.1872387
- RBI. (2021). Green Finance in India: Progress and Challenges. *RBI Bulletin*, 61–71. https://doi.org/10.52711/2321-5828.2021.00039
- Report of the World Commission on Environment and Development: Our Common Future. (1987). https://sustainabledevelopment.un.org/content/docum ents/5987our-common-future.pdf
- Rudebusch, Glenn D. (2021). Climate Change Is a Source of Financial Risk. FRBSF Economic Letter 2021-03(February 08)
- Saha, S., and Viney, B. (2019). "How Climate Change Could Spark the Next Financial Crisis." Journal of International Affairs
- Sengupta,R.(2020,June 10).Investors accuse 37 Indian companies of hiding environmental impact information.www.downtoearth.org.in/blog/environm ent/investors-accuse-37-indian companies-of-hidingenvironmental-impact-information-71670
- Singh,R. (Nov. 19, 2021). India's state coal companies target installing 5.56 GW renewable power by 2030.*S&P Global Commodity Insights*. https://www.spglobal. com/platts/en/market-insights/latest-news/energytransition/111921-indias-state-coal-companies-targetinstalling-556-gw-renewable-power-by-2030.

Souvik, G. (2020, September 6). Covid Disruption: ESG Investment catches on fast and is here to stay, *The Economic times*. https://economictimes.indiatimes.com/markets/stocks /news/covid-disruption-esg-investment-catches-onfast-is-here-to-

stay/articleshow/77958618.cms?from=md

Swiss Re Institute (2021). The economics of climate change:no action not an option. Retrieved from www.swissre.com:

www.swissre.com/institute/research

- Trumpp, C., & Guenther, T. (2017). Too Little or too much? Exploring U-shaped Relationships between Corporate Environmental Performance and Corporate Financial Performance. *Business Strategy and the Environment*, 26(1), 49–68. https://doi.org/10.1002/bse.1900
- United Nations (2022). Tackling Climate Change. Retrieved from www.un.org: https://www.un.org/sustainabledevelopment/climateaction/
- Wobus, C., Small, E. E., Hosterman, H., Mills, D., Stein, J., Rissing, M., & Martinich, J. (2017). Projected climate change impacts on skiing and snowmobiling: A case study of the United States. Global Environmental Change, 45, 1-14.Climate Disclosures Standards Board, "Developing Climate-Resilient Stock Markets".
 - Wagner, M. (2010). The role of corporate sustainability performance for economic performance: A firmlevel analysis of moderation effects. *Ecological Economics*, 69(7), 1553–1560. https://doi.org/10.1016/j.ecolecon.2010.02.017
 - World Economic Forum (2019). *The Global Risks Report* 14th Edition. https://www3.weforum.org/docs/WEF_Global_Risks Report 2019.pdf
- www.frbsf.org/economic-research/publications/economicletter/2021/february/climate-change-is-source-offinancial-risk/
- Zou,S., Morel,R., Spencer,T., Cochran,I., & Colombier,M. (2015), "Mainstreaming Climate Change into Financial Governance", CIGI Policy Brief.

A STUDY OF WORKPLACE SPIRITUALITY AND JOB COMMITMENT WITH THE MEDIATION OF OCCUPATIONAL STRESS

Archana Mittal* Geeta Mishra** Sunita Bishnoi***

Purpose: This research aims to look into the direct and indirect links between employees' spirituality and their organizational commitment.

Design/methodology/approach: Occupational stress has been examined as a relationship mediator. We collected data from 447 respondents by means of a structured survey. The data were analyzed using the SmartPLS version 3.3.2 partial least square structural equation modeling.

Findings: Workplace spirituality affected the stress of the employees negatively and the three forms of commitment i.e., affective, continuance, and normative commitment positively. Occupational stress has a detrimental influence on all categories of commitment and plays the role of mediator between spirituality and affective type of commitment.

Originality/value: This work is critical for the HR departments of IT organizations to become more familiar with the spiritual side of their staff and to develop timely, suitable policies to increase their commitment to the organization.

Keywords: Workplace spirituality, commitment, structural equation modeling, PLS-SEM, mediation, stress **JEL Classification:** M50

In organizational psychology, one of the most essential and studied variables is organizational commitment. The importance of organizational commitment is critical because it leads to a number of favorable organizational outcomes once it is developed. This research focuses on spirituality-related situational variables i.e., workplace spirituality as an antecedent of organizational commitment.

This paper examines the direct impact of workplace spirituality on work-related stress and all three dimensions of commitment as well as the indirect impact of workplace spirituality on commitment via occupational stress among Indian IT employees. Workplace spirituality literature lacks the studies on mechanism through which it attains its objectives, this study fills this gap by making an effort to test occupational stress as a mediator between spirituality at the workplace and organizational commitment.

This document is organized as follows to accomplish the objectives. We begin by reading the literature on commitment, spirituality, and stress and then forming hypotheses. Then, we'll go through the research instruments and procedures that were employed in this research. Then, the results of the survey will be presented, which include measurement model assessment as well as structural model assessment. Finally, we look at the findings, implications in practice, and the research's limitations.

I. Review of literature

Organizational Commitment

An employee's organizational commitment describes the nature of their relationship with their employer and minimizes the possibility of them leaving the organization (Allen and Meyer, 2000). Allen and Meyer, (1990) define three types of

commitment in organizations: affective commitment. continuance commitment, and normative commitment. Affective commitment entails how much a person identifies with the organization, gets involved, and enjoys being a part of it (Meyer & Allen, 1991). When an employee believes that there are no other options but to stay with the existing company, he or she develops a continuance commitment. If the major thread that binds individuals to their organizations is their resolve to continue, this attachment may result in subpar work performance (Allen and Meyer, 2000). A person's moral obligation to stay in the group is known as normative commitment. Employees who exhibit normative commitment believe they are obligated to stay for a variety of reasons, including obligation, common needs, and organizational socialization (Meyer & Allen, 1991).

Workplace Spirituality

Workplace spirituality is the "recognition that employees have an inner life which nourishes and is nourished by meaningful work taking place in the context of a community" (Ashmos & Duchon, 2000, p.137).

* Research Scholar, Amity College of Commerce and Finance, Amity University, Noida, India.

** Associate Professor, Amity College of Commerce and Finance, Amity University, Noida, India

*** Associate Professor, DAV Institute of Management, Faridabad, India Workplace Spirituality and Organizational Commitment According to Rego and Cunha, (2008), all three dimensions of organizational commitment were found to have good relationships with workplace spirituality.

A higher level of workplace spirituality leads to greater normative and affective commitment, but lower levels of continuance commitment (Jurkiewicz and Giacalone, 2004; Milliman et al., 2003; Giacalone and Jurkiewicz, 2003).

Based on the present literature, the study's hypotheses include the following:

- H1(a).Workplace spirituality positively affects affective commitment
- H1(b).Workplace spirituality negatively affects continuance commitment
- H1(c).Workplace spirituality positively affects normative commitment

Occupational stress

Motowidlo, Packard & Manning, (1986, p. 618) outline stress as "An unpleasant emotional experience connected with aspects of fear, dread, anxiety, irritation, annoyance, anger, sadness, grief, and despair".

Workplace Spirituality and Occupational Stress

According to research, workplace spirituality and work stress have a detrimental link (Jalan, S., & Garg, N., 2021; Saxena, A., Garg, N., Punia, B. K., & Prasad, A., 2020; Daniel, 2015). The following hypothesis has been developed for the investigation based on the current literature:

• H2. Workplace spirituality negatively affects occupational stress

Occupational Stress and Organizational Commitment

Saadeh, I. M., & Suifan, T. S., (2019) find a negative relationship between stress and commitment. Job stress has been found to have a favorable link with continued commitment but not with affective commitment (Lambert, Hogan & Keena, 2015). Surprisingly, while job ambiguity had no effect on continuance commitment, role conflict did. Workplace stress has bad and severe consequences on affective and continuance commitment (Cicei, 2012). Based on the present literature, the following hypotheses have been proposed:

- H3(a).Occupational stress negatively affects affective commitment
- H3(b).Occupational stress positively affects continuance commitment
- H3(c).Occupational stress negatively affects normative commitment

Given that the study suggests a clear correlation between workplace spirituality, job stress, and commitment, it's conceivable that job stress acts as a mediator, influencing the association between spirituality and commitment. As a result, the study proposes that when a person's spiritual level is really high, he or she perceives less stress at work, which increases his or her commitment to the organization. As a result, the hypotheses listed below are advanced.

- H4(a).Occupational stress mediates spirituality in the workplace and affective commitment
- H4(b).Occupational stress mediates spirituality in the workplace and continuance commitment
- H4(c).Occupational stress mediates spirituality in the workplace and normative commitment

Theoretical background

Self-determination theory and social exchange theory help explain the link between workplace spirituality and organizational commitment. According to the theory of selfdetermination (Ryan and Deci, 2000), autonomy, competence, and relatedness are three psychological requirements of human beings. They are intrinsically and naturally motivated to achieve these essential innate psychological needs. Workplace spirituality is linked to the self-determination theory's autonomous need, which drives people's desire for meaningful work and alignment with company ideals. People seek challenges and activities that are most suited to their abilities as a result of their quest for competence. As a result, higher adaptive capacity and delight, which come with competence, are developed, activating the workplace spirituality component of meaningful work. The desire to be connected is similar to the sense of community that is part of workplace spirituality, and it could be considered one of the main reasons why people strive for a sense of belonging. The theory of social exchange is a psychology theory that aims to explicate the "social elements that influence a reciprocal relationship's interaction". People measure a relationship's overall worth by subtracting its costs from the benefits it gives. An employee and the organization have a relationship in an organizational setting, and the value of the relationship for both parties will influence its fate. Employees would feel obligated to respond to the source of treatment in a kind, whether through positive attitudes or behaviors which may be in the form of enhanced commitment if they are treated favorably by the organization (Blau, 1964).

Figure1: Proposed Conceptual Model



II. Research Design and Methods

Data Collection and Sample

The study's participants were employees from IT companies all around India. A convenience sample method was used to conduct the current inquiry. To widen the scope of our survey, we used Google Surveys, LinkedIn, and personal e-mails to collect responses. There were 483 responses received in total, with 447 being finalized for analysis due to 36 incomplete/incorrect responses. 239 men were among the total number of responders. The responders' average age is 35.17 years old, and they've spent an average of 9.52 years with the organization. To achieve the goals, a descriptive research approach was adopted.

Measures: Workplace spirituality: Workplace spirituality was measured through a structured questionnaire used by Milliman et al., 2003. The measure looked at three components of workplace spirituality: meaningful work, value alignment, and a sense of belonging.

Meaningful work- The meaningful work scale includes six components from Ashmos and Duchon's, (2000) "meaningful work" scale.

Sense of community- Milliman et al., (2003) developed this scale, which consists of seven components.

Alignment with organizational values- This scale is made up of eight components from Ashmos and Duchon's, (2000) spirituality scale.

Occupational stress: Occupational stress was assessed using a Motowidlo et al., (1986) structured questionnaire. On a spectrum of one to seven, one represented "strongly disagree" and seven represented "strongly agree". Organizational commitment: Allen & Meyer, (1990) had previously designed and validated an instrument to assess organizational commitment. Affective commitment, continuation commitment, and normative commitment were the three dimensions of commitment on the scale consisting of 8 statements each. Respondents ranked their commitment levels on a seven-point Likert scale, from one ("strongly disagree") to seven ("strongly agree").

III. Results and Discussion

For the purpose of reporting results, PLS-SEM employs a two-phase strategy. The validity and reliability of the elements of workplace spirituality were determined in the first phase, and the measuring model or outer model was reviewed. Then the latent variable scores of these elements were used to measure the construct of second order, i.e., workplace spirituality. During the second phase of the study, the effect of a second-order composite of workplace spirituality was evaluated on occupational stress and all three dimensions of commitment.

Measurement model

First order CFA

The internal consistency of a measuring instrument determines a construct's reliability, which ensures its accuracy and precision. When a dependable metric is used several times, the findings are consistent (Hair, Anderson, Tatham, and Black, 1998). The value of reliability measures i.e., Cronbach's Alpha, Dijkstra and Henseler's rhoA, and Composite Reliability (CR) exceeded the 0.70 thresholds (Hair, Howard & Nitzl, 2020; Ali, Rasoolimanesh, Sarstedt, Ringle, & Ryu, 2018). The constructs' convergent and discriminant validity are used to determine validity. Convergent validity exists when many items assessing the same construct are significantly associated, showing item convergence (O'Leary-Kelly and J. Vokurka, 1998). Convergent validity was tested using factor loading and average variance extracted (AVE). Loadings of more than 0.708 and a value of AVE between 0.50- 0.95 are recommended Ringle, (Hair, Sarstedt & 2019; Diamantopoulos, Sarstedt, Fuchs, 2012; Drolet and Morrison, 2001). The internal reliability and convergent validity results of all the items were appropriate except one item of normative commitment (NC7) which had a factor loading of 0.44. As the loading was much below the threshold value and was also affecting AVE, the item was removed and then the model was run again. Outcomes are described in Table 1.

Construct	Item	Туре	Cronbach's Alpha	rhoA	CR	Factor loading	AVE
Meaningful Work	MW1	Reflective	0.884	0.893	0.909	0.796	0.560
	MW2					0.816	
	MW3					0.800	
_	MW4					0.844	
	MW5					0.704	
	MW6					0.729	
Alignment of Values	AV1	Reflective	0.895	0.900	0.917	0.709	0.583
	AV2					0.729	
	AV3					0.675	

Table 1: Outcomes of the Measurement Model Evaluation (First stage)

	A 1/4					0.616	
	AV4 AV5					0.616 0.828	
	AV6 AV7					0.836	
g (g ')	AV8	DC	0.972	0.966	0.004	0.838	0.549
Sense of Community	SoC1	Reflective	0.863	0.866	0.894	0.689	0.548
	SoC2					0.698	
	SoC3			_		0.682	
	SoC4			_		0.757	
	SoC5			_		0.752	
	SoC6					0.797	
<u> </u>	SoC7					0.795	0.555
Occupational Stress	OS1	Reflective	0.700	0.717	0.816	0.803	0.527
	OS2					0.636	
	OS3					0.724	
	OS4					0.732	
Affective Commitment	AC1	Reflective	0.884	0.893	0.909	0.763	0.560
	AC2					0.85	
	AC3					0.842	
	AC4					0.842	
	AC5					0.801	
	AC6					0.647	
	AC7					0.588	
	AC8					0.597	
Continuance Commitment	CC1	Reflective	0.898	0.901	0.917	0.753	0.582
	CC2					0.805	
	CC3					0.789	
	CC4					0.807	
	CC5					0.830	
	CC6					0.695	
	CC7					0.703	
	CC8					0.707	
Normative Commitment	NC1	Reflective	0.884	0.897	0.907	0.735	0.584
	NC2					0.735	
	NC3					0.777	
	NC4					0.729	
	NC5					0.737	
	1100						
	NC6					0.813	

Source: Calculations by the author

Discriminant validity assures that the model's constructs are distinct and unrelated to one another. Instead of using Fornell and Larker's Criterion, (1981), the discriminant validity was assessed through the more current and conservative HTMT method. Maximum ratios of 0.85 for the conservative approach (Voorhees, Brady, & Calantone, 2016; Henseler, Ringle, & Sarstedt, 2015; Kline, 2011) and 0.9 for the liberal approach (Gold, Malhotra, & Segars, 2001) are acceptable. According to the HTMT inference technique, all HTMT

values must be less than 1. The confidence intervals for HTMT inference were well inside the boundaries, demonstrating the constructs' uniqueness according to empirical standards. Table 2 displays the discriminant validity results.

Table 2: HTMT Correlation Ratio for DiscriminantValidity Evaluations

НТМТ	Ratio	of	Original	2.50	97.50
Correlations	5		Sample	%	%

	(0)		
	(0)		
Alignment of M 1			
Alignment of Values ->	0.704	0.616	0.770
Affective Commitment	0.704	0.616	0.779
Continuance Commitment ->	0.642	0.625	0.675
Affective Commitment	0.643	0.635	0.675
Continuance Commitment ->	0.720	0 (21	0.000
Alignment of Values	0.720	0.631	0.800
Meaningful Work ->	0.000	0.576	0.751
Affective Commitment	0.666	0.576	0.751
Meaningful Work ->	0.702	0.700	0.050
Alignment of Values	0.793	0.723	0.859
Meaningful Work ->	0.662	0.576	0 7 4 2
Continuance Commitment	0.663	0.576	0.743
Normative Commitment ->	0.00	0.502	0.725
Affective Commitment	0.668	0.593	0.735
Normative Commitment ->	0.750	0.604	0.020
Alignment of Values	0.759	0.684	0.828
Normative Commitment ->	0.704	0.720	0.045
Continuance Commitment	0.794	0.738	0.845
Normative Commitment ->	0.700	0.000	0.770
Meaningful Work	0.700	0.622	0.773
Occupational Stress ->	0.000	0.50	0.000
Affective Commitment	0.838	0.760	0.909
Occupational Stress ->			
Alignment of Values	0.750	0.730	0.775
Occupational Stress ->			
Continuance Commitment	0.850	0.771	0.925
Occupational Stress ->	0.045	0.005	0.047
Meaningful Work	0.847	0.827	0.867
Occupational Stress ->		0.000	0.050
Normative Commitment	0.797	0.833	0.959
Sense of Community ->		0.444	
Affective Commitment	0.737	0.666	0.797
Sense of Community ->	0.046	0.707	0.001
Alignment of Values	0.846	0.787	0.901
Sense of Community ->	0.7.0	0.500	0.010
Continuance Commitment	0.763	0.700	0.819
Sense of Community ->	0.750	0.001	0.021
Meaningful Work	0.759	0.694	0.821
Sense of Community ->	0.045		0.000
Normative Commitment	0.842	0.790	0.890
Sense of Community ->		0.01.1	0.007
Occupational Stress	0.824	0.816	0.837

Source: Calculations by the author

Second order CFA

After this, the second-order construct of workplace spirituality got tested for reliability and validity. To assess workplace spirituality, latent variable scores of meaningful work, sense of community, and alignment of values were employed. Table 3 and Table 4 explain the second-order internal reliability and convergent validity results.

Table 3: Outcomes of the Valuation of the Measurement
Model (Second order)

Construct	Cronbach'	rhoA	CR	AVE
	s Alpha			
Workplace	0.884	0.888	0.928	0.811
Spirituality				
Occupational	0.700	0.717	0.816	0.527
stress				
Affective	0.884	0.893	0.909	0.56
Commitment				
Continuance	0.898	0.902	0.917	0.582
Commitment				
Normative	0.884	0.897	0.907	0.584
Commitment				

Source: Calculations by the author

Table 4: Results of Factor Loading (Second order)

Latent Variables	Factor Loadings
Meaningful Work	0.879
Sense of Community	0.908
Alignment of Values	0.914

Source: Calculations by the author

Table 5 explains the results of discriminant validity at a 5% level of significance.

Table 5: HTMT Correlation Ratio for Discriminant Validity Evaluations (Second order)

	Original			
HTMT Ratio of	Sample			
Correlations	(O)	2.50%	97.50%	
Continuance Commitment -				
> Affective Commitment	0.814	0.788	0.826	
Normative Commitment ->				
Affective Commitment	0.668	0.596	0.737	
Normative Commitment ->				
Continuance Commitment	0.794	0.738	0.845	
Occupational Stress ->				
Affective Commitment	0.838	0.762	0.913	
Occupational Stress ->				
Continuance Commitment	0.850	0.768	0.927	
Occupational Stress ->				
Normative Commitment	0.847	0.834	0.869	
Workplace Spirituality ->				
Affective Commitment	0.779	0.708	0.842	
Workplace Spirituality ->				
Continuance Commitment	0.795	0.728	0.856	

Workplace Spirituality ->			
Normative Commitment	0.841	0.812	0.905
Workplace Spirituality ->			
Occupational Stress	0.824	0.812	0.848
Source: Calculations by the author			

Structural model

In structural model assessments (Hair, Hult, Ringle, & Sarstedt, 2017), the link between the constructs and their predictive significance was investigated. To begin, values for the Variance Inflation Factor (VIF) and tolerance were determined and discovered to be less than 3.33. (Diamantopoulos 2008) The outcomes of VIF are shown in Table 6.

Table-6: Variance Inflation Factor (VIF) of the inner model

mouer				
	Affective	Continuance	Normative	
	Commitment	Commitment	Commitment	
Occupational				
Stress	1.878	1.931	2.008	
Workplace				
Spirituality	2.763	2.281	2.563	

Source: *Calculations by the author*

The next step was to use the bootstrapping procedure with 5000 subsamples to determine the path coefficients'

significance and relevance, which might be anywhere between -1 and +1 (Hair et al., 2020). The structural model evaluations are depicted in Figure 2 and Table 7 as follows:

Figure-2: Structural path relations



Table 7: Structural Model Assessments							
Hypotheses	Path Relationship	Standardize	T Statistics	P Values	2.50%	97.50	Result
		d Beta (β)	(O/STDEV)			%	
H1a	Workplace Spirituality ->						
	Affective Commitment	0.538	5.600	0.000***	0.341	0.721	Supported
H1b	Workplace Spirituality ->						Not
	Continuance Commitment	0.564	5.959	0.000***	0.378	0.749	Supported
H1c	Workplace Spirituality ->						
	Normative Commitment	0.633	7.214	0.000***	0.456	0.802	Supported
H2	Workplace Spirituality ->						
	Occupational Stress	-0.918	12.433	0.000***	-0.932	-0.902	Supported
H3a	Occupational Stress -> Affective						
	Commitment	-0.197	2.039	0.042**	-0.386	-0.002	Supported
H3b	Occupational Stress ->						Not
	Continuance Commitment	-0.189	1.907	0.057	-0.378	0.008	Supported
H3c	Occupational Stress ->						Not
	Normative Commitment	-0.169	1.860	0.063	-0.346	0.012	Supported
H4a	Workplace Spirituality ->						
	Occupational Stress -> Affective	0.180	2.038	0.042**	0.002	0.354	Supported
	Commitment						
H4b	Workplace Spirituality ->						
	Occupational Stress ->						Not
	Continuance Commitment	0.173	1.907	0.057	-0.007	0.348	Supported
H4c	Workplace Spirituality ->			,	2.507		
•	Occupational Stress ->						Not
	Normative Commitment	0.155	1.860	0.063	-0.011	0.316	Supported
	Source: Calculations by the auth						

Source: Calculations by the author; Coefficients of the Path (*p<0.01, **p<0.05, ***p<0.001)

Table 7 demonstrates that workplace spirituality is the most prevalent element that negatively influences IT employees' occupational stress ($\beta = -0.918$, p<0.001), confirming H2. Workplace spirituality had the second-largest impact on normative commitment (β =0.634, p<0.001) and after that on affective commitment (β =0.538, p<0.001), which supported H1c and H1a respectively. Despite the fact that the association between workplace spirituality and continuance commitment has a value of 0.564, which is statistically significant, the hypothesis is rejected due to the positive relationship between the two. Occupational stress has a detrimental influence on all categories of commitment, although the outcomes are not statistically significant except for stress and affective commitment. As a result, only H3a is acceptable. Furthermore, Occupational stress appears to be a key mediator in the association between workplace spirituality and affective commitment (β =0.180, p<0.05). Spirituality at work reduces negative work stress effects, which is good for affective commitment. The next step is to calculate the values of R². In-sample predictive power is another name for the R² (Rigdon, 2012). Explanatory power is measured by R² that runs from 0 to 1, higher values suggest greater explanatory power. The R² values of 0.75, 0.50, and 0.25, respectively, are regarded as substantial, moderate, and weak, respectively (Henseler, Ringle and Sinkovics, 2009; Hair, Ringle, and Sarstedt, 2011). Affective, continuance, normative commitment and occupational stress were found to have coefficients of determination (R²) of 0.522, 0.550, 0.626, and 0.843 respectively, which is quite good. The effect size of f^2 and Q² were used to assess the predictive value and relevance of the suggested model. The proposed limits are 0.35 (for substantial impact). 0.15 (for moderate impact), and 0.02 (for modest impact) (Cohen, 1988). Workplace spirituality shows a moderate effect on normative commitment (f²=0.168) and a small to moderate effect on continuance commitment (f²= 0.111) and affective commitment $(f^2=0.095)$ while occupational stress has a small effect on affective ($f^2 = 0.013$), continuance ($f^2 = 0.012$) and normative commitment ($f^2 =$ 0.012). Following this, Stone- Geissers Q² was used to study the structural model's predictive validity, which yielded 0.257, 0.289, and 0.345 for affective, continuance, and normative commitment, respectively, showing the model's predictive significance is moderate. (Geisser, 1975; Stone, 1974). If the value is higher than 0.02, the model has sufficient predictive power and can be used to extrapolate future results (Richter, Cepeda & Roldan, 2016). Q² for occupational stress was found to be 0.441, indicating strong predictive relevance, implying that workplace spirituality, as well as occupational stress, are crucial in the study's conceptual model and the findings can be applied to new scenarios in the future. Finally, the goodness of fit criterion was assessed using SRMR indices. The SRMR value of the

study's model is 0.078, which is less than the cut-off value of 0.08, indicating that it has a high power of explanation (Henseler, Hubona & Ray, 2016; Hu and Bentler, 1999).

IV. Conclusion

With the changing times, the IT sector is facing various kinds of uncertainties. Indian IT exports have remained heavily reliant on the US and European markets. The US government's anti-outsourcing policies, the developed world's slowing economy, and the fast-growing competition from the Philippines, China, and Costa Rica are all putting pressure on India's status as the global outsourcing hub. In addition, the country's IT sector is short on professionals and venture money. Another issue that the sector is dealing with, is the heightened threat of cyber security posed by cloud computing, e-commerce, and digital payments. The number of patents linked to information technology filed by Indian enterprises has remained relatively low. The strong rivalry that exists in today's global and local marketplaces compounded by demanding work schedules and changing work environments, according to Altaf and Awan (2011), has motivated organizations to establish extremely lofty goals for their staff. As a result, employees are frustrated, vulnerable, worried to a great degree, and prone to increased levels of stress (Gupta, Kumar, and Singh, 2014). In such a painful atmosphere, cultivating a spiritual presence of God's values at work, such as caring, kindness, peace, and love, is crucial for rewarding the inner life of employees.

Limitations and future research directions

While the findings are encouraging, it's worth noting that this research relied solely on a self-report survey instrument. Though precautions have been taken while collecting data, due to the fact that this is a self-report study, it's probable that some of the scales have common method variance, inflating statistical relationships. Spirituality is subjective and very personal, thus It would be perfect if other research methodologies, such as employee interviews and supervisor assessments, were used to cross-validate these measurements. Future research may employ a two-source technique. A longitudinal study design could also be useful. We used convenience sampling in this case; however, any probability sampling approach might be used. It will improve the results' legitimacy. Finally, in this study, a cross-sectional strategy was used. Based on cross-sectional studies, no researcher can definitively demonstrate causal correlations. As a result, any long-term study could confirm the findings of the current studies. Future research could concentrate on how to conceptualize and quantify workplace spirituality using both qualitative and quantitative methods.

References

- Ali, F., Rasoolimanesh, S. M., Sarstedt, M., Ringle, C. M., & Ryu, K. (2018). An assessment of the use of partial least squares structural equation modeling (PLS-SEM) in hospitality research. *International Journal* of Contemporary Hospitality Management.
- Allen, N. J., & Meyer, J. P. (1990). The measurement and antecedents of affective, continuance and normative commitment to the organization. *Journal of occupational psychology*, 63(1), 1-18.
- Allen, N. J., & Meyer, J. P. (2000). Construct validation in organizational behavior research: The case of organizational commitment. *Problems and solutions in human assessment: Honoring Douglas N. Jackson at seventy*, 285-314.
- Altaf, A., & Awan, M. A. (2011). Moderating affect of workplace spirituality on the relationship of job overload and job satisfaction. *Journal of business ethics*, 104, 93-99.
- Ashmos, D. P., & Duchon, D. (2000). Spirituality at work: A conceptualization and measure. *Journal of management inquiry*, 9(2), 134-145.
- Blau, P. M. (1964). Justice in social exchange. *Sociological inquiry*, 34(2), 193-206.
- Cicei, C. C. (2012). Occupational stress and organizational commitment in Romanian public organizations. *Procedia-Social and Behavioral Sciences*, 33, 1077-1081.
- Cohen, J. 1988. Statistical Power Analysis for the Behavioral Sciences, 2nd edn. Hillsdale, NJ: Lawrence Earlbaum Associates
- Daniel, J. L. (2015). Workplace spirituality and stress: Evidence from Mexico and US. *Management Research Review*, 38(1), 29-43.
- Diamantopoulos, A. (2008). Formative indicators: Introduction to the special issue.
- Diamantopoulos, A., Sarstedt, M., Fuchs, C., Wilczynski, P., & Kaiser, S. (2012). Guidelines for choosing between multi-item and single-item scales for construct measurement: a predictive validity perspective. *Journal of the Academy of Marketing Science*, 40, 434-449.
- Drolet, A. L., & Morrison, D. G. (2001). Do we really need multiple-item measures in service research?. *Journal of service research*, *3*(3), 196-204.
- Fornell, C., & Larcker, D. F. (1981). Evaluating structural equation models with unobservable variables and measurement error. *Journal of marketing research*, 18(1), 39-50.
- Geisser, S. (1975). The predictive sample reuse method with applications. *Journal of the American statistical Association*, 70(350), 320-328.

- Giacalone, R. A., & Jurkiewicz, C. L. (2003). Right from wrong: The influence of spirituality on perceptions of unethical business activities. *Journal of business Ethics*, 85-97.
- Gold, A. H., Malhotra, A., & Segars, A. H. (2001). Knowledge management: An organizational capabilities perspective. *Journal of management information systems*, 18(1), 185-214.
- Gupta, M., Kumar, V., & Singh, M. (2014). Creating satisfied employees through workplace spirituality: A study of the private insurance sector in Punjab (India). *Journal of business ethics*, *122*, 79-88.
- Hair Jr, J. F., Anderson, R. E., Tatham, R. L., & Black, W. C. (1998). Multivariate data analysis. 5th Intl. ed Prentice Hall Upper Saddle River.
- Hair, J. F., Ringle, C. M., & Sarstedt, M. (2011). PLS-SEM: Indeed a silver bullet. *Journal of Marketing theory* and Practice, 19(2), 139-152.
- Hair, J. F., Hult, G. T. M., Ringle, C., &Sarstedt, M. (2017). A primer on partial least squares structural equation modeling (PLS-SEM) (2nd ed). Thousand Oaks, CA: SAGE Publications.
- Hair, J. F., Sarstedt, M., & Ringle, C. M. (2019). Rethinking some of the rethinking of partial least squares. *European Journal of Marketing*, 53(4), 566-584.
- Hair Jr, J. F., Howard, M. C., & Nitzl, C. (2020). Assessing measurement model quality in PLS-SEM using confirmatory composite analysis. *Journal of Business Research*, 109, 101-110.
- Henseler, J., Ringle, C. M., & Sinkovics, R. R. (2009). The use of partial least squares path modeling in international marketing. In *New challenges to international marketing*. Emerald Group Publishing Limited.
- Henseler, J., Ringle, C. M., & Sarstedt, M. (2015). A new criterion for assessing discriminant validity in variance-based structural equation modeling. *Journal of the academy of marketing science*, 43, 115-135.
- Henseler, J., Hubona, G., & Ray, P. A. (2016). Using PLS path modeling in new technology research: updated guidelines. *Industrial management & data systems*.
- Hu, L. T., & Bentler, P. M. (1999). Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural equation modeling: a multidisciplinary journal*, 6(1), 1-55.
- Jalan, S., & Garg, N. (2022). Exploring linkages between workplace spirituality and occupational stress among Indian police personnel. South Asian Journal of Human Resources Management, 9(1), 53-78.
- Jurkiewicz, C. L., & Giacalone, R. A. (2004). A values framework for measuring the impact of workplace

spirituality on organizational performance. *Journal of business ethics*, 49, 129-142.

- Kline, R. B. (2011). Convergence of structural equation modeling and multilevel modeling.
- Lambert, E. G., Hogan, N. L., & Keena, L. D. (2015). The impact of job attitudes on private correctional staff's continuance and affective organizational commitment. *Journal of Applied Security Research*, 10(1), 1-22.
- Meyer, J. P., & Allen, N. J. (1991). A three-component conceptualization of organizational commitment. *Human resource management review*, 1(1), 61-89.
- Milliman, J., Czaplewski, A. J., & Ferguson, J. (2003).
 Workplace spirituality and employee work attitudes: An exploratory empirical assessment. *Journal of organizational change management*, 16(4), 426-447.
- Motowidlo, S. J., Packard, J. S., & Manning, M. R. (1986). Occupational stress: its causes and consequences for job performance. *Journal of applied psychology*, *71*(4), 618.
- O'Leary-Kelly, S. W., & Vokurka, R. J. (1998). The empirical assessment of construct validity. *Journal of operations management*, *16*(4), 387-405.
- Rego, A., & Pina e Cunha, M. (2008). Workplace spirituality and organizational commitment: an empirical study. *Journal of organizational change* management, 21(1), 53-75.
- Richter, N. F., Cepeda-Carrion, G., Roldán Salgueiro, J. L., & Ringle, C. M. (2016). European management

research using partial least squares structural equation modeling (PLS-SEM). *European Management Journal, 34 (6), 589-597.*

- Rigdon, E. E. (2012). Rethinking partial least squares path modeling: In praise of simple methods. *Long range planning*, *45*(5-6), 341-358.
- Ryan, R. M., & Deci, E. L. (2000). Self-determination theory and the facilitation of intrinsic motivation, social development, and well-being. *American psychologist*, 55(1), 68.
- Saadeh, I. M., & Suifan, T. S. (2020). Job stress and organizational commitment in hospitals: The mediating role of perceived organizational support. *International Journal of Organizational Analysis*, 28(1), 226-242.
- Saxena, A., Garg, N., Punia, B. K., & Prasad, A. (2020). Exploring role of Indian workplace spirituality in stress management: a study of oil and gas industry. *Journal of Organizational Change Management*, 33(5), 779-803.
- Stone, M. (1974). Cross-validatory choice and assessment of statistical predictions. *Journal of the royal statistical* society: Series B (Methodological), 36(2), 111-133.
- Voorhees, C. M., Brady, M. K., Calantone, R., & Ramirez, E. (2016). Discriminant validity testing in marketing: an analysis, causes for concern, and proposed remedies. *Journal of the academy of marketing science*, 44, 119-134.

FINTECH AND INDUSTRY 4.0: A STUDY ON DRIVERS WITH REFERENCE TO DIGITAL BANKING SECTOR

Deepa Joshi*Sanjay Sharma**

Purpose: The purpose of this paper is to investigate drivers of FinTech in India with reference to banking sector, customers' awareness and acceptance of Digital Banking.

Design/methodology/approach: The present study is of descriptive and exploratory nature based on primary as well as secondary data. A well-structured questionnaire was circulated to collect the data using simple random sampling. The data was analyzed by using Factor analysis and Chi square test as statistical tools.

Findings: FinTech is effective in digital banking sector. Majority of respondents know about the BancTec and FinTech and are using it very often proved that there is an awareness related to Digital Banking amongst Indian banking customers The factors extracted were named as – Unique digi-fin experience, Easy and user friendly, 24*7 accessibility, Safe and secure transactions and Enhanced customer experience.

Research limitations/implications: Impact of demographics on the usage of FinTech was not studied. Geographical area of the study was restricted to Indore city only. Future research could use demographic variables and also extend the area of the study to State level to verify the impact of FinTech on banking sector.

Practical implications: The findings have implications for FinTech professionals and banking sectors to take advantages of multiple benefits of digital banking in content acquisition, confidence in knowledge and ability, and action taking when they communicate with their customers.

Originality/value: This study is the first of this kind to focus on extracting the influential drivers of FinTech and found the effectiveness of FinTech in banking sector.

Keywords: FinTech, Digital Banking, Industry 4.0, financial services, banking customers, awareness level **JEL Classification:** H80

Fintech refers to the use of technology in an individual's, businesses, and other people's financial transactions. Individuals and financial institutions' spending, saving, investing, and borrowing habits are affected. India's FinTech business is growing at a spectacular rate as a result of increased digitalization. Digital payments users are at the top of the FinTech sector in Asia in 2020, according to the Statista Research Department Report, May 2021. Around 2.1 billion Indians made digital payments this year, resulting in the greatest transaction value of nearly \$3.13 trillion USD. By 2025, it is expected that the number of users would have risen to 3.1 billion.

Evolution of Fintech in India

Between 1991 and 2000, the Indian banking industry began to fully embrace technology. During this time, the Indian government began implementing technology in banks using a variety of payment systems such as MICR, EFT, and others. FinTech dominated the US and UK banking industries from 2000 to 2005, and was also attempting to make a mark on the Indian banking business by providing a variety of customeroriented services. FinoPay Tech and EKO India were two of the most important companies at the time. The rural areas were where they were most concentrated. Between 2005 and 2010, prominent financial start-ups such as Oxigen, Paytm, Freecharge, and a slew of other mobile wallets popped up. More than 100 fintech start-ups in lending, more than 40 in personal finance management, and more than 90 in investment management were founded between 2010 and 2014. FinTech in India grew and prospered after demonetization in 2016. It introduced the India stack, which includes Aadhar, eKYC, UPI, and other technologies. NeoBanks began to take shape in 2019. NeoBanks pushed trade financing and business-to-business (B2B) in a new direction.

Financial Technology

Financial technology is a combination of technology and innovation aimed at competing with traditional financial methods in the supply of financial services. It is a new industry that employs technology to improve financial activities.

FinTech companies in India are setting new standards for financial services (Mathur, H.2020) various government programmes such as the Jan Dhan Yojana, Aadhaar, and the advent of UPI, which give a sound basis to enhance financial inclusion in India ((<u>https://www.financialexpress.com</u>), are encouraging the fintech business in India.

* Associate Professor Shri Vaishnav Institute of Management, Indore (M.P.)

** Professor & Principal Indore Management Institute, Indore (M.P.) Males and females adopted FinTech apps at a rate of 88 percent and 84 percent, respectively, while those aged 25 to 44 are the largest users of FinTech's at around 94 percent, while FinTech adoption is around 73 percent globally in the same age group. (https://bfsi.economictimes.indiatimes.com)

India is one of the world's fastest growing FinTech marketplaces. With China, India has the highest FinTech adoption rate in the world. The \$65 billion in digital payments in 2019 is predicted to grow at a CAGR of 20% until 2023. (www.investindia.gov.in) Multiple characteristics distinguish markets around the world, and the FinTech industry is no exception. India stands out as a distinct market for FinTech, particularly the payments community, with its own set of difficulties and potential. (S. Adithya, 2019)

Fintech is a colloquial phrase for financial technology, which refers to any technology utilized in the financial services business. Online payment methods such as PayPal, Apple Pay, TransferWise, and Pioneer are among the first to come to mind. FinTech is booming because it has dramatically increased access to financing for small business owners, particularly women, minorities, and immigrants, who were previously underserved before technological advancements levelled the playing field.

Fintech is a phrase that refers to new technology that automates and enhances financial service delivery. Banks, on the other hand, are financial institutions that are licensed to accept deposits and make loans from their customers.

FinTech (financial technology) has exploded all over the world. (A. S. Kavuri and A. Milne, 2018) FinTech, or Financial Technology, is a term that describes a company that combines emerging technological trends to offer better financial solutions to its customers in the form of digital payments and transactions. (Aakriti Suri Saumya Uniyaland) (,2020) In recent years, the Indian fintech industry has seen a slew of new entrants. With each new company, the industry has expanded significantly and is gaining international recognition. According to the most recent CB Insights reports, these are the 20 Indian fintech startups that have been named among the top 250 global promising startup companies (Wilson, R., 2020), and UPI, BHIM, Payment wallet, digital bank, and many other services are available, and we use them on a daily basis. Which best portrays the importance of money and technology in our lives.

The focus is on four fintech topics in this primer: digital lending, payments, blockchain, and digital wealth management, which are of particular interest because to their rapid growth, technical disruption, and regulatory and other issues.

Trends in Fintech

"In 2020, new financial services technology trends will emerge. The ability of financial organizations to shape the sharing economy and consumer intelligence, as well as deal with breakthroughs in technology like blockchain, robotics, Artificial Intelligence (AI), and more, is predicted to drive their growth." (<u>https://www.kelltontech.com</u>)

Artificial Intelligence in Fintech

The banking business has been transformed by artificial intelligence (Hill, K., 2020) Few business titans had the bandwidth to deal with the intrinsically quantitative character of our now tech-savvy society before AI and the rise of FinTech. These AI use examples show how AI has revolutionized the FinTech industry. (Buttice, C., 2020) FinTech firms and financial institutions utilize AI and machine learning to boost efficiency and give customers with more relevant, accessible, and intuitive financial services and products. (2020, Darcy Tyrrell)

Blockchain in Fintech

Fintech is revolutionizing the financial industry, and blockchain development companies in this field will have a significant competitive advantage in the future (Shah, P,2019) A paper recently released by the Ministry of Finance's steering group on Fintech-related topics highlighted the relevance of Blockchain, with a focus on four blockchain applications in Fintech (Jacob, C,2020) Blockchain technology is revolutionizing the fintech business in a variety of ways, including removing third parties, lowering operational time and cost, and improving identity verification (Srivastav, S,2020).

Cloud Computing in Fintech

Using the Cloud Fintech organizations may rapidly scale up and down with full control while remaining compliant with regulatory standards thanks to cloud infrastructure. (Naser, A. T., 2020) FinTech has experienced significant growth and continues to do so thanks to cloud technologies. FinTech cuts CAPEX and OPEX costs while expanding the service portfolio and improving the user experience. (Ray, T., 2020).

NLP-based chatbots

According to PwC India, Fintech will revolutionize mobile banking by utilizing NLP-based chatbots and inventing Conversational User Interface (CUI). These chatbots will be able to reply to consumer issues in real time and offer viable solutions, hence improving the customer experience. (https://economictimes.indiatimes.com/bfsi)

Industry 4.0

"A name for the current trend of automation and data exchange in manufacturing technologies, encompassing cyber-physical systems, the Internet of things, cloud computing and cognitive computing, and establishing the smart factory," according to the definition of Industry 4.0. Due to increased interconnection and smart automation, the Fourth Industrial Revolution, also known as Industry 4.0, envisions significant change in technology, industries, and societal patterns and processes in the twenty-first century.

Components:

The application of the Fourth Industrial Revolution operates through:

- Mobile devices
- Internet of things (IoT) platforms
- Location detection technologies (electronic identification)
- Advanced human-machine interfaces
- Authentication and fraud detection
- Smart sensors
- Big analytics and advanced processes
- Multilevel customer interaction and customer profiling
- Augmented reality/ wearables
- On-demand availability of computer system resources
- Data visualization and triggered "live" training

Relation between fintech and industry 4.0:

Industry 4.0 has had a substantial impact on financial services, just as it has on manufacturing and other global industries. Online payments, digital loans, plastic money, crypto currency, online forex trading, and many other financial activities in the financial sector are all affected by industry 4.0, often known as I-4. All of this is thanks to the information technology revolution, which is fueled by the internet, software consultancy, and software development. The global software team of qualified individuals deserves credit for this digital revolution. The fourth industrial revolution has spread its wings across all industries, having a positive impact. Financial services are another business that has made great use of industry 4.0's capabilities. Banking, insurance, mortgages, forex, stocks, and many other financial industries are growing across the globe as a result of the positive impact of digital innovation and financial process automation.

I. Review of Literature

The process of digitization of our domestic and professional lives cannot be overlooked. This digitization has introduced new ways to financial services namely online banking, ATMs, and credit cards. Microfinance institutions with the side hand of FinTech companies in India are working on technology advancement which in lead to benefitting middle and underprivileged class by providing easy access to capital (Kapadia, 2020). Fintech is able to be disturbing financial systems (Allen et. al, 2020). New and updated technology is used in introducing a way of new banking and lending products that are increasingly becoming faster, cheaper and more accessible (Barry, E. ,2020). Introduction of Modern FinTech i.e. financial technology is related to any innovation that involves financial transactions, whether it is for personal use or for business. There was a time when FinTech was used only in the back office operations of banks or stock trading companies. The boom in the internet usage and mobile computing had pushed FinTech to become a continuing global revolution. Today, FinTech has already established itself as an important element in today's digital world. With expanding technological tools for personal and commercial finance, it is expected to further grow in future (Andre, L., 2019)

The role of artificial intelligence, augmented reality and block-chain in digital banking sector. AR is pervasive now; it has a remarkable impact in every sector. AR technologies aid in increasing efficiency, reducing process costs, and various commercial benefits. At present, augmented reality technology is having a vibration impact in numerous industry sectors. From being adopted in healthcare, oil and gas construction, and retail sector as well as manufacturing, AR technologies have currently been adopted with the motive to increase process efficiency, reduce costs and bring about a broad range of commercial benefits (Dubey, 2019).

India's FinTech landscape, approaching the analysis from a FinTech, regulatory, Investment, and talent standpoint. The study explained the key pillars of a FinTech ecosystem in a country and provided a good overall view of the state of FinTech at a glance. According to him, India's FinTech sector is growing rapidly, supported by a large consumer base, unmet financial needs, SME credit gap, and a conducive environment favored by regulatory initiatives and policies. Much of FinTech adoption in the country is driven by digital payments, which has got motivation by recent innovations like United Payments Interface (UPI) platform. India has experienced a huge shift from offline cash transactions towards digitization, majorly due to Government initiatives and increasing mobile and internet penetration. Banks and the financial services industry are also working in collaboration with FinTechs which has resulted in a strong B2B FinTech presence in the country (Mittal, 2019). Singapore and India are at the first position to in the implementation of FinTech. On one side, India offers the highest global returns on investment in FinTech. Singapore has developed a cuttingedge sandbox for trying new FinTech products and has a world-class digital and physical infrastructure. There is a large scope for mutually beneficial collaboration. The researchers suggested that collaboration between governments requires information sharing on policies, consultations on data privacy, cyber security and more and interoperability allows for seamless fund transfers across geographical zones via a single account (Randhawa, Hao & Murali, 2018).

FinTech is the latest innovation in the area of banking and financial services. FinTech has to be carefully handled to challenge the domination of traditional financial institutions. In India, the traditional financial institutions are undergoing a period of turbulence, which has in the last few years witnessed the growth of bad loans, dissatisfaction among the customers regarding several financial products and services, and increasing loss of confidence among the public with an
imminent fear of a financial crisis. The opportunity seems to bloom for the emerging economies but FinTech is still a nascent player in the Indian financial sector. Therefore, collaboration with the traditional financial institutions would help in providing a new direction to India's financial sector (Gurung, 2018). Entering Industry 4.0 includes two unique aspects: first, meeting new requirements through new goods and processes, and second, increasing productivity through process improvements (Zambon et al. 2019). To bring a new generation of clients into the marketplace and do business in this transformation, traditional banking services and products must be completely rethought. Some of the benefits derived from technology in banks include increased productivity, creative products, quick transactions, seamless fund transfers, real-time information systems, and effective risk management (Saravanan and Muthu Lakshmi 2016).

Surprisingly, Industry 4.0 need its own finance system. One of the most essential tactics of Banking 4.0 is the use of Industry 4.0 technology for digitizing assets, creating a digital identity, delivering special offers to consumers, offering personalization, and so on. For example, behind Japan and the United States, South Korea now has the third-largest crypto currency market, and Shinhan Bank, the country's secondlargest bank, has recently joined KT Corp, the country's second-largest service provider, and the country's telecommunications have cooperated. The creation of a block chain-based platform was the focus of this collaboration. For a long time, the banking industry was in charge of deciding how to connect with customers and deliver customer care. The consumer determines the best interaction mix in Generation Banking.

Objectives of the study:

- To identify the drivers of FinTech in India with reference to banking sector.
- To understand the customers' awareness and acceptance of Digital Banking.

Hypotheses:

 H_{01} : FinTech is not effective in digital banking sector.

 H_{02} : There is no awareness related to Digital Banking amongst Indian banking customers.

II. Research Design and Methods

The present study is a blend of descriptive and exploratory concept. The study is based on primary as well as secondary data. The primary data is collected from 250 respondents from Indore region. A well-structured questionnaire was circulated to collect the data using simple random sampling. The data was analyzed using Percentage analysis, Factor analysis and Chi square test as statistical tools. The software used for data analysis are SPSS 25 and Excel.

III. Result and Discussion

Table -1 Reliability Test					
Cronbach's Alpha No. of Items					
0.872	22				

The test of reliability is applied to test whether the data collected for the study is reliable or not. The value of the Chronbach alpha shows the accuracy of the data. In this case the value of C alpha was found to be 0.872, which depicts that the data collected is reliable and can be analyzed further.

Table - 2Sphere city Test				
KMO	and Bartlett's Test			
Kaiser-Meyer-Olkin Measure of Sampling Adequacy 0.897				
Bartlett's Test of	Approx. Chi-Square	1250.560		
Sphericity	df	231		
sphericity	Sig.	0.000		

The test was applied to check the normality and validity of the data. The value was found to be 0.897, which indicated that the data is valid.

Table -3
Demographic Distribution of Respondents

Gender	r	
	Ν	%
Male	112	44.8
Female	138	55.2
Total	250	
Age		
Below 20 years	9	3.6
20 to 25 years	95	38
25 to 30 years	107	42.8
30- to 35 years	19	7.6
35 to 40 years	10	4
Above 40 years	10	4
Total	250	
Annual Income		
Less than 3L	27	10.8
3L to 6L	182	72.8
6L to 9L	34	13.6
More than 9L	7	2.8
Total	250	
Area		
Rural	5	2
Semi urban	33	13.2
Urban	212	84.8
Total	250	

The above table shows the demographic profile of the respondents. The gender wise distribution shows that 55.2% of the total respondents were female whereas 44.8% were males. Age wise the major category of respondents belonged to age group of 25 to 30 years with 42.8%, whereas other respondents belong to below 20 years, 20 to 25 years, 30 to 35 years, 35 to 40 years and above 40 years with percentage 3.6%, 38%, 7.6%, and 4% respectively.

For analysis annual income was considered in 4 levels i.e. Less than 3 Lakhs, 3L to 6L, 6L to 9L and More than 9L. The major respondents had income level 3L to 6L as their income level with 72.8% as the contribution in income category. The remaining levels had 10.8%, 13.6% and 2.8% respectively. The geographical distribution of respondents was divided into 3 categories i.e. Rural, Semi-Urban and Urban. It was found that respondents from urban areas participated the most with 84.8% whereas Semi-Urban and Rural areas had 13.2% and 2% respectively.

Exploratory Factor Analysis:

EFA was performed to extract the factors which drive respondents to use the FinTech services. Total 5 factors were extracted from 22 items.

Table – 4				
Factors				
3.58				
2.92				
2.63				
2.42				
1.40				

The above table shows the Factor name and the loading of each factor. The factors extracted were named as – Unique digi-fin experience, Easy and user friendly, 24*7 accessibility, Safe and secure transactions and Enhanced customer experience.

The most loaded factor amongst all five was found to be Unique digi-fin experience with factor load 3.58, the second most loaded factor was ease and user friendly with factor load 2.63. Factor named 24*7 accessibility was found to have 2.63 as the factor load where as Safe and secure transactions and Enhanced customer experience had 2.42 and 1.40 loading respectively.





Hypotheses Testing

H0: There is no awareness related to Bank-Tech amongst Indian banking customers.

Table -5 Number of respondents having awareness towards online payment Apps

Are you aware about online payments app like (example Paytm, Google Pay for investment or other financial Services)				
Yes	211	84.4		
No	39	15.6		
Total	250			

Figure -2

Graphical presentation of Number of respondents having awareness towards online payment Apps



diffe	different online apps					
When it comes to payment through apps you prefer						
Wallet to Wallet	8	3.2				
UPI to Bank Transfer	143	57.2				
Bank to Bank Transfer	67	26.8				
Wallet to UPI	32	12.8				
Total	250					







Table -7 Number of respondents having different frequencies of Etransactions

LI di	ti alisactions					
How often do you opt for online transactions or E- transactions?						
Once a week	102	40.8				
Daily 53 21.2						
Very Often	34	13.6				
Once a month	26	10.4				

Very Rare	35	14
Total	250	

Figure - 4

Graphical presentation of number of respondents having different frequencies of E-transactions



From the above charts it is evident that the respondents have awareness about the bank tech. Majority of respondents know about the BancTec and fintech and are using it very often. So the null hypothesis on the basis of percentage analysis is rejected.

Chi-Square Test

The Chi-Square test is a statistical tool used by researchers to assess the differences between categorical variables in the same population. And also helps to find out the effectiveness. In this case the Chi Square test is applied to check the effectiveness of the fintech

H01:	Fintech	is not	effective	in	digital	banking sector.
**U1•	1 muuum	10 1100			uigitui	building beccore

			User profile		Tatal
		Non-U	Users	Total	
	No	Count	4	35	39
	INU	Expected Count	0.8	38.2	39.0
Fintech use	Fintech use Yes	Count	1	210	211
		Expected Count	4.2	206.8	211.0
Total		Count	5	245	250
		Expected Count	5.0	245.0	250.0

Table -8, Fintech use * User profile Cross tabulation

	Value	df	Asymptotic Significance (2-sided)	Exact Sig. (2-sided)	Exact Sig. (1-sided)
Pearson Chi-Square	16.071ª	1	0.000		
Continuity Correction ^b	11.468	1	0.001		
Likelihood Ratio	10.527	1	0.001		
Fisher's Exact Test				0.002	0.002
N of Valid Cases	250				
a. 2 cells (50.0%) have expect	ted count less th	an 5. The min	imum expected count is .78.		
b. Computed only for a 2x2 t	able				

Table -9, Chi-Square Tests

If this number (Pearson chi square value) is equal to or less than the prescribed alpha level, the result is significant (normally .05). Because the p-value is less than the typical alpha value in this circumstance, we'd reject the null hypothesis that the two variables are unrelated. To put it another way, the result is significant: the data indicates that the variables are linked. Hence we reject the null hypothesis **"FinTech** *is not effective in digital banking sector".* The rejection indicates that FinTech is effective in digital banking sector.

IV. Conclusion

In many ways, the development and application of financial technologies has altered the Banking and Financial Services businesses. Individuals may establish a culture of banking and investing, as well as a cashless society, using the range and depth of FinTech applications available today. Payments are made digitally. FinTech has the ability to transform the finance industry by utilizing new technologies. Technology, as well as cultivating better interactions between merchants and consumers tackling the problem of financial inclusion Block chain, Big Data, Artificial Intelligence (AI), Machine Learning, and a slew of other cutting-edge technologies will be beneficial to growing economies such as India. The study proved that there are five drivers of FinTech namely Unique digi-fin experience, Easy and user friendly, 24*7 accessibility, Safe and secure transactions and Enhanced customer experience. Out of which Unique digi-fin experience was found to be the most influential driver. It was also found that majority of the respondents were aware about the FinTech.

Limitations of the study:

- Impact of demographics on the usage of FinTech was not studied.
- Geographical area of the study was restricted to Indore city only.

Scope of the study:

- The present study can be explored to know the Impact of demographics on the usage of FinTech.
- Respondents from metro cities can be studied to have a better understanding and wider scope of FinTech.

References

- Allen, F., Gu, X., & Jagtiani, J. (2020). A Survey of Fintech Research and Policy Discussion. Review of Corporate Finance 1 (2021), 259-339, FRB of Philadelphia Working Paper No. 20-21, Available at <u>SSRN: https://ssrn.com/abstract=3622468 or http://dx</u> .doi.org/10.21799/frbp.wp.2020.21
- Andre, L. (2019). What is Fintech? Examples of Types, Products & Regulations. Retrieved from <u>https://financesonline.com/what-is-Fintech/</u>
- Agarwal, S., & Hauswald, R. (2010): Distance and private information in lending. The Review of Financial Studies, 23(7), 2757-2788.
- Adithya, S. (2019). India Is Quite a Different FinTech Market: Great for Consumers and Very Trying for Start-ups. Retrieved from <u>https://gomedici.com/indiais-a-different-fintech-market-great-for-consumersand-very-trying-for-startups/</u>
- Bhandari, M. (2016): India and the Pyramid of Opportunity. In: The FinTech Book: The Financial Technology Handbook for Investors, Entrepreneurs and Visionaries, 81-83.
- Buttice, C. (2020). Top 12 AI Use Cases: Artificial Intelligence in FinTech. Retrieved from <u>https://www.techopedia.com/top-12-ai-use-cases-</u> artificial-intelligence-in-fintech/2/34048
- CB Insights (2017): The CB Insights FinTech 250. https://www.cbinsights.com/researchFinTech250.
- Chan, Y. S.& Thakor, A. V. (1987): Collateral and competitive equilibrium with moral hazard and private

information. The Journal of Finance, 42(2), p. 345-363.

- Chesbrough, H.W. (2007): Business model innovation: it's not just about technology anymore. Strategy & Leadership, 35(6), 12-17.
- Hill, K. (2020). How AI is Revolutionizing the Process of Fintech Firms? Retrieved from <u>https://medium.com/towards-artificial-</u> <u>intelligence/how-ai-is-revolutionizing-the-process-of-</u> <u>fintech-firms-87e057a48d83#:~:text=AI helps</u> <u>Fintech companies in resolving human problems, of</u> <u>Human Intelligence at a beyond human scale.</u>

Jacob, C. (2020). Blockchain in Fintech - Technology - India.

- Retrieved from <u>https://www.mondaq.com/india/fin-</u> tech/897490/blockchain-in-fintech
- Kavuri, A. S., & Milne, A. (2018). Fintech and the Future of Financial Services: What are the Research Gaps? Retrieved from <u>https://papers.ssrn.com/sol3/papers.cfm?abstract_id=</u> <u>3215849</u>
- Kim, Y., Park, Y. J.& Choi, J. (2016). The Adoption of Mobile Payment Services for "FinTech". International Journal of Applied Engineering Research, 11(2), 1058-1061. PwC (2016): Global FinTech Report, Blurred lines: How FinTech is shaping the financial world? Lon-don: PwC.
- Naser, A. T. (2020). The Impact of Cloud Computing in Fintech. Retrieved from <u>https://vexxhost.com/blog/cloud-computing-in-</u> <u>fintech/#:~:text= Critical Trends and Benefits of</u> <u>Cloud Computing, account from a simple... 3</u> <u>Security More</u>
- Ray, T. (2020). Scopes and Impact of Cloud Computing on Banking & FinTech. Retrieved from <u>https://www.stoodnt.com/blog/cloud-computing-banking-fintech/</u>
- Shah, P. (2019). How fintech is revolutionizing with blockchain technology. Retrieved from https://yourstory.com/2019/10/fintech-blockchain-technology.
- Shrivastava, D. (2020). Fintech Industry in India: History, Growth, And Future of Fintech in India. Retrieved from <u>https://startuptalky.com/fintech-industries-in-india/</u>
- Uniyal S. & Suri A. (2020). Most Successful Fintech Startups & Companies in India as of 2020. Retrieved from <u>https://timesnext.com/10-most-successful-fintech-startups-companies-in-india/</u>
- Wilson, R. (2020, September 05). 20 Most Promising Indian Fintech Start-ups as Per CB Insights 2020 List. Retrieved from <u>https://www.marketingmind.in/20-</u>

most-promising-indian-fintech-startups-as-per-cbinsights-2020-list/

- Vivek Dubey (2019). FinTech Innovations in Digital Banking, International Journal of Engineering Research & Technology (IJERT), 8(10), 597-601.
- Gurung, S. (2018). FinTech: A Messiah for the Ailing Banking Industry in India. Journal of Emerging Technologies and Innovative Research, 5(10), 159-164. Retrieved from https://www.jetir.org/papers/JETIRG006020.pdf
- Kapadia, S. (2020). How Digitization Is Impacting Banking Transactions and Financial Markets in India? A Discussion Paper Retrieved from <u>https://www.researchgate.net/publication/339016332</u> <u>How_Digitization_Is_Impacting_Banking_Transact</u> <u>ions_and_Financial_Markets_in_India_A_Discussion</u> <u>Paper/citation/download</u>
- Mittal, V. (2019). FinTechs vs Banks Winning the heart of the customer. Retrieved from <u>https://www.researchgate.net/publication/347441491</u> <u>FinTechs_vs_Banks_Winning_the_heart_of_the_cu</u> <u>stomer/citation/download</u>
- Chan Jia Hao, Vani Swarupa Murali, Dipinder Singh Randhawa (2018). India-Singapore FinTech Cooperation: Opportunities and Challenges. ScholarBank@NUS Repository. Retrieved from https://scholarbank.nus.edu.sg/handle/10635/188260

Webliography:

https://enterslice.com/learning/types-and-state-of-fintechcompanies-in-india/ https://www.financialexpress.com/ https://www.statista.com https://www.statista.com/ https://mobilityforesights.com/product/india-fintech-market/ https://economictimes.indiatimes.com/ https://www.fintechindiaexpo.com/about-us.aspx https://www.investindia.gov.in/sector/bfsi-fintech-financialservices https://www.investindia.gov.in/sector/bfsi-fintech-financialservices#:~: https://bfsi.economictimes.indiatimes.com https://enterslice.com/learning/types-and-state-of-fintechcompanies-in-india/ https://www.financialexpress.com/ https://www.statista.com https://mobilityforesights.com/product/india-fintech-market/ https://economictimes.indiatimes.com/ https://www2.deloitte.com/in/en/pages/financialservices/articles/fintech-india-ready-for-breakout.html

A STUDY ON ANTECEDENTS OF FOUNDER ENDORSEMENT AND THEIR IMPACT ON BRAND EQUITY AND CONSUMER BUYING INTENTIONS IN JEWELLERY INDUSTRY

Devi Prasad Ungarala*SudhaVemaraju**

Purpose: The current research measures the impact of the attractiveness of founder endorsement on brand equity and consumer buying intentions in the jewellery industry. A jewellery purchase decision is related to investment, social esteem, beauty, and aesthetic aspects. Considerable costs in terms of monetary and psychic are involved. In the chosen brand understudy, the founder endorsement states that prospective jewellery buyers can photograph the gold ornament in any competitor's retail outlet along with price, making charges and depreciation value and inviting the buyers to compare with his brand with the other brands.

Design/methodology/approach: A structured questionnaire was designed to capture the impact of founder endorsement on brand equity and the effect on buying intention on a sample of 302. Confirmatory factor analysis was conducted to find the relationship among the constructs.

Findings: The findings reveal that attractiveness positively influences likability, meaning, and recall and these determinants, in turn, impact consumer buying intention.

Originality/value: The study indicates that founders can promote their products to effect positive consumer buying intention. An analytically proved model and outcomes from the analysis would enable marketers to plan an effective marketing communication strategy.

Keywords -Founder endorsement, Attractiveness, Likability, Meaning, Brand Equity and Buying Intention. **JEL Classification-** M 310

Founder endorsement is a relatively new marketing communication approach to build brands and to increase sales and profits. Founder endorsement, a kind of celebrity endorsement, has been less prevalent in the promotion of brands than celebrity endorsements. About 25 percent of advertisements use celebrity endorsement (Watch, 2006), advertisers induce desirable associations in the endorsed brands (Keller K., Strategic Brand Management, Building, Measuring and Managing Brand Equity., 2013) (Essex. Till, 1998), hoping to create suitable customer-based brand equity (Keller K., 1993) Though extant study explores the effect of celebrity supporters on brand recall (Kahle L., 1985), and BA (brand attitudes) (Till, Sarah M., and Randi, 2008) and intention to buy (Kahle L, 1985), academic study into the outcomes of celebrity supporters on BE remain scant (Seno and Lukas, 2007).

Either celebrity or founder endorsement plays a vital role in the attainment of insolence, brand recall, and buying purpose as well as the bottom line of business (profit and sales) (Agarwal and Kamakura, 1995; Elberse and Verleun, 2012; Erdogan, 1999; Erdogan, 1999; Chan and C.H, 2011; Mathur, Mathur, and Rangan, 1997; Kaikati, 1987). 25 per cent of American commercials (Shimp, 2000) and 24 per cent of Indian commercials (Creswell, 2008) used celebrity endorsers. Research reveals that endorsements through celebrities can result in positive advertisement responses and product assessments (Dean, Dwane , & Abhijit).

I. Review of Literature

The review of literature has been organized into four sections

- Impact of founder endorsement on brand equity.
- Theories of endorsement.
- Impact of founder endorsement factors on purchase intention.
- Effect of brand equity (brand likability, brand meaning, and brand recall) on purchase intention.

Impact of founder endorsement on brand equity:

25 percent of advertisements use celebrity endorsement (Watch, 2006) Associating personalities with a brand, promoters permeate the recognized brand with required relations (Keller K, 2013; Essex.Till, 1998), expected to build positive brand equity among consumers (Keller K, 1993). Though previous research has looked into the effects of celebrity endorsers on brand recall (Kahle L, 1985), endorsed brand ambassador (Till, Sarah M and Randi, 2008),

* Professor, GITAM School of Business, GITAM (Deemed to be University), Hyderabad campus. India

** Associate Professor GITAM School of Business, GITAM (Deemed to be University), Hyderabad campus India and purpose to purchase (Kahle L, 1985), academic study into the implications of celebrity endorsement on brand equity has been scarce (Seno and Lukas, 2007).Given high costs in hiring celebrities, a few firms have deployed their founder as celebrities based on the suitability of the founder with the advertised brand/product.

Impact of founder endorsement factors on purchase intention:

Pradhan et al., 2016 studied and tested the abstract context in comparison of the relative efficacy of celebrity-user, brandpersonality, and user-brand behavior resemblance on brand ambassador and buying intent and found user-brand and brand-personality behavior similarity has a substantial influence on brand ambassador, and intention to purchase, personality-user equivalence does not (Amanda Spry et al., 2011). In general, it may be stated that celebrity qualities have an impact on customer purchase intent.

Attractiveness:

It is an essential determinant in a person's initial decision of another individual, according to a large body of study in marketing and communication (Baker and Chruchill Jr, 1977; Chaiken, 1979; Joseph, 1982; Kahle L, 1985; Mills and Aronoson, 1965; Widgery and Ruch, 1981).

Expertise:

It is well-defined by (Hovland, Janis, & Kelly, 1953) expertise is the additional level of trustworthiness. This dimension is sometimes called "authoritativeness" (McCroskey, 1966), "competence", "ex-pertness" (Applbaum and Anatol, 1972) or "qualification" (Applbaum & Anatol, 1972). (Berlo, Lemert, & Mertz, 1969). The effect of apparent communicator ability ("Dr." versus "Mr.") and the strong point of advice (positive versus unbiased) on the social agreement was investigated by (Crisci & Kassinove, 1973). The results of this learning show that respondents' inclination and act by the source's advice was directly related to the supposed close to fitness and the strength of the assistance.

Trustworthiness:

It is the mark of self-assurance that a consumer reposes in the celebrity and the acceptance level of the celebrity and the essence. (Giffen, 1967) revised the perception of trust from Aristotle to King and felt that Aristotle's "ethos," and (Hovland, Janis, & Kelly, 1953) "credible source" is similar to trustworthiness. The outcomes showed that when the promoter is apparent to be extremely truthful, a prejudiced note was further real than a non-assertive message in the change of perspective.

Elements of the Keller model:

When a product is well-known and has durable, favorable, and separate associations in a buyer's mind, it is said to have brand equity. (Keller K, 1993). Salience, presentation, imagery, findings, opinions, and resonance are the 6 blocks of brand building that make up these developments. The ultimate aim is to reach resonance, the peak of the customer-based brand equity pyramid, where clients and brands have a perfectly harmonic association.

Celebrities and Brand Meaning:

Marketers have typically relied on the charisma of celebrities to give their meaning through their brand. Escalas and Bettman look at celebrities as transporters and pitchers of meaning that assist customers in creating their own identities. As a result, celebrities' influence stems from their capacity to serve as meaning enablers, assisting customers in meeting their own needs rather than their ability to be active relationship partners.

Hypothesis development:

Martin Eisend and Tobias Langner studied (physical) beauty and knowledge as significant aspects of advertising success in their study "Instant and late publicity impacts of celebrity endorser' attractiveness and expertise." When encountering advertisements with celebrity endorsers, these factors were identified as the most appropriate variables to inspect the interchange of effect and perception because they can be measured whichever a reasoning or affecting stimulus that activates whichever is more affect-based or extra cognitionbased procedures. As an affect-based stimulus characteristic, attractiveness states to the recipient's insight of the presenter's adorableness (McGuire, 1985). Physically beautiful personalities are rated higher on a variety of personality attributes (Kahle L., 1985), and the earlier study indicates that good-looking correspondents are further effective at altering views and attitudes, as well as creating intentions to purchase (Joseph, 1982) (Erdogan, 1999).While (Amos et al, 2008) found that similar organized impact sizes for celebrity beauty and proficiency, the resulting scope of solo research is influenced by moderating variables. According to the latest research, purchasing incentives limit the effect of knowledge on brand boldness and intention to buy (Lord and Sanjay, 2009). As a result, knowledge affects brand boldness and buying intention only for informational items (such as ovens or aspirin), but not for transformational products (such as chocolates or ethnic clothes). The impacts of appeal are inverted, with transformational items having stronger effects than informational products. According to this hypothesis, endorser has a greater impact on clients when the celebrity and the endorsed brand have a high level of congruence or fit (Kamins, 1990; D.Till, 1998; Keel and Rajan, 2012). (Brian D. Till & Busler, 1998) on the other hand, discover no connection between attraction and product classification, suggesting that supporter attractiveness affects attitudes about promotions and buying purposes for both attractive-related and attractive-unrelated goods.

Conceptual framework:

It proposed in Figure 1 is based on (McCracken G., 1989) meaning transfer model, Classical conditioning, Elaboration Likelihood Model, and 'Tears Model'.

Figure 1: Conceptual Framework



Source: Framework designed by authors

Theories of celebrity endorsement:

McCracken's meaning transfer model: Research on endorsement offers many interpretations about how celebrities improve an image of a product. McCracken G., 1989 meaning transfer model indicates that the public endows sense to personalities. Thus, personalities stand for a certain explanation, and advertisement makers select celebrities based on the meanings that would be conveyed to representative possessions for a product. Attractiveness and capability are regularly found in personalities who can be transferred to a substantial form to the consumer. Customers can derive control of their explanations and include them into self.

Associative learning or classical conditioning:

Associative learning, also known as classical conditioning, describes how customers shift an unconditioned stimulus (endorser) that causes an undefined reply (e.g., like the celeb supporter) to an accustomed stimulus (a brand), which causes the corresponding feedback (e.g., fondness of the product). A learning method created on (recurring) experience to together stimuli cause response (Tom et al., 1992; D.Till, 1998). The study has concentrated on the source impacts of celebrity endorser to investigate these transfer effects.

Celebrity endorsement impact on brand equity: Increased brand recall (Speck, et.al., 1988), endorsed brand assessment (Marla Royne Stafford, et.al., 2002), and consumer based brand equity are all positively associated with endorser credibility, attractiveness, skilfulness, and honesty (Ohanian, 1990; Spry, et.al., 2011). Endorser reliability is defined as an endorser's morality, truthfulness, and level of confidence (Ohanian, 1990).Endorser skill refers to a buyer's view of an endorser's knowledge, experience, or aptitude with the product being

endorsed. Perceptions of customers on endorser's physical appearance are referred to as attractiveness (Ohanian, 1990). Amanda Spry et al., 2011 investigated the influence of personality credibility on the recognized brand's consumer-based equity.

H1: Attractiveness of the founder positively impacts the likeability dimension of the brand equity.

H2: Attractiveness of the founder positively impacts the meaning dimension of the brand equity.

H3: Attractiveness of the founder positively impact the recall dimension of the brand equity.

H4: *The likability dimension of brand equity positively influences the buying intention.*

H5: *Meaning dimension of brand equity positively influence the buying intention.*

H6: *Recall dimension of brand equity positively influence the buying intention.*

II. Research Design and Methods

Data Collection Method

Data were collected through a study using a structured questionnaire. The self-report questionnaire took between 20-30 minutes to administer. 302 students at a deemed university in India were engaged in the survey. The students were from university- postgraduate students who decided to take part in academic research. The students were instructed to choose respondents who got exposed to founder endorsement advertisements. Hence the sampling technique is purposive sampling. Respondents were chosen from urban and rural places of Telangana state in India.

Questionnaire design:

The questionnaire was designed with the identification of constructs namely founder attractiveness, likability, meaning, and recall. Items for founder attractiveness, likability, meaning, and recall were drawn from validated item scales developed by DeSarbo and Harshman (DeSarbo and Richard A, 1985). The questionnaire has 22 questions and was categorized on a five-point Likert scale from strongly agree to strongly disagree. The pilot study was held with 30 respondents to check the correctness of the dimension items. The respondents replied to the queries relating to attractiveness, likability, meaning, and recall. A few comments were offered by the members to progress the items. The reliability of the scales was tested using Cronbach's alpha and the validity was assessed using confirmatory factor analysis.

Sample and procedure:

A structured questionnaire was designed to capture the effect of founder endorsement on brand equity and the effect of brand equity on buying intention. Initially, the effect of celebrity influencing factors such as expertise, attractiveness, and reliability on brand equity factors namely likability,

meaningfulness, and brand recall, and their effect on buying intention were included in the questionnaire. The model suggested including only celebrity attractiveness and excluding the other factors. A sample of 500 was targeted but after data cleansing, the sample was confined to 302. A structured questionnaire was designed with 22 items. The items of the questionnaire were drawn from the validated scales from the extant literature. Founder attractiveness, brand likability, meaningfulness. brand recall, and buying intention correspondingly have 4, 4, 4, 3, and 7 items. AMOS was deployed to calculate the relevant statistics needed for the study. Cronbach's reliability test was conducted on the 22 items. Descriptive statistics like mean, standard deviation, and variance were calculated. Sample adequacy test KMO and sphericity test Barlett were also done. ANOVA with Tukey's Test for Nonadditivity and Hotelling's T-Squared Test were also conducted. Finally, Confirmatory factor analysis was conducted and various model fit indices were established.

Figure 2: CMA Model Fit Diagram



III. Result and Discussion

The Analysis of data was presented in order; first, Table 1 indicates the reliability of the measurement model was tested using the metric Cronbach's alpha to ensure scale has measurement error less than the threshold value (Nunnally, 1967). The results of reliability were found to be 0.933 which is greater than the threshold level of 0.70. Second, Descriptive statistics like Mean, standard deviation, and variance were calculated. Table 2 shows that sample adequacy test KMO and sphericity test Barlett were also done. Kaiser-Meyer-Olkin Sampling Adequacy measured at 0.909 at .000 significance. Third, Anova with Tukey's test for non-additivity was tested to find out whether the factor variables were additively related to the expected value of the response variable as seen in Table 3. In our case the test for non-additivity was statistically significant, the data are no additive because Mean Square was .781. Fourthly, Hotelling's T squared test was performed to determine the distribution of the sample is a natural generalization or not. Table 4 indicates that Hotelling's tsquared statistic as a generalization of t-statistic confirmed significance in multivariate distribution proportional to the Fdistribution and distribution of a set of statistics which are natural generalizations of the statistics underlying tdistribution.

Lastly, Confirmatory factor analysis was executed to find out whether the factor loading is standardized or not, the model fit is significant or not and the correlation between all factors is less or more. SEM was chosen as an analysis technique primarily because of its ability to explain models with several connected dependant variables, which we encountered in our research. (Hair, Jr, Anderson, Tatham R.L, and Black , 1995).

Table 1: Reliability Statistics					
Cronbach's Alpha N of Items					
22					

Source: Data Analysis

Table 2: KMO and Bartlett's Test

f Sampling	<u>.909</u>
Approx.	2676.59
Chi-	0
Square	
Df	435
Sig.	.000
U	
	Approx. Chi- Square

Source: Data Analysis

			Sum of Squares	Df	Mean Square	F	Sig
Between People		1340.416	291	9.244			
Within People	Between Ite	ems	75.105	19	2.590	5.432	<u>.000</u>
	Residual	Nonadditivity	23.895ª	1	23.895	50.714	<u>.000</u>
		Balance	1980.833	4204	.471		
		Total	2004.728	4205	.477		
	Total		2079.833	4234	.491		
Total		3420.249	4379	.781			
Grand Mean =	3.78			1	I	1	1
a. Tukey's estin	nate of power	to which observation	ons must be raised	to achieve	additivity = 4.8	358.	

Table 3: ANOVA with Tukey's Test for Nonadditivity

Source: Data Analysis

Table 4: Hotelling's T-Squared Test

Hotelling's T-Squared	F	df1	df2	Sig
120.515	3.353	19	234	.000

Source: Data Analysis

Measurement Model Validation:

All CFA estimates that showed a value above 0.60, which satisfied the basic threshold condition for convergent validity (Kline, 2015) are included.

Results of CFA Model:

Confirmatory factor analysis was conducted to test the conceptual model by (Anderson and Gerbing, 1988; Hair et al, 1995). All loadings with factors are standardized and significant in this model. The correlation between all factors is very less and distinct factors. Table 5 indicates that the model is fit and RMSEA values is 0.065 (>= 0.05 and <=0.1). Table 6 depicts that attractiveness model is significant as the RMSEA value is 0.077. Table 7 shows that attractiveness and consumer buying intentions framework model is also significant with a CMIN value of 1.905 (<=3.0). The fit indices of Model met the required threshold expectations; $\chi 2/df$ (< 3) = 1.905, CFI

(>=0.8 and <=1.0) = 0.903, RMSEA (>=0.05 and <=1.0) = 0.078. The path coefficients (β values) for all the hypotheses are significant and hence the hypotheses are supported.

Table 5: Confirmatory factor Analysis (CFA) Model check and Significance measures

Model	RMSEA	LO 90	HI 90	PCLOSE
Default model	.065	.056	.074	.005
Independence model	.189	.182	.195	.000

Source: Data Analysis

Table 6: Attractiveness Model

Model	RMSEA	LO 90	HI 90	PCLOSE
Default model	.077	.057	.097	.014
Independence model	.260	.246	.273	.000

Source: Data Analysis

CIVILIN							
Model	NPAR	CMIN	DF	Р	CMIN/DF		
Default model	101	346.647	182	.000	1.905		
Saturated model	252	.000	0				
Independence model	21	1926.854	231	.000	8.341		

Table 7: Chi-square values of Attractiveness and Consumer Buying Intention CMIN

CMIN value is significant and Model Fit value 1.905 (<=3.0)

Baseline Comparisons							
Model	NFI Delta1	RFI rho1	IFI Delta2	TLI rho2	CFI		
Default model	.820	.772	.906	.877	.903		
Saturated model	1.000		1.000		1.000		
Independence model	.000	.000	.000	.000	.000		

CFI value is significant and Model fit compare to any other Models and value is 0.903 (>=0.8 & <=1.0)

RMSEA							
Model	RMSEA	LO 90	HI 90	PCLOSE			
Default model	.078	.065	.090	.000			
Independence model	.221	.212	.230	.000			
DMSEA wal	PMSEA value is significant and value is $0.078 (>-0.05 \text{ k} < -1.0)$						

RMSEA value is significant and value is 0.078 (>= 0.05 & <= 1.0)

Source: Data Analysis

IV. Conclusion

Today consumers are informed, intelligent and rational especially while buying costly products and the decisions have high involvement. Organizations invest heavily in endorsers' characteristics to align with the attributes of their products. (McCracken G., 1989; Walker and Daniel, 1992). The impact of advertisement on consumer liking, meaning, and recall are significant. The collective effect of consumer liking, meaning, and recall on buying decisions is also significant. Founders as celebrities and their endorsed advertisements are perceived as attractive with both good-looking and likable abilities (Atkin & Martin, 1983). It is a known fact that countries like India, China buy more jewellery ornaments than western countries in the world. Hence investigating the attractiveness of the founder in the founder endorsement advertisement on the brand equity composed with liking, meaning, and recall and the impact of brand equity on buying intention is a significant task. In this situation, an analytically proved model and outcomes from the analysis would enable marketers to plan an effective marketing communication strategy. Founder attractiveness influences brand equity. All three constructs of brand equity namely liking, meaning, and brand recall was positively influenced by the founder's attractiveness. This could be attributed to that founder's attractiveness and his appeal to the consumers in the advertisement impressed in generating liking, meaning, and recall. In comparison to a non-celebrity endorser, celebrities have been widely used for promotional activities because they may elicit favourable emotions and perceptions (RN Kanungo and S Pang, 1973; Cacioppo and Richard, 1980). When clients are uncovered to some personality-brand blends, optimistic associations with the celebrity are transferred to the brand (David H. Silvera and Benedikte, 2004). According to numerous studies, the impact of celebrity endorsement on brand equity nurtures over a period (B.D, 1998)

Implications:

The study can be used for founder endorsement advertisement for products and services in the high involvement category if the founder has attractiveness. The study proved that high involved products when endorsed by the founder lead to positive brand equity. Given that a brand is a significant asset for a company and brand equity is a key strategic concern, marketing managers can effectively utilize the findings of this study. The current study finds that celebrity endorsement can help create brand equity by increasing brand appeal, and it offers principles for choosing a successful founder. According to the findings, the impact of source attractiveness on product purchase intentions is mediated by the level of customer brand equity aspects about the product. By analysing real brands with real potential purchasers, the current study captures important insights and critical lessons from the field in respect to how brand equity should be imagined and quantified. This study is both accessible and appropriate for marketers, and it even includes a tried-and-true questionnaire based on Keller's approach.

Limitation and scope for future research:

The current research is focused on the Jewellery sector. Future studies can empirically observe the impact of founder endorsement in creating brand equity in additional service and product contexts. The outcomes of this study will be more generalizable if it is replicated with diverse product and service categories. The importance of the founder in developing brand equity will be further illuminated by empirical support for the suggested framework in various circumstances. Second, the information gathered for this research is cross-sectional. Customers automatically connect each new experience, positive or negative, to their prior ones and rate it consequently, therefore the future study can use a longitudinal study design (Meyer and André, 2007). Longitudinal facts will enlighten on how customer perceptions of the brand are changing. This will aid marketers in developing a strong brand by allowing them to take corrective action. Market environments, opposition, and the client's state of affairs can all influence founder endorsement, and potentials can be formed by market settings, opposition, and the customer's situation (Meyer and André, 2007). As a result, founder approval may alter depending on market conditions. As a result, we recommend that upcoming research look into the regulating impacts of marketing conditions and individual factors on the link between founder endorsement and brand equity.

References

- Agarwal, J., and Kamakura, W. (1995). "Analysis, The Economic Worth of celebrity Endorsers: An Event Study. *Journal of Marketing*, *59*(*3*), 56-62.
- Applbaum, R., and Anatol, K. (1972). the factor structure of source credibility as a function of the speaking situation. *Taylor and Francis online*, 39(08), 216-222.
- Baker, M., and Chruchill Jr, G. (1977). the impact of physically attractive models on advertising evaluations. *Journal* of Marketing Research, 14(4), 538-555.
- Berlo, D., Lemert, J., and Mertz, R. (1969). dimensions for evaluating the acceptability of message sources. *public opinion quarterly*, *37*(8), 563-576.
- Brian D. Till, and Busler, M. (1998). Matching products with endorsers: attractiveness versus expertise. *Journal of Consumer Marketing*, 15(6), 576-586.
- Cacioppo, J. T., and Richard, E. (1980). Persuasiveness of Communications is Affected by Exposure Frequency and Message Quality: A Theoretical and Empirical

Analysis of Persisting Attitude Change. *Current Issues* and Research in Advertising, 3.

- Chaiken, S. (1979). Communicator physical attractiveness and persuasion. *journal of personality and social psychology*, *37*(8), 1387.
- Cooper, M. (1984). Can celebrities really sell products. *Marketing and Media Decisions*, 19(3), 64.
- Creswell, J. (2008, june 22). Nothing sells like celebrity. *New york times*.
- Crisci, R., and Kassinove, H. (1973). effect of perceived expertise, strengthof advice, and environmental setting on parental compliance. *the journal of social psychology*, 89(2), 245-250.
- D.Till, B. (1998). Using celebrity endorsers effectively: lessons from associative learning. *Journal of Product and Brand Management*, 7(5).
- David H. Silvera, and Benedikte, A. (2004). Factors predicting the effectiveness of celebrity endorsement advertisements. *European Journal of Marketing*, 38(11), 1509-1526.
- Day, G. S. (1969). A two-dimensional concept of brand loyalty. Journal of Advertising Research , 9, 29-35.
- Dean, Dwane , H., and Abhijit, B. Third-Party Organisation Endorsement of Products: An Advertising Cue Affecting consumer pre-purchase evaluation of goods and services. *Journal of advertising*, *30*, 41-57.
- D Pradhan, I Duraipandian, D Sethi (2016). Celebrity endorsement: How celebrity–brand–user personality congruence affects brand attitude and purchase intention, *Journal of Marketing Communications*,
- Debevec, K., and Kernan, J. (1984). . More evidence on the effects of a presenter's physical attractiveness: some cognitive, affective andbehavioral consequences; Advances in Consumer Research. *Association for Consumer Research*, 127-132.
- DeSarbo, W. S., and Richard A,, H. (1985). Celebrity brand congruence analysis . *Current issue and research in Advertising*.
- Elberse, A., and Verleun, J. (2012). "The Economic Value of Celebrity Endoresements. *Journal of Advertising Research*, 52(2), 149-165.
- Essex.Till, B. (1998). "using celebrity endorsers Effectively:lessons from associative learning". *Journal of Product and Brand management.*, 7, 400-409.
- Fishbein, M., and Ajzen, I. (1975). Belief, attitude, intention and behaviour: An introduction to theory and research. *Psychological bulletin*.
- Forkan, J. (1980). Product matchup key to effective star presentations. *Advertising Age*, 51(6), 42.
- Gautam, D., and Sajeeb Kumar, S. (2018). IMPACT OF BRAND EQUITY ON PURCHASE INTENTION OF

SMARTPHONES. Amity Journal of Strategic Management.

- Giffen, K. (1967). The contribution of studies of source credibility to a theory of interpersonal trust in the communication process. *psychological bulletin*, 68(2), 104.
- Hair, J. F., Jr, Anderson, R., Tatham R.L, and Black , W. (1995). *Multivariate Data Analysis* (Vol. 3). new york: Macmillan Publishing Company.
- Hershey H. Friedman, Salvatore Terminin, and Robert Washington. (1976). The Effectiveness of Advertisements Utilizing Four Types of Endorsers. *Journal of Advertising*, 5, 22-24.
- Hovland, C., Janis, I., and Kelly, H. (1953). communication and persuasion.
- Ilicic, J., and Webster, C. (2011). Effects of multiple endorsements and consumer-celebrity attachment on attitude and purchase intention. *Australasian marketing journal*, 19(4), 230-237.
- JC Anderson, and Gerbing, D. (1988). Structural equation modeling in practice: A review and recommended two-step approach. *Psychological bulletin*, 411-423.
- K Debevec, and E Iyer . (1986). Sex roles and consumer perceptions of promotions, products, and self: what do we know and where should we be headed? ACR North American Advances, 13.
- K.K.W.Chan, and C.H, T. (2011). Assessing celebrity endoresements effects in china. *Journal of Advertising Research*, 51(4), 608-623.
- Kahle, L. and. (1985). Physical Attractiveness of the Celebrity endorser: A social adaption perspective. *Journal of consumer research*, 11(4), 954-961.
- Kaikati, J. (1987). Celebrity Advertising: a Review and synthesis. *International Journal of advertising*, 6(2), 93-105.
- Kamins. (1990). An Investigation into the "Match-up" Hypothesis in Celebrity Advertising: When Beauty May Be Only Skin Deep. *Journal of Advertising*, 19(1).
- Karuna Krishna Gauns; Subhash Kizhakanveatil Bhaskaran Pillai; Kaustubh Kamat; Ruey Feng Chen and Ling-Chen Chang, (2018), Impact of Celebrity Endorsement on Consumer Buying Behaviour in the State of Goa, *IIM Kozhikode Society and Management Review*, 7, (1), 45-58
- Keel, A., and Rajan , N. (2012). Celebrity Endorsements and Beyond: New Avenues for Celebrity Branding. *psychology and Marketing*, 29(9), 690-703.
- Keller, K. (1993). Conceptualizing, measuring and managing customer-based brand equity. *Journal of marketing*, 57(1), 1-22.

- Keller, K. L. (2003). Strategic brand management: building, measuring and managing brand equity. *upper saddle river*, NJ .
- Keller, K. (2013). Strategic Brand Management, Building, Measuring and Managing Brand Equity. Pearson Education.
- Kim, J., Allen, C., and Kardes, F. (1996). An investigation of the mediational mechanisms underlying attitudinal conditioning. *Journal of Marketing research*, 33(3), 318-328.
- Kline, R. (2015). *Principles and Practice of Structural Equation Modeling* (4 ed.). Guilford Press.
- Lord, K. R., and Sanjay, p. (2009). Informational and Transformational responses to celebrity endorsements. *Research gate*, *31*(*1*), 1-13.
- Lu, L., Chang, W., and Chang, H. (2014). Consumer attitudes towards bloggers sponsored recommendations and purchase intention.the effect of sponsorship type,product type,and brand awareness. *computers in human behavior*, *34*, 258-266.
- Marla Royne Stafford, Thomas F., s., and Ellen, D. (2002). A Contingency Approach: The Effects of Spokesperson Type and Service Type on Service Advertising Perceptions. *Journal of Advertising*, 17-35.
- Marshall, C. (1987). It seemed like a good idea at the time. *Forbes*, 98-99.
- Mathur, L., Mathur, I., and Rangan, N. (1997). The wealth effects associated with a celebrity endorser: The Micheal jordan phenomenon. *Journal of adversiting research*, *37*(*3*), 67-74.
- McCracken. (1986). Culture and consumption: A theoretical account of the structure and movement of the cultural meaning of consumer goods. *Journal of consumer research*, 13, 71-84.
- McCracken, G. (1989). who is the celebrity endorser?cultural foundations of the Endoresement process. *Journal of consumer research*, *16*, 310-321.
- McCroskey, J. (1966). Scales for the measurement of ethos. *33*, 65-72.
- McGuire, W. (1985). attitudes and attitude change. *Hand book* of social psychology , 233-346.
- Meyer, C., and André , S. (2007). Understanding customer experience. *Harvard business review*.
- Mills, J., and Aronoson, E. (1965). opinion change as a function of the communicators attractiveness and desire to influence. *journal of personality and social psychology*, *1*(2), 173.
- Nunnally, J. C. (1967). *Psychometric Theory* (1 ed.). McGraw-Hill.
- Ohanian, R. (1990). Construction and validation of a scale to measure celebrity endorsers perceived expertise, trustworthiness, and attractiveness. *Journal of advertising*, 19(3), 39-52.

- RN Kanungo, and S Pang. (1973). Effects of human models on perceived product quality. *Journal of Applied Psychology*, 57(2), 172-178.
- Seno, D., and Lukas, B. (2007). The equity effect of product endorsement by celebrities: A conceptual frame work from a co-branding perspective. *European Journal of Marketing*.
- Shimp, T. (2000). Advertising Promotion: Supplemental Aspects of Integrated Marketing Communications (5 ed.). dryden press.
- Speck, P., Schumann, D.C, and Thomson, C. (1988). Celebrity endorsements-scripts, schema and roles: theoretical framework and preliminary. *Advances in Consumer Research*, 15(1), 69-76.
- Till, B. D., Sarah M., S., and Randi , P. (2008). Classical conditioning and celebrity endorsers: An examination of belongingness and resistance to extinction. *Psychology and Marketing*, 25(2), 179-196.
- Watch, M. (2006). "A list celebrity endorsements are failing to dazzle consumers". *European Journal of Marketing* · , 5, 29-30.
- Widgery, R., and Ruch,, R. (1981). beauty and the machiavellain. *communication quarterly*, 29(4), 297-301.

WHAT MAKES ONLINE GAMING PLATFORMS STICKY: A STUDY OF FIVE INDIAN METROS

Soumitra Bose* Kiran Bala**

Purpose- The online gaming industry, the fourth largest segment in India's media and entertainment industry, has been growing at an exponential rate. With 420 million-plus active gamers, India is among the top three nations in the online gaming business. Despite its growing popularity, there is a lack of rigorous academic research to examine reasons for customer stickiness towards gaming platforms.

Methodology- Research has employed quota sampling, at least 400 respondents each from five Indian metros – Delhi, Kolkata, Mumbai, Chennai and Bengaluru – were selected for the survey. Respondents were administered a questionnaire with closeended questions. Data was collected by Computer Aided Personal Interview (CAPI) method. For this quantitative research, statistical tools like Weight of Evidence and Information Value helped identify the strongest factors that influence customer stickiness.

Findings-*Features like playing with stakes, having subscriptions and quality and quantity of gamers emerged as some of the significant factors that determine the degree of a customer's loyalty towards a platform.*

Originality/ **Value-***This study, for the first time, provides insights to platform owners whose monetization models and customer retention plans can be structured around the findings of this paper.*

Keywords: Online Gaming; Real money gaming, Stickiness of Gaming Platforms; Sports economics; media and entertainment; Information Value

JEL Classification Code: Z2 (Sports Economics)

Online gaming is a fast-growing business in India. According to a recent report by FICCI-Ernst and Young on online gaming and Esports, this sector is expected to reach INR 153 billion by 2024 at a CAGR of 15% to become the fourth largest segment (Mondal, 2022) of the Indian media and entertainment sector. The exponential rise in the number of casual online gamers, estimated at 420 million and growing, and its economic consequence on platform owners is an interesting case study. In India, there are several platforms offering skill-based real money games or pay-to-play games like fantasy, selected card games, ludo, chess etc. These games fall under skill and although they may be played with stakes, are not classified under betting or gambling, both of which are forbidden under Indian law. Our research is purely restricted to skill games which don't have 'violence, addictive or sexual content.' (Barik, 2023).

While legal and regulatory challenges remain ("Karnataka High Court on Online Gaming", 2022), online gaming is seen as sunshine sector under the sports and economy vertical. The Indian government itself keen to promote the Animation, Visual Effects, Gaming And Comics (AVGC) sector ("Animation, Visual Effects, Gaming and Comics (AVGC) Sector", 2022). Gaming platforms in India provide both single genre of games like fantasy cricket or rummy or a bouquet of games ranging from casual, hyper-casual to Esports. Potential gamers thus have a wide range of platforms to choose from. In an extremely competitive environment where internet data is available at low cost, smart phones are in comfortable reach of the common man and telecommunication technology is constantly improving, platforms have to continuously innovate to be able to keep their business models attractive and arrest customer churn. Not only do they have to develop games but win the trust of customers and also retain them.

The decision to choose a platform is dependent on several motivational factors like, intrinsic (enjoyment, challenge meaning and purpose) and extrinsic (cash, applause, trophy and status). These have been examined through our research.

Our study is aimed at finding the most compelling factors that propel a gamer to choose and stay on a platform. It is imperative for platform owners to retain their customers and arrest churn. Loss of customers (churn) leads to loss of revenue. Hence, stickiness or platform loyalty is imperative for good business.

In the absence of in-depth research on customer motivation, there is a need to employ structured analysis in the context of the online gaming industry and consumer stickiness and our attempt towards that makes this study unique.

* Research Scholar, School of Journalism and Mass Communication, K.R, Mangalam University, Sohna Road, Gurugram, Haryana, India.

** Professor and Head of the Department of Journalism and Mass Communication, Faculty of Media Studies and Humanities, Manav Rachna International Institute of Research and Studies, Faridabad The main objective of this work is to mathematically predict the factors that propel gamers to choose a particular platform. Advanced statistical analysis is performed on the data to select the main features that drive platform stickiness. We studied a wide range of Information Value calculated on the basis of Weight of Evidence (Bhalla, 2022) to understand our main research question – "why should an online gamer be loyal to one platform or explore multiple options."

I. Review of Literature

The online gaming industry is extremely competitive in India with multiple platforms, regardless of their financial turnover, constantly innovating and modifying their offerings to the gaming public at large. Several factors come into play when a gamer chooses a platform and more importantly, decides to pay and play.

In context of this study, largely aimed at identifying the tangible factors that lead to brand awareness and consumer satisfaction, we have based the survey on three main theories – the 1985 propounded Self Determination Theory (SDT) of well-known psychologists Ryan and Deci, Game Theory propounded by (Burguillo, 2018) and the hedonic (HED) and utilitarian (UT) aspects of consumer behaviour in a digital gaming world by (Storgards, 2011).

The SDT model puts forward how competence, autonomy and relatedness determine the degree of satisfaction and drives motivation – extrinsic and intrinsic. SDT links personality, human motivation and optimal functioning. It posits how motivation is a powerful force in shaping how we behave and react (Ackerman, 2018).

The motivational theory of (Deci, 1975) lays the basis for the understanding of how digital games are chosen and why they are played. (O'Hara, 2017) encapsulated how SDT 'toppled the dominant belief that the best way to get human beings to perform tasks is to reinforce their behaviour with rewards.' Ryan says three basic psychological needs (autonomy, competence and relatedness) drive motivation.

Storgards highlights the importance and purpose of HED (pleasure and enjoyment) and UT (choosing a platform) when studying entertainment information systems. He also speaks about the attitude towards brands and sources of product (platform) recommendation and what features it must have to stand out from the mass. In his study, based on the theories on motivational aspects of digital games consumption, Storgards employed the triangulation method and observes that digital games (and platforms) are high in hedonic value at the product brand level.

The result of Storgards' thesis indicates that digital games should be viewed as experience information goods and how combination of unique value characteristics can be used to create a successful gaming platform that stands out. Using these theories and concepts as foundation, our research attempts to mathematically investigate factors that lead to stickiness.

The Game Theory (Burguillo, 2018), can be described as the mathematical study of decision-making, of conflict and strategy in social situations. Burguillo's mathematical framework works excellently while studying the economics of the entertainment and gaming world where there is fierce competition. It helps explain how we interact in key decision-making processes in order to achieve a flowing or an optimal and seamless mental state of mind (Acland, 2020) to engage in an online platform offering various types of games. In the world of media and entertainment, that is becoming increasingly customer-centric, quantitative analysis is therefore a must.

Researchers have largely adopted logistic regression analysis to understand why clients quit a particular service, or, continue to remain on particular companies. Through customer segmentation (Zhang et al 2022) analysed data from three major Chinese telecom companies to develop a churn prediction model. They opined that by analysing results, telecom companies can effectively predict the possibility to avoid churn and increase their profits.

Churn is a significant problem for companies operating in the banking, telecom, online gaming space. The internet medium offers the scope to explore several options and hence platform owners face massive challenges to retain customers and remain operationally attractive. Churn clearly hurts a business. There is a greater cost to acquire customers than retaining them. Stakeholders are keen to understand how accurately they can identify customers that are about to leave. On the contrary, there is a need to understand why should customers have a greater propensity towards a platform. It is exactly here that there is no in-depth research in India. Reports by industry bodies are not enough.

(Nazarko, 2020) says logistic regression and decision trees are commonly used methods to analyse churn but "sometimes the management only looks for answers which features may suggest that a customer is dissatisfied with the product, rather than for state-of-the-art predictive solutions." She justifies the usage of attribute relevance analysis. It has two important functions like recognising the variables that have the greatest impact on target variable and understanding the relationship between most important predictor and target variable. To execute this type of analysis, Information Value and Weight of Evidence approach is used.

Nazarko has used this attribute relevance analysis technique on huge telecom data sets and concludes that "IV and WoE concepts are simple yet powerful techniques to perform variable transformation and selection," adding: "in contrast to more sophisticated models, they provide high interpretability."

(Xu et al, 2022) have used a similar technique to analyse churn in China telecom, one of the world's largest integrated information service providers. Retaining customers in the face of fierce competition is an important issue for telecom companies and they took the IV and WoE route to determine the factors that may lead to loss of customers.

(Sana et al, 2022) have employed a similar technique to propose a customer churn prediction (CCP) model for the telecommunication industry. To optimize the prediction models, they have done feature selection using the best hyperparameters on publicly available data.

The burgeoning online gaming business in India is dependent on telecom providers and both are strongly interconnected from a commercial aspect. We thus opted to explore the IV and WoE path as there are no papers that investigate business characteristics of the online gaming business and establish a relationship between gamers and factors that influence stickiness (or churn) in active platforms.

In India, there are some reports (Mondal, 2022) from industry bodies like FICCI, Ernst & Young and IAMAI, but they are largely a bird's eye view on how the industry is shaping up in terms of their business models and under certain regulatory frameworks. These industry reports by no means reflect consumer behaviour and what drives potential players to engage with online games or choose specific platforms.

II. Research Design And Methods

The ground realities in the Indian online gaming environment were considered in creating the blueprint for data collection through a robust survey method. Online gaming is a state subject. From a practical standpoint, we wanted a Pan-Indian overview but with a little more emphasis on southern India where the online gaming industry is facing serious regulatory and legal challenges (Sachthey, 2022). Hence out of the five metros, two south Indian cities -- Chennai and Bangalore -were selected. It may be pertinent to mention here that the states of Maharashtra (represented by Mumbai), Delhi and West Bengal (represented by Kolkata) are not against online gaming.

Real money gaming is big in India (Sharma, 2022) and since gamers choose to select games from platforms, the schedule of questions was devised to determine the possible factors that lead to platform stickiness. Vis-à-vis the theories of motivation on which this study is based on, we earmarked questions to measure motivation, satisfaction, human-device interaction, economic flexibility, type of games preferred, internet factor, and continuous intention. All these factors have a benefit-linkage and value-linkage relationship with each other and the best combination are strong factors for consumer loyalty (Ghosh Sharma et al, 2020).

The survey contained a range of question-and-answer system. The choices were from single answers, multiple tick boxes and questions on a five-point Likert scale (Borah et al, 2020). Our Survey Methodology was Face to Face Surveys using CAPI (Computer Aided Personal Interviews). The survey questionnaire (data collection instrument) was scripted using the Sawtooth Software. Sawtooth Software provides advanced analytics and insights to help organizations understand what's important to their customers and to predict what they will buy or choose. The Sawtooth Software app then helped us administer the questionnaire via smartphone/tablets.

We used Quota Sampling and targeted the age group between 12 to 54 years and sub-divided this into five different age categories – 12-17 years, 18-24 years, 25-34, 35-44, 45-54. We also maintained a 50:50 male-female ratios in all five cities. Every city covered in the survey was first divided into North, South, East, West and Central Zones. Within each zone of the city, we selected two-three areas/localities and within each area/locality, we selected five starting points – popular places like shopping malls and multiplexes that have large footfalls - and from each starting point, we physically reached out to people and enquired if they played online games. If they said yes, we administered the questionnaire to 8 to 10 respondents. The response rate was thus almost 100 percent. It took about 20 minutes per interview.

The areas/localities that we picked were a mix of affluent and middle-class. The 12-17 years' group largely comprised school-going children. This process enabled us to reach a wide range of respondents from various income groups: less than INR 5000 (Indian Rupees) to more than INR 100,000.

Age Group (in years)	Bangalore	Chennai	Delhi	Kolkata	Mumbai	Grand Total
12 – 17	44	42	42	45	46	219
18 – 24	94	91	100	95	97	477
25 – 34	107	91	95	95	93	481
35 – 44	95	95	92	94	97	473
45 – 54	100	91	95	93	90	469

Table 1: Sample breakdown per city in India (mbers per city include female and male respondents)

We targeted at least 400 online gamers from each city and we defined the specifics of the sample in terms of age, sex, education level, income level, device used and internet usage. The schedules were filled by enumerators by personally visiting the respondents. The final data was collected in the first week of April 2022. The final sample breakdown by age group and city is shown in Table 1.

Identifying Key Questions

We received responses from 2116 online gamers. To analyse the collected data, a funnel approach was used. We shortlisted 25 questions (like game time, subscription, playing with stakes/money) that were directly related to gamers' choice of platforms. Our Target Variable (TV) for analysis is the question from the survey on "whether gamers wanted to play on one platform or multiple platforms." This question helps us to measure a gamer's stickiness towards a platform.

Correlation Analysis

Correlation coefficient was calculated between the target feature i.e. platform stickiness versus features like monthly subscription, playing with stakes etc. Correlation enabled us to measure the strength of a relation between our target variable and shortlisted features.

As part of the pruning process, we picked the questions with the highest correlation value. Values, which were less than 0.1, were discarded. The number of questions were thus cut down to 14 (*Appendix 1*). This correlation value can either be positive or negative. Positive refers to where one influences the other in a positive direction. For example, the question of 'Do you play with money' had a correlation value of 0.28. It meant stickiness to a platform that had a strong relationship with people who play with stakes. The question on 'Do you have a paid subscription' had a negative value of -0.31. It suggested most people don't prefer to have subscriptions and prefer a free entry.

 Table 2: Table of correlation values of selected features

 from questionnaire to the target feature

 (nletform stickings)

Features	Correlation
(Selected from the survey	value with
Questionnaire)	Platform
	Stickiness
Earning Money	0.20
Monthly Expense On Internet Data	0.19
One Or Multiple Platforms	1,00
Play With Money	0.28
Average Hours Spent	0.13
Preferred Subscription	0.34
Budget For Paying Online Games	0.19
Have You Ever Paid For Subscription	-0.31
Have A Paid Subscription	-0.30

SMS In-App Ad Emailers	0.08
Wide Variety of Games Important	0.16
Quality Of Players	0.16
Quantity Of Players	0.17
In-App Purchases	0.13

It can be noted from Table 2, the highest correlation value is 0.34. This is between target feature and the question: "preferred mode of subscription" which represents both the economic condition and level of commitment of a gamer to online gaming. When responses to the subscription question was further investigated, it was found that 70% of the sample opted for a free subscription while only 10% said that they prefer a fully-paid model.

Another feature that is well correlated to the target variable is whether users play with money or not (0.28). There seems to be an affinity of platform and playing with money, since more players who prefer to play with money do so on a single platform. For a more focused mathematical approach, we sought to identify the strongest features determining stickiness of a platform by using the statistical concept of Weight of Evidence and Information Value (IV) for all the (14) features that had correlation values higher than 0.1.

Information Value explains how strongly independent features influence the target variable, i.e., platform stickiness.

Platform stickiness is modelled as a binary event - 1 (stays on single platform) or 0 (multiple platform), hence an approach could be to use a logistic regression (with stickiness as dependent variable) and the 14 independent variables, however this would mean assume a linear relationship between dependent and combination of independent variables. This assumption is hard to justify in our research context, without transforming our discrete variables suitably.



Figure 1: Relationship between subscription type and choice of platform

At the core, our independent variables such as platform subscriptions, budget for online games, gaming motivation, using money to play are complex behavioural indicators which have a non-linear relationship with the Target variable choice of single vs multiple platforms. Such relations are illustrated below in Figure 1 & 2. Given this behaviour, linear regression-based model is a sub-optimal choice.



Figure 2: Relationship between platform stickiness and gamer's monthly income

Another significant challenge is existence of multicollinearity which can cause a decrease in the performance of (linear) regression type models, since there is a chance for overfitting. However, Information Value and Weight of Evidence are not influenced by multi-collinearity.

Feature Selection

WoE and IV are common techniques used for selection of features (in our case the questions) that influence platform stickiness.

The WoE is calculated as follows:

WoE = ln(non - event/events)

where, In stands for natural log. 'Event' means players preferring to play on single platform and 'Non-event' means players preferring to play on multiple platform (i.e. churn)

$$IV = \sum (\% \text{ non} - events - \% \text{ events}) * WoE$$

The IV is calculated per response option in a question. Then the individual IVs per option is summed to give the total IV for that question.

Our target variable, as discussed above, is stickiness (single platform vs multiple platforms) which is a question in the survey. The following thresholds values were used to classify the IV's predictive power (Kulkarni, 2022): When.

- IV < 0.02 (Not useful for prediction)
- 0.02 < IV < 0.1 (Weak predictive power) •
- < IV < 0.3 (Medium Predictive power) •
- < IV < 0.5 (Strong predictive power) •
- IV > 0.5 (Suspicious predictive power)

III. **Results and Discussion**

The IV analysis selects six different features covering economic ability, motivation and continuous intention that show good predictive power on the basis of the thresholds discussed above. The six features pick a range of behavioural and economic indicators that describe a gamer's choice to stick to platforms and their behaviour on them.

Options	Multiple platform (non-event)	Single Platform (event)	Total	Proportion	WoE	IV
Without stakes	245	1132	1377	65%	0.59	0.19
Sometimes with stakes	217	262	479	23%	-0.75	0.15
Frequently with stakes/Always	131	129	260	12%	-0.96	0.13
Total	593	1523	2116			0.47

Feature 1: Pay to play - Economic intention (Utilitarian motivation)				
reature 1. ray to blay - Economic intention (Otintarian motivation)	Footure 1. Dov to	nlow Foonomia	intention (Litilitarian	motivation)
	reature 1. ray to	Diav - Economic		mouvation)

Table 3: WoE calculation for the feature 'Playing with Stakes' is shown.
 This question measures economic ability of an online gamer

'Playing with stakes' measures a gamer's ability to spend and his/her urge to win. Since the IV is high (0.47), it is a strong predictor for platform stickiness. It is observed that 1132 out of the total sample of 2116 respondents (roughly 54%) who prefer to play without stakes usually stick to a single platform. Those who play with stakes tend to go to multiple platforms (perhaps to increase their chances of winning).

Options	Multiple platform (non-event)	Single Platform (event)	Total	Proportion	WoE	IV
Yes	320	338	658	31%	-0.89	0.29
No	271	1187	1458	69%	0.53	0.17
Total	591	1525	2116			0.45

Feature 2: Paid subscription - Continuous intention (Utilitarian motivation)

Table 4: WoE calculations for the feature 'Have a paid subscription'.
 This feature measures continuous intention and economic ability

'Having a paid subscription' indicates a gamer's economic ability and his level of loyalty to a platform. According to the IV thresholds, this is also a strong predictor. It is observed that in our sample, almost 70% of respondents don't have any kind of subscription and there is an overwhelming bias towards staying on a single platform. Of the respondents who have said yes to having a subscription – there is a marginal inclination towards staying on a single platform (perhaps the platform to which they subscribe). One of the reasons why we don't see a major difference between single platform subscription and multiple platform subscription is probably these people have the means to pay and play on more than one platform. Thus, it is hard to conclude from this data that people who have subscriptions will almost certainly stick to a single platform

Options	Multiple platform (non- event)	Single Platform (event)	Total	Proportion	WoE	IV
Don't care (Neutral)	95	434	529	25%	0.57	0.07
Want	247	676	923	44%	0.06	0.00
Strongly Want	249	415	664	31%	-0.44	0.07
Total	591	1525	2116			0.14

Table 5: WoE calculation for the feature 'Quantity of Players Importance'.
 This feature measures utilitarian motivation.

'Quantity of Players Important' measures the number of gamers that engages in a platform. Numbers are indicative of the popularity of a platform. It is a medium predictor but extremely significant when it comes to choosing a platform. It is observed that 75% of the sample believe that the quantity of players (subscriber base) is important to join a platform. This base is a direct indication of the popularity of a platform and there could be several factors like security, customer service, peer suggestion, celebrity endorsement, ease of operation, bouquet of games, influencing public choice.

Most platforms offer more or less similar content in terms of games and therefore innovation and trust factor in service are critical. It is observed that 1091 people, who want/strongly want a solid subscriber base, prefer a single platform.

It is by far the majority. It is also observed that a section of the sample respondents, who want/strongly want to be influenced by the number of players, is inclined towards exploring more than one platform. Interestingly, a quarter of the sample is not influenced by the number of players. This portion of the sample is largely in favour of a single platform. Overall, respondents tend to have preferences and popularity largely drives gamers to a single platform.

Options	Multiple platform (non-event)	Single Platform (event)	Total	Proportion	WoE	IV
Yes	448	1390	1838	87%	0.18	0.03
No	143	135	278	13%	-1.01	0.15
Total	591	1525	2116			0.18

Feature 4. Gratification – Extrinsic Motivation

Table 6: WoE calculation of the feature 'Earning money'.
 This feature helps to measure utilitarian motivation

'Earning money' is a medium predictor for stickiness but high on motivation level as 87% of the sample say that earning money drives them to play online games. And when it comes to choosing a platform, there is strong tendency towards a single platform. People who pay-to-play online games look

for security, trust and ease of transaction from platform owners. If the majority of the sample is banking on a single platform, it can be safely argued that these gamers feel secure and confident in using the interface of a platform they engage in.

Options	Multiple platform	Single Platform	Total	Proportion	WoE	IV
	(non-event)	(event)				
Less than 1 Hour	102	427	529	25%	0.48	0.05
1-3 Hours	401	960	1361	64%	-0.07	0.00
More than 3 hours	88	138	226	11%	-0.50	0.03
Total	591	1525	2116			0.08

Feature 5: Immersion – (Hedonic and Utilitarian Motivation)

Table 7: WoE calculation for 'amount of gaming time'. This feature measures hedonic (pleasure) motivation

The feature 'amount of gaming time' may be a weak predictor but gives an idea of how much time gamers are spending on platforms. Sixty-four percent of respondents were playing between 1-3 hours per day. Irrespective of the number of hours spent on game time, it is noticed that there is a strong preference (1525 vs 591) to choose one platform, although the 1-3 hours' group (401) may like to try out more than one.

Options	Multiple platform (non-event)	Single Platform (event)	Total	Proportion	WoE	IV
INR 0-500	1163	356	1519	72%	-0.24	0.04
INR 501-1000	315	170	485	23%	0.33	0.03
Greater than INR 1001	47	65	112	5%	1.27	0.10
Total	1525	591	2116			0.17

Table 8: WoE calculation for the feature 'monthly budget for playing games'. This feature measures economic ability.

'Monthly budget for playing games' is a medium predictor but extremely significant considering monies are involved. A large majority 72 % of the sample has a monthly budget of 0-500 INR to play online games. And they choose multiple platforms. We have seen in feature 4 that earning money is a significant driver and hence gamers with relatively lower budgets are exploratory in nature and seek a return. This group may also include those who use platforms as a hobby or simply, time pass. They have no attachment towards any particular platform.

As budget increases, the tendency to lean towards a single platform also increases and is the maximum in the INR 1000+ bracket. More sign of preference is evident from the data and it can be observed that more serious players, albeit much lower proportion of the sample, stay on a single platform. This may be because they have a preference of the type of game they play.

It can be surmised that the six features discussed above greatly impact stickiness towards a platform. It is observed that money and time spent on playing games offered by a platform is directly related to the state of flow, which effectively indicates the level of satisfaction - both intrinsic and extrinsic – a subscriber has derived from playing on a particular platform. These fall in line with Ryan and Deci's Self Determination Theory (1985). It is thus imperative that an online gaming platform provides an ecosystem that helps gamers achieve a state of mind that enhances motivation, both hedonic and utilitarian, to engage with a platform, remain focused and stay free from distraction.

Our study, based on extensive survey from five Indian metros, thus meets our basic endeavour to explore the factors that contribute to stickiness towards a gaming platform. As Storgards (2011) suggested, the creation of unique value is vital and that can be achieved by examining the motivational aspects of digital games consumption among others. Needless to say, our study is unique to the Indian ecosystem but consumer behaviour is very much in sync with the theories on which we based our research.

IV. Conclusion

Considering that there are multiple platforms jostling for space in a highly competitive industry, this research sheds light on some of the behavioural patterns that drive gamers preferences towards choice of platform. It is evident from the study that utilitarian motivation is a strong driver and a large portion of our sample paid to play with the sole objective to win money. Features 4 and 6 show that most prefer a single platform. We think that this may be because the user interface is easy and friendly for carrying out financial transactions. It is also seen that gamers have a strong affinity towards platforms with high user base (Feature 3) and also prefer to stay on single platforms that are trustworthy. Any game involving stakes involves a risk factor and it is observed that people with higher monthly budgets (Feature 6) and earnings tend to opt for a single platform.

It is also observed that the age-group between 18-34 is the most active with an average playtime of 1-3 hours per day. This age group has access to good mobile devices, has relatively more disposable income to play with stakes and are always looking for innovations. This study will help online gaming platforms and other stakeholders involved with the ecosystem to understand behaviour of players and make strategies to retain their customer base.

Restricted to just five cities, this study is by no means a complete picture of how gamers choose their platforms to play. Similar studies could be extended to Tier-2 and Tier-3 cities, where a large chunk of India's 420 million casual gamers resides (Chakravorty, 2022) and feeling the pulse of India's digital revolution thanks to the availability of low-cost internet.

Acknowledgement

This research paper would not have been possible without the financial support of MPL Sports Foundation. From conceptualisation, theoretical frame-working, research designing, data calculation and final interpretation of results, we are thankful to Dr Anwesha Bose and Ritwik Ghosh, the principal research assistants in this project. Professor Dr Ravi K. Dhar also played a crucial role in reviewing and structuring this paper.

References

Ackerman, C. (2018, June 21). Self-Determination Theory of Motivation: Why Intrinsic Motivation Matters. PositivePsychology.com. https://positivepsychology.com/self-determination-

theory/

- Acland, D. J. (2020). An Investigation of Flow Theory in an Online Game. *Review of Behavioural Economics*, 7(4), 317–336. https://doi.org/10.1561/105.00000127
- Animation, Visual Effects, Gaming and Comics (AVGC) Sector. Drishti IAS. (2022). Retrieved 26 July 2022,

from https://www.drishtiias.com/daily-updates/dailynews-analysis/animation-visual-effects-gaming-andcomics-avgc-sector.

- Barik, S. (2023, January 3). Draft rules for online gaming: What are the regulations govt has released, and why. The Indian Express. https://indianexpress.com/article/explained/draftrules-for-online-gaming-released-8358527/
- Bhalla, D. (2022). Weight of Evidence (WOE) and Information Value (IV) Explained. ListenData. Retrieved 26 July 2022, from https://www.listendata.com/2015/03/weight-ofevidence-woe-and-information.html.
- Borah, S.B., Prakhya, S. & Sharma, A. Leveraging service recovery strategies to reduce customer churn in an emerging market. J. of the Acad. Mark. Sci. 48, 848– 868 (2020). https://doi.org/10.1007/s11747-019-00634-0
- Burguillo, J.C. (2018). Game Theory. In: Self-organizing Coalitions for Managing Complexity. Emergence, Complexity and Computation, vol 29. Springer, Cham.
- Chakravorty, P. (2022). Retrieved 27 July 2022, from https://www.outlookindia.com/business/while-india-s-got-talent-the-gaming-industry-struggles-to-find-it-news-200287.
- Deci, E.L. (1975) Intrinsic motivation Plenum, New York.
- India's Booming Gaming Industry | IBEF. (2022, November 18). India Brand Equity Foundation. https://www.ibef.org/blogs/india-s-booming-gamingindustry
- Tripti Ghosh Sharma, Preeti Tak & Ankit Kesharwani (2020): Understanding
- Continuance Intention to Play Online Games: The Roles of Hedonic Value, Utilitarian Value and Perceived Risk, Journal of Internet Commerce. https://doi.org/10.1080/15332861.2020.1756189
- Kulkarni, K. (2022). Understand Weight of Evidence and Information Value! - Analytics Vidhya. Analytics Vidhya. Retrieved 27 July 2022, from https://www.analyticsvidhya.com/blog/2021/06/unde rstand-weight-of-evidence-and-information-value/
- Karnataka High Court on Online Gaming. Drishti IAS. (2022). Retrieved 26 July 2022, from https://www.drishtiias.com/daily-updates/daily-newsanalysis/karnataka-high-court-on-online-gaming.
- Mondal, P. (2022). Online gaming will continue to grow and reach over 500 million gamers by 2022: FICCI EY report - AnimationXpress. AnimationXpress. Retrieved 26 July 2022, from <u>https://www.animationxpress.com/games/online</u> gaming-will-continue-to-grow-and-reach-over-500million-gamers-by-2022-ficci-ey-report/.

- Nazarko, K. (2020, November 30). Churn Analysis Using Information Value and Weight of Evidence. Medium. https://towardsdatascience.com/churn-analysisinformation-value-and-weight-of-evidence-6a35db8b9ec5
- O'Hara, D. (2017, December 18). The intrinsic motivation of Richard Ryan and Edward Deci. Apa.org. aahttps://www.apa.org/members/content/intrinsicmotivation#:~:text=Deci%20and%20Ryan%20devel oped%20the
- Sample Surveys. Sawtoothsoftware.com. (2022). Retrieved 27 July 2022, from https://sawtoothsoftware.com/academics/teachingresources/sample-surveys.
- Sachthey, A. (2022). Are India's online gaming infrastructure and regulations adequate | The Financial Express. Financialexpress.com. Retrieved 27 July 2022, from https://www.financialexpress.com/brandwagon/writer s-alley/are-indias-online-gaming-infrastructure-andregulations-adequate/2270041/.
- Sana JK, Abedin MZ, Rahman MS, Rahman MS (2022) A novel customer churn prediction model for the telecommunication industry using data transformation methods and feature selection. PLOS ONE 17(12): e0278095. https://doi.org/10.1371/journal.pone.0278 095Sharma, S. (2022). Real Money Gaming: The 101 with fact-checks. Times of India Blog. Retrieved 27 July 2022. from https://timesofindia.indiatimes.com/blogs/voices/realmoney-gaming-the-101-with-fact-checks/
- Storgards, Jan. (2011). The Influence of the Hedonic and Utilitarian Value of Digital Games on Product Recommendation. 17th Americas Conference on Information Systems 2011, AMCIS 2011. 3.
- Xu J, Li X, He Z and Zhou J (2022) Early Warning of Telecom Customer Churn Based on Multialgorithm Model Optimization. *Front. Energy Res.* 10:946933. doi: 10.3389/fenrg.2022.946933
- Zhang, Tianyuan & Moro, Sergio & Ramos, Ricardo. (2022). A Data-Driven Approach to Improve Customer Churn Prediction Based on Telecom Customer Segmentation. Future Internet. 14. 1-19. 10.3390/fi14030094.

(eom)

IKEA INDIA: BUILDING DIVERSITY AND INCLUSION

Arulsenthilkumar S* Punitha N**

The IKEA was started by "Ingvar Kamprad" In 1943. He started IKEA when he was 17 in Sweden. The first store was opened in Norway outside Sweden in 1963 followed by Denmark in 1969². Subsequently, the company expanded its branches across the world. Due to government policies, IKEA was not very keen to start a business in India. Later, the Indian government allowed 100% foreign direct investment for single-brand retail. In 2018, the first store of IKEA was opened in Hyderabad, India¹. IKEA believed that the heart of the business is "Equality, Diversity and Inclusion"³. When IKEA was started in India, the objective of the company was to give equal opportunity to all employees. For improving the diverse culture, Mr. Peter Betzel, CEO of Indian operation said diversity was an integral part of IKEA'S DNA⁴. In India, inclusive culture was the highest priority at the workplace. So that, all the employees were valued, empowered, appreciated and motivated for their unique contribution⁵. Already, IKEA started a couple of stores in India. In addition, the company had a plan to start 2 more stores in Noida and Bangalore. Next 7 years, the company's objective would be to open 30 stores across India⁶. Based on the corporate strategy theory, the company was doing well. However, IKEA had to face a lot of challenges such as heavy competition, policy regulations of India, customers' minds differed from other countries, the rapid expansion of the branches and high land cost⁷. Having said that, the company had a plan to expand its branches across India. Moreover, the company's important objective was to maintain a gender diversity ratio. To achieve the same, the company needs more women employees. Recruiting, training and keeping women employees on the board is a huge challenge In India. So, the challenges are huge; how is IKEA going to take all these challenges in the future? would it be a smooth journey for IKEA in India?

Keywords: Diversity, Inclusion, Diversity and Inclusion, Organizational culture, IKEA, IKEA India.

IKEA is one of the well-reputed and known companies in the furniture segment across the world. The objective of the company was to create a better everyday life for people by offering well-designed products, home furnishing products at low cost to as many as people possible⁷. The secret behind the brand name was the first two letters of IKEA, the founder's name "Ingvar Kamprad". The E stands for "Elmtaryd", it was a form where he grew up. The 'A' stands for Agunnaryd, it was his village close². "Kamprad" was born in rural Sweden in 1926. When he was five years old, he started the business. At the age of 17 years, he started IKEA in 1943. Initially, the company was selling pens, pencils, wallets and frames like small household goods⁸. After five years, started selling furniture and opened the first store in 1959 in Almhult, Smaland. The first store was called Mobel-Ikea. In Sweden 'Mobel' is meant for furniture². Initially, prices were low. So, customers were skeptical about the products. He took the old workshop for rent and displayed furniture to create reliability among customers. Further, the company faced issues in shipping the large size of furniture due to high expenses. To tackle those issues, the parts of the furniture were dismantled and packed in such a way that was convenient for shipping. On account of that, the company could reduce the cost and price of the products too⁸. Nevertheless, the price of the products got reduced, it did not give a good result. Due to this, in 1955, manufacturers boycotted the company which propelled Kamprad to go for in-house manufacturing. He started all the processes such as designing, production and showcase, etc... Which was one of the important driving strategies of Kamprad⁸. In 1963, the first store was opened in Norway outside Sweden followed by Denmark in 1969. Since 2008, IKEA had been the market leader in its segment across the world. So far, IKEA's branches are spread across 49 countries with 415 branches and the company had a plan to open more stores in the future. Around 1,50,000 employees were working. The annual revenue was around \$37.8 billion. In the UK, IKEA sold one-tenth of furniture annually.

Entry in India

In India, the organized furniture market was only 4%¹⁰. Since it was not organized, the early entry of IKEA could support capturing more market as well as attract more customers and instill confidence among them⁹.

Mr. Betzel who handled the Spanish and German operations of IKEA said that ample opportunities for many companies in India. "*a big need in India -- we are one part of that.*"¹¹. In 2006, IKEA applied for permission to do the business. But the Indian government refused.

Because the company wanted to enter without any local joint venture which was against the Indian policies¹². In 2013, IKEA was the first single retailer to get 100 percent regulatory approval from the government to open the store.

* Associate Professor, School of Business (VSB), VIT-AP University.

** Assistant Professor, Ramaiah College of Law, Bangalore

However, it was not materialised¹³. The company took another five years to set up a strategy and invested significantly to open its first store in Hyderabad, India¹⁴. The showroom was 4 lakh square feet and spread across 13 acres. In India, the company offered 7500 different products under its roof. The company focussed variety of product areas such as bedrooms, kitchen, living room and children's room¹³. Around 4 million customers visited the Hyderabad store during the financial year 2019¹⁵. IKEA focused to create 10,000 Employment Opportunities in Maharashtra which led IKEA to open its second store in Mumbai. IKEA's Mumbai store is spread across 23 acres and the company invested \$7 billion in its operation. Subsequently, the company had a plan to open its third store in Bangalore. The company had a plan to open 25 stores by 2025 across India with a total investment of ₹105 billion¹⁶ and generate employment opportunities for 15.000 people¹⁷.

Operations in India

Mr. Jesper Brodin CEO of the IKEA group happily said that IKEA had been partnering with Indian manufacturers for more than 30 years¹³. The company's growth was not only connected to the store sales, the food store was also contributing to good business. IKEA's restaurant business was contributing 5.4 percent of the total revenue. The company followed a cost leadership approach believing it was an apt one for the Indian business environment. Moreover, the company applied strategic management theory by which new designs, methods as well as all strategic issues were resolved⁹. Betzel said that during the lockdown, IKEA's online market picked up and which contributed 40% of the total sales. However, online sales did not hamper the offline business. IKEA practiced sourcing the products in India as its worth was 450 million US\$9. Some market analysts Nirmalya kumar and Lee kong chian, Professor, at Singapore Management University predicted that IKEA would perform well in the long run in India¹⁰.

Culture

Ms. Masson, Country HR Manager, IKEA India, had shared the core values of the company²².

"The next core value we stress on is simplicity. We try to keep things simple. We have a simple and flat structure at IKEA, a simple way of talking to each other, a simple way of behaving. Simplicity allows us to focus on our tasks, on the right things," she adds²².

The company strongly believed in three core values such as togetherness, Simplicity and Consciousness²³. IKEA had been offering very safe, healthy, periodic training programs and a conducive environment for each employee. The company imbibed strong work cultures such as teamwork, cost awareness, modesty and rationale. IKEA encouraged their employees to share their opinions on a companywide survey named "Voice" through which, employees were empowered and company culture also improved. Besides, employees

could share their opinion about the management through the "Ikea Leadership Index". The company practiced strong internal communication at the workplace. IKEA had a "Web-enabled service" by which all employees were connected across the world. IKEA followed the informal management style²⁴.

Challenges

When IKEA entered the Indian market, one of the important challenges was, whether Indian customers would accept IKEA or not? Because, the income levels, taste and resources of the country were different from other countries²⁵. The cultural barrier and how to promote the unknown brand to the Indian market was the important challenge²⁶. Whenever furniture was required for Indian customers, they could order from traditional carpenters directly. It was convenient and the price is also relatively less. It had been a long-time practice in India²⁷. Moreover, carpenters were able to make a design at par with IKEA products and they were using strong wood rather than light one which was being used by IKEA ²⁵. In such a case, how customers were going to order the furniture in a big showroom like IKEA. Certainly, that was the biggest challenge for IKEA. The model of IKEA was "ready to assemble products". However, In India, customers were used to "ready to use products". Analysts felt, changing customers' mood from ready to use to ready to assemble was very tough ²⁷. Betzel said, at the workplace, women employees were supported by the company in terms of language skills, behavior skills and retail operation. Nevertheless, the company supported, employees' families did not allow them to go to work. Initially, that was a big challenge²⁸.

Strategies

The company was working to start smaller stores between 5000 and 10.000 square meters within the city limit to reduce the traveling time of the customers and easy access. On the other hand, the company was tapping small cities as well³⁶. Another strategy was to start the Ingka centres or shopping malls. The size of the shopping malls is between 20,000 to 2,50,000 square feet which included restaurants. entertainment zones, lifestyle-oriented shops and theatres³⁶. Mr. Mia Lundstrom, Creative Director of Life at Home, IKEA India said that IKEA introduced the Scandinavian concept in many countries such as North America, Russia, China, etc. India was also part of that list³⁷. The company visited 1000 homes to understand the customer's tastes, preferences and living standards in India. Based on that experience, the products were designed. Another important strategy was every year around 25 percent of the products used to be changed with new ones³⁷. To satisfy the Indian customers, the company adopted a price-value equation right strategy³⁸.

"Our range will have 500 products below Rs. 100 (\$2) and 1000 products below Rs. 200 (\$4) from the start," Antoni said. "We need to strike a balance between quality and price and this will be a change-driver for us"³⁸.

On the other hand, the company focused on tapping the uppermiddle-class people with western design, very affordable and aspirational products³⁹. The company focused on an omnichannel strategy to improve business performance⁴⁰. Across the world, the company practiced DIY (Do-it-Yourself) strategy. On average, the product could be assembled within 30 minutes whereas in India it was just a minute. The clicking systems were introduced instead of screws said Engman⁴².

Diversity & Inclusion

*IKEA believed, the heart of the business was "Equality, Diversity and Inclusion"*⁴⁴.

When IKEA was started in India, the objective of the company was to give equal opportunity to all employees. To achieve this, IKEA focussed on Equality, Diversity and Inclusion (ED & I). IKEA believed ED & I was not a strategy but an inherent nature of a company⁴⁵. The company strongly believed in gender diversity and considered it as equal as basic human rights. Especially, in the Indian operation, it was one of the key prerequisites⁴⁶. Therefore, Diversity & Inclusion was a part of the company's vision and value system. The company realized that it brought a win-win situation⁴⁷. For improving the diverse culture in India, Mr. Peter Betzel, CEO of Indian operation said diversity was an integral part of IKEA'S DNA and likely his successor could be a woman from India.

Also, he said at the Bloomberg Equality Summit

"We thought it is essential that India's diversity is represented in our workforce. In India we decided consciously to have 50% women in the workforce. Generally, in India, the female workforce is very low. So, that was why we consciously decided to make a difference,"⁴⁸

IKEA's first store had 48% women employees in India. Initially, the company had a plan to equalize the gender ratio. However, it was 2% lesser than global operations. But, for Indian companies, it was a big trendsetter⁴⁹. When the first store was launched in Hyderabad, Mr. Ramesh Abhishek, DIPP secretary appreciated IKEA's efforts in terms of diversity and inclusive culture and he said

"Women are better than men in every respect and they just need these opportunities and by giving it to them, Ikea is actually showing the way to other corporates. It is actually a win-win for everyone,"⁴⁹.

As per the "Jobs for her's DivHERsity" benchmarking report 2019, in India, only 10% of women employees were included in the decision-making process by more than 50% of employers. In addition to that, 49% of Diversity & Inclusion programs were implemented in India after 2015⁵⁰. Hence, for the betterment of ED & I and employee engagement, the company focussed on three important steps. The first one was

to balance the gender workforce and infuse leadership among all people. The second one was to focus on people's careers and to provide leadership opportunities to more women. The third one was to allow all employees to share their opinions on the development of the organization⁴⁵. In addition, IKEA was practicing a 50:50 gender balance at the workplace and never compromised on this. The company kept a women's ratio at 60% and 48% at the senior leaders' level and coworkers respectively across all the units⁴⁶. Further, the company gave more importance to recruiting specially-abled talented people and was very keen to give a safe, secure and pleasant environment to those employees⁴⁵. For maintaining equality, on May 17, 2018, IKEA celebrated "IDAHOT". The objective behind this was to inculcate Diversity, Equality and Inclusion in the workplace. Besides, IKEA conducted a special campaign "Besomeoneshome". It was all about welcoming the LGBT+ people to the board⁵¹.

IKEA celebrated an international women's day celebration in which fun programs were conducted and couples were asked to share their work-life balance plan at home. Also organized another program named "fiftyfifty" intending to practice equality at home and in society. Hence, IKEA encouraged people to open and honest conversations at home. IKEA's Instagram had "fiftyfifty" and it was available across 31 countries. That apart, each year, on March 8, the company celebrated ED & I day and welcomed more women to work after their maternity⁵². Moreover, in India, the company launched transportation facilities for women employees, child care facilities on-site for both men and women and part-time work option also announced⁵³. Already, in IKEA the diversity level reached a 50:50 ratio for all the management positions. Now, the company focussed on the same at the board level too⁵². IKEA had taken a good initiative for pay disparity. In India, the pay gap between men and women was -0.095%, whereas in other industries it was -30% to -40%. Therefore, the company had an idea to close the pay gap completely⁴⁵. In India, inclusive culture was the highest priority at the workplace. So that, all the employees were valued, empowered, appreciated and motivated for their unique contribution⁵⁴. Therefore, IKEA created a strong inclusive work culture at the workplace. Further to accomplish this objective, the company joined with "Workplace Pride" which is a non-profit umbrella organization and it had been

supporting LGBT communities at the workplace⁴⁷.



IKEA India Wins Asia Pacific UN Women Award for Most Gender-Inclusive Workplace, https://humancapitalonline.com/News/details/2040/ikea-india-

https://humancapitalonline.com/News/details/2040/ikea-indiawins-asia-pacific-un-women-award-for-most-genderinclusiveworkplace, accessed on 25.06.21.

Road ahead

India is a wide market; many foreign players were ready to do business. Hence, there was a huge challenge for IKEA, in terms of customers' bargaining power. As IKEA was new to the market, the company did not understand what kind of products were loved by the Indian customers⁵⁵. Already IKEA started two stores in India. That apart, the company had a plan to start 2 more stores in Noida and Bangalore. Next 7 years, the company's objective was to open 30 stores across⁵⁶. Will these rapid expansions give an expected result to IKEA? In India, acquiring land was a big challenge. As everything was getting commercialized, the land cost skyrocketed. Therefore, the company had to spend huge money to acquire land. Moreover, it was tougher to acquire land in big cities like Mumbai, Delhi, Bangalore and all⁵⁷. At the same time, the company had a plan to expand its branches across India. Moreover, the company's important objective was to maintain a gender diversity ratio. To maintain the same, the company needs more women employees. Recruiting, training and keeping the women employees on board was a huge challenge for the company. Therefore, IKEA has to face all different challenges in India. Will it be a cakewalk or a Nightmare for IKEA to do business in India?

Reference

- Yogita Limaye, *Will the first Ikea in India succeed?*, https://www.bbc.com/news/business-45122848, BBC News correspondent, Hyderabad, Published 9 August 2018, accessed on 31.08.21.
- Diversity and Inclusion, https://www.ikea.com/in/en/this-isikea/work-with-us/diversity-and-inclusionpub3d25e8e8, accessed on 25.06.21.
- SaimaAfreen, Equality at work rewarded, https://www.newindianexpress.com/cities/hyderabad/ 2021/feb/02/equality-at-work-rewarded-2258195.html, accessed on 25.06.21.
- Lalatendu Mishra, *IKEA India focuses on gender equality to excel in business*, https://www.thehindu.com/business/ikeaindia-focuses-on-gender-equality-to-excel-inbusiness/article29694142.ece, accessed on 25.06.21.
- IKEA India Wins Asia Pacific UN Women Award for Most Gender-Inclusive Workplace, https://humancapitalonline.com/News/details/2040/ik ea-india-wins-asia-pacific-un-women-award-formost-genderinclusive-workplace, accessed on 25.06.21.
- Muskaan Tekwani, https://www.knocksense.com/india/ikeasexpansion-in-india-takes-an-ambitious-route-newoutlets-in-store-for-30-cities-in-7-years, April 08, 2021, accessed on 01.09.21.
- Sushant Jain, https://www.linkedin.com/pulse/ikea-indiasushant-jain/, Published on April 13, 2021, accessed on 31.08.21.
- Chris Kay, Bloomberg, Ikea rushes to make up lost time on India expansion after delays. https://www.businessstandard.com/article/companies/ikea-rushes-to-makeup-lost-time-on-india-expansion-after-delays-121030600351_1.html, Last Updated at March 6, 2021 12:21 IST, accessed on 31.08.21.
- Sushant Jain, https://www.linkedin.com/pulse/ikea-indiasushant-jain/,_Published on April 13, 2021, accessed on 31.08.21.
- Leeza Mangaldas, *IKEA Will Finally Open Its First India* Store, But Winning Customers Won't Be Easy, https://www.forbes.com/sites/leezamangaldas/2018/0 6/26/ikea-will-finally-open-its-first-india-store-butwinning-customers-wont-be-easy/?sh=edb2cc347fa3, Jun 26, 2018,12:00am EDT, accessed on 31.08.21.
- Chris Kay, *Ikea rushes to make up lost time on India expansion after delays*, https://www.businessstandard.com/article/companies/ikea-rushes-to-makeup-lost-time-on-india-expansion-after-delays-121030600351_1.html, Bloomberg, Last Updated at March 6, 2021 12:21 IST, accessed on 31.08.21.

- Amy Kazmin, https://www.ft.com/content/81f6b6fa-9b78-11e8-9702-5946bae86e6d, Hyderabad, India, August 9 2018.
- Charu Lamba, *The IKEA India experience, strategies and future plans*, https://www.indiaretailing.com/2018/09/10/retail/the-

ikea-india-experience-strategies-and-future-plans/, September 10, 2018, accessed on 31.08.21.

- Ivy Buche and Amit Joshi, The battle for India's \$32 billion furniture market heats up. https://www.fastcompany.com/90388400/the-battlefor-indias-32-billion-furniture-market-heats-up, Published 08-12-19, accessed on 31.08.21.
- Niharika Sharma, https://qz.com/india/1914184/ikeas-onlinestrategy-in-india-has-paid-off-during-the-pandemic/, Published October 7, 2020, Last updated on October 9, 2020, accessed on 01.09.21.
- BI INDIA BUREAU, Focusing on gender equality, IKEA India plans on creating 10,000 employment opportunities in Maharashtra, https://www.businessinsider.in/focusing-on-genderequality-ikea-india-plans-on-creating-10000employment-opportunities-in-the-states-of maharashtra/articleshow/66508251.cms, NOV 5, 2018, 13:18 IST, accessed on 30.08.2021.
- Rishi Iyengar, https://money.cnn.com/2017/03/15/news/india/ikeaindia-parental-maternity-leave/index.html, March 15, 2017: 1:08 PM ET, accessed on 31.08.2021.
- Madhavan Narayanan, *IKEA might face big challenges in India's unique market*, https://asiatimes.com/2018/08/ikea-might-face-bigchallenges-in-indias-unique-market/ August 10, 2018, accessed on 31.08.21.
- Muskaan Tekwani, https://www.knocksense.com/india/ikeasexpansion-in-india-takes-an-ambitious-route-newoutlets-in-store-for-30-cities-in-7-years, April 08, 2021, accessed on 01.09.21.
- Muskaan Tekwani, https://www.knocksense.com/india/ikeasexpansion-in-india-takes-an-ambitious-route-newoutlets-in-store-for-30-cities-in-7-years, April 08, 2021, accessed on 01.09.21.
- Yogita Limaye, Will the first Ikea in India <u>succeed</u>?, https://www.bbc.com/news/business-45122848, BBC News correspondent, Hyderabad, Published 9 August 2018, accessed on 31.08.21.
- Charu Lamba, Understanding the HR Policies of IKEA, https://www.indiaretailing.com/2018/12/17/retail/und erstanding-the-hr-policies-of ikea/#:~:text=The%20ambition%20for%20IKEA%20 in,drives%20and%20assembling%20co%2Dworkers, December 17, 2018, accessed on 30.08.2021.

- Swetha Modgil, You need to be forever curious to grow in IKEA: Anna Carin Mansson, Country HR Manager, IKEA, India. https://www.peoplemattersglobal.com/article/employ er-branding/you-need-to-be-forever-curious-to-growin-ikea-anna-carin-mansson-country-hr-managerikea-india-19207, Sep 10, 2018, accessed on 30.08.2021.
- https://www.ukessays.com/essays/marketing/the-backgroundof-ikea-marketing-essay.php, Published: 10th Jul 2017 in Marketing, accessed on 23.09.21.
- Madhavan narayanan, *IKEA might face big challenges in India's unique market,* https://asiatimes.com/2018/08/ikea-might-face-bigchallenges-in-indias-unique-market/ August 10, 2018, accessed on 31.08.21.
- Learning as we go the challenges of entering a new market, https://about.ikea.com/en/about-us/ikearetail/learning-as-we-go-the-challenges-of-entering-anew-market, accessed on 31.08.2021.
- IKEA opens first store in India but here are five challenges, https://www.timesnownews.com/businesseconomy/companies/article/ikea-opens-first-store-inindia-but-here-are-five-challenges/266799, Updated Aug 09, 2018, 12:09 IST, ET Now Digital, accessed on 31.08.2021.
- Lalatendu Mishra, *IKEA India focuses on gender equality to excel in business*, https://www.thehindu.com/business/ikeaindia-focuses-on-gender-equality-to-excel-inbusiness/article29694142.ece, accessed on 25.06.21.
- B2B Desk, *the big marketing challenge for Ikea in India*, https://business2business.co.in/article/2674/the-bigmarketing-challenge-for-ikea-in-india, 2021-01-06 13:00:20, accessed on 31.08.21.
- I choose candidates who talk about their failures too: IKEA India's HR head https://www.businessstandard.com/article/companies/i-choose-candidateswho-talk-about-their-failures-too-ikea-india-s-hrhead-118070500895_1.html, Last Updated at July 5, 2018 19:22 IST, accessed on 30.08.2021.
- Rishi Iyengar, *Ikea gives India employees six months paid parental leave*, https://money.cnn.com/2017/03/15/news/india/ikea-india-parental-maternity-leave/index.html, March 15, 2017: 1:08 PM ET, accessed on 31.08.2021.
- Taniya Tikoo, *Talent Acquisition strategy for 2018*, http://bwpeople.businessworld.in/article/-Talent-Acquisition-strategy-for-2018/08-05-2018-148489/8, May, 2018, accessed on 30.08.2021.

Charu Lamba, Understanding the HR Policies of IKEA, https://www.indiaretailing.com/2018/12/17/retail/und erstanding-the-hr-policies-of ikea/#:~:text=The%20ambition%20for%20IKEA%20 in,drives%20and%20assembling%20co%2Dworkers, December 17, 2018, accessed on 30.08.2021.

- Swetha Modgil , You need to be forever curious to grow in IKEA: Anna Carin Mansson, Country HR Manager, IKEA, India, https://www.peoplemattersglobal.com/article/employ er-branding/you-need-to-be-forever-curious-to-grow-in-ikea-anna-carin-mansson-country-hr-manager-ikea-india-19207, Sep 10, 2018, accessed on 30.08.2021.
 Charu Lamba, Understanding the HR Policies of IKEA,
 - https://www.indiaretailing.com/2018/12/17/retail/und erstanding-the-hr-policies-of ikea/#:~:text=The%20ambition%20for%20IKEA%20 in,drives%20and%20assembling%20co%2Dworkers, December 17, 2018, accessed on 30.08.2021.
- Surajeet Das Gupta, https://www.rediff.com/business/report/how-ikeaplans-to-power-its-india-growthstory/20201114.html, November 14, 2020 08:30 IST, accesses on 01.09.21.
- Charu Lamba, https://www.indiaretailing.com/2018/09/10/retail/theikea-india-experience-strategies-and-future-plans/, September 10, 2018, accessed on 31.08.21.
- Leeza Mangaldas, *IKEA Will Finally Open Its First India Store, But Winning Customers Won't Be Easy.* https://www.forbes.com/sites/leezamangaldas/2018/0 6/26/ikea-will-finally-open-its-first-india-store-butwinning-customers-wont-be-easy/?sh=edb2cc347fa3, Jun 26, 2018,12:00am EDT, accessed on 31.08.21.
- B2B Desk, *The big marketing challenge for Ikea in India*, https://business2business.co.in/article/2674/the-bigmarketing-challenge-for-ikea-in-india, 2021-01-06 13:00:20, accessed on 31.08.21.
- Niharika Sharma, https://qz.com/india/1914184/ikeas-onlinestrategy-in-india-has-paid-off-during-the-pandemic/, Published October 7, 2020Last updated on October 9, 2020, Accessed on 01.09.21.
- Amit bapna, *Why Ikea is taking the app-route to fast-track its India* growth story. https://www.thedrum.com/news/2021/07/13/whyikea-taking-the-app-route-fast-track-its-india-growthstory - July 13, 2021, accessed on 31.08.21.
- Apparel Resources News-Desk, *IKEA reveals its biggest* challenge just days before foraying into Indian market, https://apparelresources.com/businessnews/retail/ikea-reveals-biggest-challenge-just-daysforaying-indian-market/, 22-June-2018, 2 mins read, accessed on 31.08.21.
- BW Online Bureau, *IKEA India Partners with Tata STRIVE To Build Employability Among Youth.* http://bwpeople.businessworld.in/article/IKEA-India-

Partners-With-Tata-STRIVE-To-Build-

Employability-Among-Youth-/13-08-2021-400507/ 13, August, 2021, accessed on 30.08.2021.

- Saima Afreen, *Equality at work rewarded*, https://www.newindianexpress.com/cities/hyderabad/ 2021/feb/02/equality-at-work-rewarded-2258195.html, Accessed on 25.06.21.
- Parineeta Cecil Lakra, Our approach is to have a genderbalanced workplace, IKEA India, https://hr.economictimes.indiatimes.com/news/workp lace-4-0/diversity-and-inclusion/our-approach-is-tohave-a-gender-balanced-workplace-parineeta-cecillakra-ikea india/80090626, Accessed on 25.06.21.
- Yashmin Raj, Building equitable workplaces for the now and future: IKEA India's Parineeta Cecil Lakra, https://www.peoplematters.in/article/diversity/buildin g-equitable-workplaces-for-the-now-and-future-ikeaindias-parineeta-cecil-lakra-29183, 30 April 2021, accessed on 30.08.2021.
- *Diversity and inclusion*, https://www.ikea.com/in/en/this-isikea/work-with-us/diversity-and-inclusionpub3d25e8e8, Accessed on 25.06.21.
- Lalatendu Mishra, *IKEA India focuses on gender equality to excel in business*, https://www.thehindu.com/business/ikeaindia-focuses-on-gender-equality-to-excel-inbusiness/article29694142.ece, Accessed on 25.06.21.
- POORVI GUPTA, *Ikea Opens First Store In Hyderabad With* 48% Women Staff, https://www.shethepeople.tv/news/ikea-opens-firstindia-store-hyd-48-women-staff/, August 9, 2018, accessed on 30.08.2021.
- Poorvi Gupta, Lessons from ikea diversity and inclusion policies, https://www.ungender.in/lessons-from-ikea-diversity-and-inclusion-policies/, Accessed on 25.06.21.
- *IKEA celebrates people on the occasion of IDAHOT*, https://www.afaqs.com/companies/ikea-celebratespeople-on-the-occasion-of-idahot, Published: 18 May 2020, 4:51 PM IST, accessed on 30.08.2021.
- IKEA drops online game to spark conversation about equality in the home, https://www.adgully.com/ikea-dropsonline-game-to-spark-conversation-about-equalityin-the-home-101790.html.
- Cheryl Cole, Sari Brody, *IKEA: gender equality is a fundamental human right*, https://diversityq.com/ikea-gender-equality-is-a-fundamental-human-right-1506984/ | 17 Jun 2019, accessed on 30.08.2021.
- IKEA India Wins Asia Pacific UN Women Award for Most Gender-Inclusive Workplace, https://humancapitalonline.com/News/details/2040/ik ea-india-wins-asia-pacific-un-women-award-for-

most-genderinclusive-workplace, Accessed on 25.06.21.

- Sushant Jain, *IKEA IN INDIA*, https://www.linkedin.com/pulse/ikea-india-sushantjain/, Published on April 13, 2021, accessed on 31.08.21.
- Muskaan Tekwani, *IKEA's expansion in India takes an ambitious route; New outlets in store for 30 cities in 7 years*, https://www.knocksense.com/india/ikeas-expansion-in-india-takes-an-ambitious-route-new-

outlets-in-store-for-30-cities-in-7-years, April 08 | 2021, accessed on 01.09.21.

IKEA opens the first store in India but here are five challenges, https://www.timesnownews.com/businesseconomy/companies/article/ikea-opens-first-store-inindia-but-here-are-five-challenges/266799, Updated Aug 09, 2018 | 12:09 IST | ET Now Digital, accessed on 31.08.2021.

Jagannath International Management School Vasant Kunj, New Delhi

presents



Radio JIMS Vasant Kunj 90.4 MHz **Voice of The Voiceless**

Jagan Institute of Management Studies

Rohini, Delhi



JIMS Rohini Community Radio 96.9 MHz

This radio is being run by the students and is providing an opportunity to develop programmes for community broadcast. The radio station is used by the college as laboratory for training students specializing in radio broadcast and they work in close coordination with community representatives and leaders. At present the radio broadcasts daily for eight hours with original programme of four hours in morning which is repeated in the afternoon. The students are encouraged to explore the needs of the society, thereafter, they conceive, design and broadcast their own programmes in a real life environment.

RNI No. 64562/96









Nurturing talent

Re-defining Setting new excellence standards...



JIMS creating the future!

Jagan Nath Gupta Memorial Educational Society was established in 1993 to develop & train the next generation of professionals who would contribute towards the economic and social development of our country. The delivery standards, thus have been ensured to provide an inspiring learning environment which helps in transforming learning minds into result oriented professionals.

Commitment to the cause of education

An infrastructure of around 10,00,000 sq. feet spread over 9 State-of-the-Art campuses, cutting-edge technology, professional guidance, practical training, international placements, ever evolving curriculum, choice of the best available professional courses... that's not all, the thrust is on the realization of your highest aspirations.

Enviable Infrastructure

All campuses are hi-tech, wi-fi enabled with state-of-the-art laboratories, Labs, well-stocked along with complete recreational facilities. The classrooms are equipped with multimedia and audio-visual equipments to facilitate effective learning and are designed to promote maximum interaction between the faculty and the students.

Guru Mantra

One of our biggest strengths is our faculty members, who have distinguished academic achievements to their credit and are actively involved in teaching, training, research, consultancy and a big pool of expert guest faculty, comprising specialists from industry, government and research institutions for ensuring a new edge to corporate learning and striking a balance between theory and practice.

The distinct Edge

• First Institute among private sector institutes to have been granted a license for FM Community Radio in Delhi • Hotel Management course ranked 2^{std} in Delhi as per GHRDC survey for CSR 2009 • International Partnerships: Collaboration between Cologne University of Applied Sciences, Germany (CUAS) and JIMS for academics, faculty and student exchange programmes. • Workshops are conducted every year in collaboration with PHDCCI and KAF, Germany for executives of SMEs to develop their awareness, knowledge and skills in areas of Personality Development, Team Building, Total Quality Management, Retail, Banking and Insurance, Project Management etc. • International Conferences with participation of experts from World Bank, International Monetary Fund (IMF), Asian Development Bank, DFID (UK), UK Trade & Economic Policy Division and Australian Trade Commission.

Academic Programmes*

The academic programmes are specifically designed keeping in mind the current Indian economic scenario and the requisite corporate needs that expose the students to concepts, techniques and decision-making tools through an interactive learning process.

The courses are offered at various post graduate and under graduate levels at various campuses according to the needs of the aspirant at large:

Management	Commerce	Engineering					
Information Technology	Journalism (Mass Comm.)	Hotel Management					
Art & Design	Architecture	Law					
		*Select programmes offered at select campuses					
An excellent learning environment is ensured at all times to display superior leadership qualities along with a value driven mindset and sharp intellectual acumen by way of constant interaction with industry professionals through summer internships, industry visits, guest lectures, seminars, mock interviews, pre-placement talks, campus interviews. Mentoring and Personal Enhancement To prepare and equip students with requisite skills to face the corporate world, Personality Development sessions are organised to help build self-awareness and develop a positive attitude amongst students to cope with time and stress issues.							
For further information contact:							
elhi: ROHINI 45184100 www.jimsindia.org KALKAJI 4061	9200 www.jagannath.org VASANT KUNJ 40619300 www.jims	d.org LAJPAT NAGAR 49219191 www.jimssouthdelhi.com					
lajasthan: JAIPUR 0141-4071551/52/52/52 www.jimsjaipur.org SITAPURA 0141-4071500/555 www.jnit.org Uttar Pradesh: GREATER NOIDA 0120-3819700 www.jimsgn.org							
Haryana	Haryana: BAHADURGARH 0127-699700-715 www.jagannathuniversityncr.ac.in						

Printed, Published and owned by Manish Gupta, at MOR Pkt. 105, Kalkaji, (Opp. Kalkaji Police Station), New Delhi - 110019. Printed by Applauze Creative, No. 3, 2nd Floor, DSIDC, Okhla Indl. Area, Phase-I, New Delhi - 110020. Editor: Amit Gupta